

Elastic wave energy scattering and propagation in composite structures.

Nurkanat Aimakov

Thesis submitted to the University of Nottingham for the degree of Doctor of Philosophy

May 2021

To my family.

Abstract

This thesis considers the problem of high-frequency wave energy scattering and propagation in composite structures. In particular, structures made of the twodimensional plate- or shell-like elements with composite laminate material are of interest. We propose an extension of the Dynamical Energy Analysis (DEA) method for such structures. For this purpose, we develop a semi-analytical and a hybrid Finite Element (FE) and Wave and Finite Element (WFE) methods to compute wave energy reflection/transmission at junctions of arbitrarily layered composite plates.

In the first part of the thesis, a brief review of high-frequency numerical methodologies is presented. Advantages and limitations of each method are discussed. Furthermore, a review of analytical and numerical methods for calculating wave propagation characteristics such as dispersion relations, group velocity and scattering coefficients is presented.

In the second part of the thesis, the main theoretical basics of the DEA method, Classical Laminated Plate theory and the WFE method are demonstrated.

In the third part of the thesis, semi-analytical method for computing the energy scattering coefficients of structural junctions made up of thin composite laminated plates is developed. Expressions quantifying transmission and reflection coefficients as a function of the frequency and the angle of incidence are derived. An effective scattering matrix for a plate with multiple finite stiffeners attached to it is obtained.

In the fourth part of the thesis, a hybrid FE/WFE model that predicts the scattering properties for different junctions of two-dimensional anisotropic composite plates is developed. The influence of the angle of incidence and the frequency on the distribution of the power flow of incident bending, shear and longitudinal type waves is investigated. A detailed comparison with semianalytical evaluations of scattering coefficients derived in the third part of the thesis is presented. The method gives for the first time a detailed recipe for computing scattering coefficients for the generic case of an arbitrary number of composite plates connected at a junction without restrictions on the angles at which the plate meet or the orientation of the principal axis of individual plates.

In the last part of the thesis, the theoretical base of the DEA method for composite structures is developed and discussed. The findings of the third and fourth parts of the thesis are used to derive the stationary wave energy density arising in the composite structure due to a harmonic point and edge sources. Numerical results for the cases of a polygonally shaped plate, an L-shaped composite plate and an electric vehicle gearbox are presented.

Declaration

The work presented in this thesis is the result of my research project as a PhD student at the University of Nottingham and was conducted between October 2016 and October 2020. I declare that the work is my own and has not been submitted for a degree elsewhere.

Acknowledgements

This work has been undertaken as part of the INNOVATIVE doctoral training programme, partially funded by the Marie Curie Initial Training Networks (ITN) action (project number 665468) and partially by the Institute for Aerospace Technology (IAT) at the University of Nottingham.

I would like to express my sincere gratitude to my supervisors, Dr. Dimitrios Chronopoulos and Prof. Gregor Tanner for giving all the possible guidance and support at all stages of this work. Their stimulating suggestions and excellent advice helped me throughout my research and the writing of this thesis. I also would like to thank my assessors, Dr. Neil Ferguson and Prof. Stephen Creagh for their insightful comments and suggestions to improve the current thesis. I would like to offer my special thanks to Dr. Sharad Jain and all colleagues from Romax Technology for their assistance at every stage of my industrial placement. I also wish to express my appreciation and thanks to Dr. Timo Hartmann and Dr. Martin Richter for many interesting and useful discussions. I would like to thank the INNOVATIVE doctoral training programme and the University of Nottingham for funding my study and, in particular, all my colleagues and managers from the Institute for Aerospace Technology (IAT) for the help and support on the day-to-day basis. Finally, I want to thank my parents for their continuous support over my whole life, my wife, Aigerim, for always being there for me, my daughter, Ayana, for being my source of happiness, my brother, Nurbolat, and all my friends.

List of Publications

- N. Aimakov, G. Tanner, D. Chronopoulos, Wave scattering in coupled composite plates, *INTER-NOISE NOISE-CON Congr. Conf. Proc.*, *In*stitute of Noise Control Engineering, Ibiza, 2018, pp. 421-431. [Chapter 4]
- N. Aimakov, G. Tanner, D. Chronopoulos, Wave scattering in coupled composite plates, *INTER-NOISE NOISE-CON Congr. Conf. Proc.*, *Vol. 259, Institute of Noise Control Engineering*, 2019, pp. 4389-4400. [Chapter 4]
- N. Aimakov, G. Tanner, D. Chronopoulos, Transmission and reflection of waves at structural junctions connecting thin orthotropic plates. *Wave Motion.* 102 (2021) 102716. [Chapter 3]
- N. Aimakov, G. Tanner, D. Chronopoulos, Wave energy scattering at intersections of composite plates - a wave finite element approach. *Scientific Reports, submitted.* [Chapter 4]

Contents

\mathbf{A}	Abstract i			
D	Declaration ii			
A	cknov	wledge	ements	iii
Li	st of	Publi	cations	iv
1	Intr	oduct	ion	1
	1.1	Motiv	ation and literature review	1
		1.1.1	Review of high-frequency methods	2
		1.1.2	Review of methods for evaluation of wave propagation	
			characteristics	5
	1.2	Aims	and objectives	13
	1.3	Thesis	soutline	14
2	Bac	kgrou	nd	16
	2.1	Dynar	nical Energy Analysis for isotropic structures	16
		2.1.1	Wave equation	17
		2.1.2	The free-space Green function	18
		2.1.3	Eikonal approximation and ray tracing	20
		2.1.4	Liouville equation and boundary mapping $\ldots \ldots \ldots$	23
	2.2	Classi	cal Laminated Plate theory	29
	2.3	The V	Vave Finite Element method for composite plates	33
		2.3.1	Statement of the problem	33
		2.3.2	Periodic structure theory and eigenvalue problem \ldots .	35

3 Energy scattering properties of line joints connecting compos-

	ite j	plates		40
	3.1	Introd	uction	40
	3.2	Staten	nent of the problem	41
		3.2.1	Governing equations of motion	42
	3.3	Disper	rsion relations, the group velocity and Snell's law	43
		3.3.1	Relations between incoming and outgoing waves at junc-	
			tions for composite plates	44
		3.3.2	Snell's law for composite plates - an example	47
	3.4	Deriva	ation of the dynamic stiffness matrix	52
		3.4.1	Energy flow and the response of the plate	52
		3.4.2	The dynamic stiffness matrix	57
	3.5	Reflec	tion and transmission at plate junctions	59
	3.6	Stiffen	ed plate	61
		3.6.1	A composite plate with one stiffener	64
		3.6.2	A composite plate with a pair of stiffeners	68
		3.6.3	A composite plate with multiple stiffeners	70
	3.7	Comp	utational results	72
		3.7.1	Reflection at the free edge of a composite plate \ldots .	72
		3.7.2	Two composite plates joined at a right angle	77
		3.7.3	Three composite plates joined in a T-junction	83
		3.7.4	Stiffened composite plates	85
	3.8	Conclu	usion	94
4	Way com	ve and posite	Finite Element Analysis of wave propagation in plates	95
	4.1	Introd	uction	95
	4.2	Summ	ary of the WFE method for composite plates	96
	4.3	Calcul	ation of wave numbers and group velocity	98
		4.3.1	Incoming, outgoing waves and wave basis	99
		4.3.2	Group velocity estimation	102

	4.4	Comp	utation of scattering coefficients at junctions of composite	
		plates		104
		4.4.1	Reflection matrix for a single plate system	104
		4.4.2	Scattering at the shared edge between shell plates	105
		4.4.3	Hybrid FE/WFE method of computation of scattering	
			coefficients in three-dimensional solid plates	107
	4.5	Nume	rical results	112
		4.5.1	Cross-ply laminated plates	113
		4.5.2	Angle-ply composite plates	119
	4.6	Concl	usion	122
5	Dyr	namica	l Energy Analysis for composite structures	126
	5.1	Introd	luction	126
	5.2	Ray ti	racing for composite structures	127
		5.2.1	Hamilton function and boundary mapping	127
		5.2.2	Transformation between 2D and 4D phase space densities	3130
	5.3	Basis	set expansion of the problem	131
		5.3.1	Boundary integral operator	132
		5.3.2	Initial phase space densities	135
		5.3.3	Stationary energy and intensity	139
	5.4	Nume	rical results	142
		5.4.1	Test case for validation	143
		5.4.2	Two composite plates joined at a right angle	150
		5.4.3	Electric Vehicle Gearbox	153
	5.5	Conclu	usion	158
6	Cor	nclusio	n	160
	6.1	Main	Contributions	160
	6.2	Furthe	er work	161
Bi	ibliog	graphy		164
\mathbf{A}	ppen	dices		180

Α	Special laminated plate stiffness coefficients derivation			
	A.1 Cross-ply symmetric laminates	. 180		
	A.2 Angle-ply symmetric laminates	. 181		
в	Plane stress-reduced stiffnesses	182		
С	List of Abbreviations	183		

List of Tables

3.1	Engineering constants of Epoxy Carbon UD (230 GPa) material	
	used for individual laminas of a composite plate	47

5.1 Engineering constants of aluminium and stainless steel materials.144

List of Figures

2.1	The set up of the local coordinate system at the boundary
	$X = (s, p_{\shortparallel})$. A ray emitting from the point \mathbf{r}_0 reach the point \mathbf{r}_s ,
	which has the local position s on the boundary Γ . The momen-
	tum vector ${\bf p}$ has normal and tangential components p_{\perp} and p_{\shortparallel}
	at the point s, and θ denotes the angle between the ray con-
	necting \mathbf{r}_0 and s and the normal to Γ
2.2	The lay-up configuration of a composite laminate
2.3	A periodic segment of a composite plate with two different al-
	ternating plies modelled with SOLID185 (a) and SHELL181 el-
	ements (b) in Ansys. The degrees of freedom are grouped into
	internal \mathbf{q}_I , edge \mathbf{q}_L , \mathbf{q}_R , \mathbf{q}_B , \mathbf{q}_T and corner \mathbf{q}_{LB} , \mathbf{q}_{RB} , \mathbf{q}_{LT} , \mathbf{q}_{RT}
	degrees of freedom
3.1	The set up of N plates joined together at a common interface
	along the x_g axis
3.2	Local coordinate system, displacements and tractions on the
	common edge "e" of the <i>j</i> th plate. $\ldots \ldots \ldots \ldots \ldots \ldots 42$
3.3	Schematic of an L-junction of two orthotropic plates. The angle
	ψ_2 here is set to 90°. Red and blue lines represent the fiber
	direction of the plates. The local angles of orientations ϕ_1 and ϕ_2
	are both set to 45°

- 3.6 Longitudinal and shear wave vector curves in a 45° rotated orthotropic plate at a frequency 3000 Hz. Dots represent wave numbers related to outgoing waves while squares - incoming waves. Wave number components k_L^* and k_S^* equal to 6 m^{-1} and 9.3 m^{-1} , respectively - the critical values for longitudinal and shear waves 50

3.8	Longitudinal and shear wave vector curves in a $45^\circ/-45^\circ/45^\circ/-$	
	$45^\circ/45^\circ$ composite plate at a frequency 3000 Hz. Dots repre-	
	sent wave numbers related to outgoing waves while squares -	
	incoming waves. Wave number components k_L^\ast and k_S^\ast equal	
	to $3.71 \mathrm{m^{-1}}$ and $7.8 \mathrm{m^{-1}}$, respectively - the critical values for	
	longitudinal and shear waves	51
3.9	Decay direction angles of various evanescent waves as a function	
	of a ply direction angle in an orthotropic plate with material	
	parameters from Table 3.1	53
3.10	Contour plot of the real part of the out-of-plane displacement	
	produced by the outgoing evanescent bending wave in a 45° -	
	rotated orthotropic plate plotted in the (x, y) plane. Brighter	
	colours represent positive z values, whilst darker ones denote	
	negative z values	54
3.11	A ground composite plate with m stiffener plates attached onto	
	it. The ground plate is considered infinite in both directions,	
	whereas stiffeners are infinite only in their local x coordinate	
	directions.	62
3.12	The schematic of incoming (shown in red) and outgoing (shown	
	in blue) waves travelling in the stiffened plate. A_1^- and A_N^- are	
	the incoming waves from infinity.	62
3.13	A schematic view of a stiffened plate in the yz plane. Here,	
	a_{ij}^- represent the amplitudes of incoming waves, and a_{ij}^+ , the	
	amplitudes of outgoing waves	64
3.14	A schematic view of a stiffener plate in the xy plane. Blue lines	
	represent the direction of the <i>i</i> th ply of the plate. \ldots \ldots \ldots	66
3.15	A schematic view of a plate with a pair of finite stiffener plates	
	in the yz plane. Here, a_{ij}^- represent the amplitudes of incoming	
	waves, and a_{ij}^+ , the amplitudes of outgoing waves	68

- 3.16 A schematic view of a stiffened plate with two stiffeners represented by effective scattering matrices s_{eff}^k and s_{eff}^{k+1} in the yzplane. Here, a_{ij}^- represent the amplitudes of incoming waves, and a_{ij}^+ , the amplitudes of outgoing waves.

71

3.22	Energy scattering coefficients of an L-joint of regular cross-ply
	symmetric laminates with the lamination scheme $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$
	for various incident waves as a function of wave number compo-
	nent k_x at a frequency 3000 Hz

- 3.26 Energy scattering coefficients of an L-joint of regular angleply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency for a wave number component $k_x = 0 \text{ m}^{-1}$ 80
- 3.27 Energy scattering coefficients of an L-joint of regular angleply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency for a wave number component $k_x = 5 \text{ m}^{-1}$ 81
- 3.28 Schematic representation of a T-junction connecting three orthotropic plates. The angles ψ_2 and ψ_3 here are set to 180° and 90°, respectively. Red, blue and green lines represent the ply direction of the plates. The local angles of orientations ϕ_1 and ϕ_3 are both set to 45°, whereas ϕ_2 is equal to $-45^\circ. \ldots \ldots 82$

- 3.29 Energy scattering coefficients of an T-joint of regular angleply symmetric laminates with the lamination scheme $45^{\circ}/ 45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of wave number component k_x at a frequency 3000 Hz. 83
- 3.30 Energy scattering coefficients of an T-joint of regular angleply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency at a wave number component $k_x = 5 \text{ m}^{-1} \dots \dots 84$
- 3.31 Energy scattering coefficients and resonance conditions of a stiffened plate for an incident bending wave as a function of wave number component k_x at a frequency 3000 Hz. The length of the stiffener plate is 30 cm. The lamination scheme of plates is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}...$ 87

3.	35	Energy scattering coefficients of a symmetrically stiffened plate
		for an incident bending wave as a function of frequency at a
		wave number components $k_x = 0 \text{ m}^{-1}$ and $k_x = 5 \text{ m}^{-1}$. The
		lengths of the stiffener plates are 30 cm . The lamination scheme
		of plates is $0^{\circ}/90^{\circ}/0^{\circ}/0^{\circ}$
3.	36	Contour plots of energy scattering coefficients of a symmetri-
		cally stiffened plate with four pairs of stiffeners for an inci-
		dent bending wave as a function of frequency and wave number
		component k_x . The spacings between stiffeners and lengths of
		the stiffeners are 30 cm. The lamination scheme of plates is
		$0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$
3.	37	Contour plots of energy scattering coefficients of a symmetri-
		cally stiffened plate with four pairs of stiffeners for an inci-
		dent bending wave as a function of frequency and phase angle
		$\theta = \arctan(k_y/k_x)$. The spacings between stiffeners and lengths
		of the stiffeners are 30 cm. The lamination scheme of plates is
		$0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$
4.	1	Bending wave vector curve in a $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ com-
		posite plate at a frequency 3000 Hz. Blue dots represent wave
		numbers related to outgoing waves while red squares correspond
		to incoming waves. Angles $\alpha_{1(2)}$ and $\theta_{1(2)}$ represent group and
		wave vector or phase angles, respectively. The wave number
		component k_x^{max} is equal to $53 \mathrm{m}^{-1}$
4.	2	A schematic representation of N plates connected via the shared
		edge. The plates are modelled using two-dimensional SHELL181
		elements. The x axis of all plates is the same
4.	3	A schematic representation of the joint connecting N plates.

- 4.4 A finite element model of the joint connecting N plates. The degrees of freedom are grouped into internal \mathbf{q}_I , edge \mathbf{q}_B , \mathbf{q}_T , $\mathbf{q}_{L,1}$, ..., $\mathbf{q}_{L,N}$ and corner $\mathbf{q}_{LB,1}$, ..., $\mathbf{q}_{LB,N}$, $\mathbf{q}_{LT,1}$, ..., $\mathbf{q}_{LT,N}$ degrees of freedom. 109
- 4.6 Energy scattering coefficients of an L-junction of two SHELL181based cross-ply laminated plates for various incident waves plotted with respect to the wave number component k_x at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 114
- 4.7 Energy scattering coefficients of an L-junction of two SOLID185based cross-ply laminated plates for various incident waves plotted with respect to the wave number component k_x at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 115
- 4.8 Energy scattering coefficients of an L-junction of two SHELL181based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 0 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. . 116
- 4.9 Energy scattering coefficients of an L-junction of two SOLID185based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 0 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 116

- 4.10 Energy scattering coefficients of an L-junction of two SHELL181based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 5 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 117
- 4.11 Energy scattering coefficients of an L-junction of two SOLID185based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 5 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 118
- 4.13 Energy scattering coefficients of an L-junction of two SHELL181based angle-ply laminated plates for various incident waves plotted with respect to the wave number component k_y at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 121
- 4.14 Energy scattering coefficients of an L-junction of two SOLID185based angle-ply laminated plates for various incident waves plotted with respect to the wave number component k_y at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3. 122

4.15	Energy scattering coefficients of an L-junction of two SHELL181-
	and SOLID185-based angle-ply laminated plates for various in-
	cident waves plotted with respect to the frequency at the fixed
	wave number component $k_x = 0 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical
	scattering coefficients obtained using the method presented in
	Chapter 3
4.16	Energy scattering coefficients of an L-junction of two SHELL181-
	and SOLID185-based angle-ply laminated plates for various in-
	cident waves plotted with respect to the frequency at the fixed
	wave number component $k_x = 5 \text{ m}^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical
	scattering coefficients obtained using the method presented in
	Chapter 3
5.1	An example of an electric vehicle gearbox being meshed us-
	ing two-dimensional shell elements. Colours represent elements
	with different thicknesses
5.2	A schematic representation of the boundary mapping procedure
	in the triangular mesh element. Green ovals depict wave vector
	curves. The orientation of the principal material axes is defined
	in the element's coordinate system (x_g, y_g, z_g) . A ray starting at
	the position s of edge a points in the direction of the outgoing
	group velocity vector \mathbf{c}_g^+ . The group velocity vector is defined
	by the momentum component p_{\shortparallel} . The ray hits edge a' at the
	position s' with the new momentum component p'_{\shortparallel} defined in
	the coordinate system (e'_x, e'_y) . However, momentum vectors
	\mathbf{p} and \mathbf{p}' are equal

5.3	A representation of the geometrical relations used for computa-
	tion of the position integral in Equation (5.19) . A ray emitting
	from edge a can only hit edge a' if its direction angle $\alpha^+ > \xi^-$.
	$D(s, s')$ is the ray length. $s_{\min(\max)}$ is the minimum (maximum)
	value of the position s in the local coordinate system (e_x, e_y)
	such that a ray with the direction angle α^+ reach edge a' at
	some position s'
5.4	Geometrical relations between differentials $d\phi$ and $d\alpha^-$. Rays
	emitting from the point source at \mathbf{r}_0 intersect edge a at the
	incoming angles α^- . Similarly as in the case considered in Fig-
	ure 5.3, rays can only reach edge a if their direction angles
	$\alpha^- \in [\xi^-, \xi^+]. \dots \dots \dots \dots \dots \dots \dots \dots \dots $
5.5	Partitioning of the typical wave vector curve (depicted by green
	ovals) between edges of the mesh element. Red dashed arcs
	represent $\omega_a^{-1}(\omega_0)$ in Equation (5.36). They are defined by im-
	posing the outgoing group velocity vector angle α^+ to be in
	range $[\xi^-, \xi^+]$, so that outgoing rays from edge <i>a</i> can reach the
	point r
5.6	Partitioning of the shear wave vector curve (depicted in green)
	between edges of the mesh element. Red dashed arcs represent
	$\omega_a^{-1}(\omega_0)$ in Equation (5.36). They are defined by imposing the
	outgoing group velocity vector angle α^+ to be in range $[\xi^-, \xi^+]$,
	so that outgoing rays from edge a can reach the point \mathbf{r} 140
5.7	A polygonally shaped semi-infinite plate meshed with two-dimensional
	triangular shell elements. An edge and a point sources are as-
	sumed to be applied on the plate on the vertical and horizontal
	edges, respectively. The diagonal edge is free, and rays hitting
	this edge reflect back into the plate. Reflection of rays at other
	edges is prohibited thus imitating "infinity" of the plate through
	these edges

5.8	A geometrical illustration of the ray paths (shown in red) start-	
	ing from the edge source at the vertical edge. Wave vector	
	curves (shown in green) are a circle for the isotropic material	
	case (a) and an oval for the composite material case (b). \ldots	144
5.9	The mean acceleration response levels of the polygonally shaped	
	semi-infinite isotropic plate due to an edge source as a function	
	of the number of Legendre polynomials N_m . The acceleration	
	is given in m/s^2	145
5.10	The mean acceleration response levels of the polygonally shaped	
	semi-infinite composite plate due to an edge source as a function	
	of the number of Legendre polynomials N_m	147
5.11	The intensity vector field of the isotropic (a,c) and composite	
	(b,d) plate cases arising due to an edge source at the vertical	
	edge as a function of the number of Legendre polynomials used	
	in the basis set	148
5.12	The mean acceleration response levels of the polygonally shaped	
	semi-infinite isotropic plate due to an edge source as a function	
	of the number of Legendre polynomials N_m . The acceleration	
	is given in m/s^2	149
5.13	The mean acceleration response levels of the polygonally shaped	
	semi-infinite isotropic plate due to an edge source as a function	
	of the number of Legendre polynomials N_m . The acceleration	
	is given in m/s^2	150
5.14	An L-shaped composite plate meshed with two-dimensional tri-	
	angular shell elements. A point source is assumed to be applied	
	on the middle of the ground plate	151

	5.15	The mean DEA and FEM acceleration response levels of an
		L-shaped composite plate due to a harmonic point excitation
		force of 1 N at the frequency $f = 3150$ Hz. FEM results are
		frequency-averaged over one-third octave band with the centre
		frequency $f = 3150$ Hz. The acceleration is given in m/s ² 152
	5.16	The intensity vector field of an L-shaped composite plate arising
		due to a point source at the middle of the ground plate. Vector
		colouring represents intensity amplitudes
	5.17	The FE meshes of the full gearbox housing (a) and the gear-
		box housing without stiffeners along the longitudinal axis of the
		cylindrical part (b)
	5.18	The mean DEA and FEM acceleration response levels of the
		gearbox housing of an electric vehicle powertrain due to a har-
		monic point excitation force of 1 N at the frequency $f = 8000$ Hz.
		The acceleration is given in mm/s^2
	5.19	The intensity vector field of the gearbox housing with stiffen-
		ers (a) and without stiffeners (b) arising due to a point source
		at its front side. The structure colouring represents the mean
		acceleration levels of the bending mode
	5.20	Two modification methods of the FE model of the non-stiffened
		structure to obtain the same structural mass as the full struc-
		ture. The first one (a) is to compute and assign an effective
		value of material density to the cylindrical part 2, which is dif-
		ferent from 1, where the structure is made of aluminium. The
		second one (b) is to compute and assign effective thickness of
		values of shell elements in parts 1 and 2 by calculating the to-
		tal volume of removed stiffeners and keeping the same material
		density

	1 The nodes of the FE mesh, where the acceleration response	5.21
	amplitudes were computed. Input node and nodes 1, 2 and 3 $$	
	are located on the non-stiffened part, whereas nodes 4 and 5 are	
. 157	on the stiffened part of the structure	

Chapter 1 Introduction

In this introduction, we describe the problem of interest and provide motivation for the work presented in this thesis. Further on, we present a review of existing literature and discuss relevant mathematical methods. Finally, we outline the structure of the thesis, summarising key ideas and results from each chapter.

1.1 Motivation and literature review

Composites are nowadays widely used within many transport sectors such as aerospace, automotive and naval architecture industries [1, 2]. In comparison to isotropic materials such as aluminium and stainless steel, composites provide comparable stiffness and strength whilst being significantly lighter [3]. Furthermore, the mechanical characteristics of fibre-reinforced composites can be tailored to suit particular purposes [1, 4]. Over the past several decades, such advantages of composites have led to growing usage of them to construct primary structural components in the aerospace and automotive industries.

However, despite their superior structural characteristics, composites exhibit reduced vibro-acoustic performance levels due to the large variety of waves propagating in them. Modelling noise and vibration in composite structures play an essential role in the various stages of their life cycle. For example, at the design phase of a vehicle, vibro-acoustic analysis can help figure out zones potentially prone to excessive noise. At the post-built stage, non-destructive testing techniques help monitor and assess structures' performance to avoid sudden failure. Therefore, there is a need for quick and accurate methods of evaluating the vibrational response of composite structures.

1.1.1 Review of high-frequency methods

Structural vibrations of a complex structure are in general modelled using deterministic schemes such the Finite Element (FE) [5–7], Finite Difference (FD) [8, 9] or Boundary Element (BE) methods [10-12]. These methods are particularly useful in providing the full phase and amplitude information of the wave field in the low-frequency regime. However, at higher frequencies, these methods become inefficient and computationally expensive. This is because wavelengths are much smaller than the characteristic dimensions of the structure in the high-frequency range. Many degrees of freedom are required to accurately represent the wave field on the wavelength scale.

At higher frequencies, statistical approaches, energy and ray-tracing methods are favoured. The main statistical approach used extensively in the industry today is Statistical Energy Analysis (SEA) [13–16]. This method is based on several assumptions that are often hard or not possible to validate *a priori* [17, 18]. The structure is divided into subsystems, in which wave fields are assumed to be diffuse. SEA computes the steady-state averaged energy levels in subsystems. Since wave fields in anisotropic media are non-diffuse [19–22], the application of SEA to composite structures might be significantly errorprone [23].

To account for non-diffuse vibration fields and to represent the energy distribution locally in subsystems, several energy methods such as Wave Intensity Analysis (WIA) [24, 25], Energy Finite Element Analysis (EFEA) [26–30] and radiosity method [31–34] have been developed over the past several decades. The WIA and EFEA methods both rely on the reverberant wave field assumption; that is, the vibrational response of the structure is assumed to be a superposition of plane waves. Therefore, in the presence of strong sources and heavy damping condition, both methods underestimate wave energy lev-

els around the excitation points and overestimate the energy levels in the far field [35-37], since the direct field is dominant over the reverberant part in such cases. In EFEA, energy flow is required to be equally probable in all directions, and the reflection of energy at boundaries follows Lambert's law [35, 37, 38]. In other words, coupling between subsystems is represented by angle-averaged energy scattering coefficients - these are called coupling loss factors in SEA. Another limitation of the EFEA method concerns the vibrational response of two-dimensional systems such as plates and membranes. The far field solution is proportional to $1/\sqrt{r}$, where r is the distance between source and receiver, whereas it must be proportional to 1/r in such systems. Other limitations and the validity region of the EFEA method have been presented in several works [35-40]. Nevertheless, the EFEA method has been applied to calculate the vibrational response of composite structures [41]. The radiosity method is technically a ray-tracing technique, and it is a special case of the radiative transfer method in vibro-acoustics [33]. However, this method also employs Lambert's law to represent the directivity of energy reflection at boundaries.

The ray-tracing methods include ray- and beam-tracing techniques [42–47], the radiative transfer method [33, 47–49] and Dynamical Energy Analysis (DEA) [50–53]. Unlike diffusive methods, ray-tracing methods do not employ the statistical law of energy reflection distribution at boundaries. Instead, waves are treated as uncorrelated rays that carry energy along the structure and reflect according to Snell's law. Full ray paths can be computed once an initial source point and a ray direction are provided. Furthermore, one can retain full amplitude and phase information in the ray- and beam-tracing techniques, whereas no phase information is included in SEA and diffusive methods. This is important when considering transient time-dependent sources [43, 54]. However, proper ray- and beam-tracing techniques often are much more computationally expensive than SEA and diffusive methods [43, 45, 55]. In fact, the wave energy density at a certain point r is computed by summing over contributions from all rays that reach the point r starting from the source points. Therefore,

to guarantee the sufficient density of rays at the point r, one would need to increase the number of rays to a value that is not determined a priori or by any analytical methods, but rather empirically [50, 56].

The radiative transfer and DEA methods are based on the integral equation for the density of propagated rays. The phase information is no longer retained as in SEA and diffusive methods, and only spatial- and (or) frequency-averaged energy amplitudes are computed. Nevertheless, the deterministic law of reflection is retained as in the pure ray-tracing method. In other words, detailed information about waves' reflection and transmission behaviour at structural discontinuities is required and routinely used.

The radiative transfer method is based on the *Huygens principle* [32]: the vibrational field is assumed to be a superposition of the direct field arising from actual sources and the reverberant field emerging from fictive sources placed at the boundary. Consequently, by expressing the energy density and intensity as a sum of the direct and fictive source contributions, a Fredholm integral equation of the second kind is postulated for the fictive sources [33]. Diffuse and specular types of energy reflection for the determination of fictive sources have been considered in [32, 39, 57] and [33, 49], respectively. The former case reduces the radiative transfer method to the radiosity method and subsequently to the SEA method. The resulting equation for the fictive sources is solved numerically by the BE method for a limited number of cases. The correspondent software is called "CeReS" [39]. It is mentioned by Le Bot in [39] that only the case of diffuse energy reflection is implemented in the software, thus limiting its practical application to homogeneous isotropic structures. Moreover, the specular reflection case yields the functional equation, which depends on the contributions from a possibly infinite number of ray paths, thus increasing the practical complexity of the method [49].

Finally, the DEA method is a recently developed technique to evaluate the mean vibrational response of complex structures in high-frequency regimes. This method is implemented directly on the FE meshes [58, 59], thus avoiding

time-consuming remodelling. The main idea is similar to the Huygens principle considered in the radiative transfer method. Namely, the wave energy density is represented as a sum of direct and reverberant field contributions. In contrast to the radiative transfer method, the reverberant part is a solution of the Liouville equation [51] and is expressed in terms of an integral transfer operator T relating densities of rays between mesh cell boundaries as

$$\rho_{\infty}(\mathbf{r}, \mathbf{p}) = \sum_{n=1}^{\infty} T^n \rho_0(\mathbf{r}, \mathbf{p}) = (1 - T)^{-1} \rho_0(\mathbf{r}, \mathbf{p}).$$
(1.1)

Infinite ray reflection events are naturally obtained by the above sum, and in dissipative systems, this yields a linear system of equations relating the initial stationary ray densities on the union of all mesh cell boundaries $\rho_0(\mathbf{r}, \mathbf{p})$ and $\rho_{\infty}(\mathbf{r}, \mathbf{p})$, respectively. Note that the ray densities are expressed in phase space; that is, the position \mathbf{r} and the momentum vectors \mathbf{p} are considered as independent variables.

The DEA method has been successfully applied to model the mean vibrational response of a stiffened double-hull structure and a shock tower of a vehicle [59], the floor structure of caravan cars [60, 61], an agricultural tractor [53, 62] and a vehicle cavity [63]. Still, as the radiative transfer method, the DEA method has been developed and applied only for isotropic structures. This motivates the development of an extension of the DEA method for composite structures, and it is one of the primary concerns of this thesis.

1.1.2 Review of methods for evaluation of wave propagation characteristics

To accomplish this goal, one would need to compute wave propagation characteristics such as dispersion relations, the group velocity vector and scattering coefficients at discontinuities of composite structures.

Dispersion relations

Dispersion relations play an important role in many parts of the mathematical modelling of the wave propagation problem. They can be computed by solving governing equations of harmonic motion for composite laminates obtained using analytical approaches such as the Equivalent Single Layer (ESL) [64–68] and the layerwise theories [69–73] or numerical methods such as the spectral element (SE) [74–78] and the wave finite element (WFE) [79–85] methods. Extensive literature reviews of different analytical and numerical methods for composite laminated plates can also be found in [86–88].

The ESL theories include the Classical Laminated Plate (CLP) theory, Firstorder Shear Deformation Laminated (FSDL) and Higher-order Shear Deformation Laminated (HSDL) theories. In these theories, three-dimensional elasticity equations of a laminated plate are reduced to a two-dimensional problem using the Kirchhoff-Love or shear deformation hypotheses.

The CLP theory neglects the transverse shear strain, which can lead to substantial estimation errors for thick laminates and sandwich plates [89–92]. Nevertheless, for thin plates, this theory is quick and simple in providing reasonable approximate solutions of elastic problems and dispersion curves of thin laminates [64, 65, 93, 94].

The FSDL theory is an improved version of the CLP theory for thick laminated and sandwich plates. The main difference is that the transverse shear strain takes a non-null constant value in the thickness direction. However, to account for correct representation of the transverse shear stress, the shear correction factors that vary for different laminates are included in this method, [95–97]. Ghinet and Atalla [68] computed dispersion relations of waves propagating in sandwich composite plates using the FSDL theory. The sandwich plate considered in this work were made of transversely isotropic skins and isotropic rigid foam core, thus making the plate to be effectively isotropic. Dispersion relations for heading directions of 0° , 45° and 90° were presented in [98] for a symmetrically laminated composite made up of seven layers. Mejdi and Atalla [99] extended this approach to laminate panels stiffened by composite stiffeners and computed their vibration and acoustic responses.

The HSDL theories are based on the non-linear variation of transverse strain

and stress in the thickness direction. They include several different formulations of the mid-plane displacement field as a function of the thickness direction variable. The Third-order Shear Deformation Laminated (TSDL) theory has been developed by Reddy [67, 100]. It was applied for computing the deflection, the buckling load and natural frequencies of composite laminated plates in [67, 101, 102].

In contrast to ESL methods, all layers of a laminate are treated individually in the layerwise theories. Consequently, the number of unknown variables depends on the number of layers, which can yield large complex models, thus increasing the computational cost. In [71, 98], the Discrete Laminate (DL) theory was applied to obtain dispersion curves of a curved sandwich composite panel, and the results were compared with the FSDL theory estimations. Ghinet and Atalla [103] utilised the DL method to calculate the vibration response of a multi-layer construction consisting of composite and viscoelastic layers.

The SE method relies on constructing the exact dynamic stiffness matrix from the general solution of governing differential equations in the frequency domain. The response is assumed to be the superposition of wave modes at different frequencies based on Discrete Fourier Transformation theory. The Fast Fourier Transform can be applied to get the solution in the time domain. An excellent review of the method and its applications can be found in [74, 104]. Most of the literature on the SE method concerned one-dimensional waveguides such as beams and rods. These waveguides are assumed to be homogeneous and extended to infinity in one direction with arbitrary cross-sections. The dispersion curves of composite beams were obtained and analysed using spectral finite elements in [105–107]. Two-dimensional waveguides such as plates were considered as well; however, several assumptions either on boundary conditions (simply supported or cantilever) or on the material of plates were put so that one could reduce the two-dimensional waveguide problem to a onedimensional beam-type problem. Datta et al. [75] considered composite plates with transversely isotropic laminas having the symmetry axis aligned with either one of the plane dimensions, thus allowing to separate the problem into two one-dimensional beam waveguide problems. Dispersion relations for a composite plate consisting of four layers were computed. A similar approach was employed by Mukdadi et al. [77, 108], in which the dispersion relations of a layered plate with rectangular cross-section were obtained. Moreover, the SE method was also applied to an infinite composite plate for the first time, and dispersion curves with directional dependence over the plane dimensions were obtained. Recent works considered the wavelet spectral finite element method to account for short waveguides, where the effect of boundaries was included in the model [109–112].

The WFE method is a technique to study wave motion in periodic structures. For such structures, the dynamic vibro-acoustic behaviour of the whole structure can be described through the analysis of a single period [113]. A periodic cell is modelled using conventional FE methods. Mass, damping and stiffness matrices thus obtained are used to construct the dynamic stiffness matrix. Periodic structure theory is then applied, and the eigenvalue problem is postulated which eigenvalues and eigenvectors can be used to calculate dispersion relations and wave modes, respectively. Moreover, wave modes can be used to compute the reflection/transmission coefficients of joints in coupled structures. One of the main advantages of this method is that since only one periodic segment is used, the size of the WFE model does not depend on the dimensions of the waveguide, and the computational cost of the method is low. In addition, since a single periodic cell is discretised using conventional FE matrices, the full potential of existing conventional FE tools can be exploited. The WFE method was originally proposed by Mead in [114], where he studied the harmonic wave propagation in one-dimensional periodic systems. A huge contribution to the analysis of wave propagation in various periodic structures using a FE model of a single periodic section has been made by Abdel-Rahman [115]. The free wave propagation in one-dimensional isotropic and composite waveguides was analysed by Mace et al. in [80] and Duhamel et al. in [116]. Dispersion relations of waves were computed using one-dimensional WFE method for structures such as stiffened cylinders [117], car tyre [118], thin-walled structures [119, 120], inhomogeneous cylindrical [121] and fluid-filled pipes [122, 123], sandwich beams and panels [124, 125] and laminated cylinders [126]. Numerous applications of the one-dimensional WFE method can be found in the PhD work of Waki [84]. Also, Waki [127] outlined several numerical issues of the WFE method and how to alleviate errors associated with these issues. It is worth mentioning a work of Mencik and Ichchou [128], where they suggested a substructuring technique to address the problem of numerical issues for the case of multi-layered structures.

The basics of the WFE method for two-dimensional periodic systems was also presented in the work of Mead [114]. Later, Mead and Parthan [79] showed how the problem of defining the dispersion relations in the general direction over the plate's plane dimensions could be reduced to an array of one-dimensional WFE problems with varying periodic distances. This approach has been recently used by Chronopoulos [129] to compute the wave slowness and group velocity curves of an orthotropic graphite-epoxy monolithic plate. Rigour mathematical models of the WFE method for two-dimensional periodic isotropic and composite systems have been developed by Manconi and Mace [82, 83, 130]. Several forms of the eigenvalue problem that lead to the computation of dispersion relations were postulated. Alimonti et al. [131] extended these works by presenting a contour integral method to compute the non-linear eigenvalue problem arising from governing equations of motion upon fixing frequency and propagation direction. Dispersion relations were computed for two-dimensional arbitrarily thick layered panels in [132, 133] and periodic textile composites in [134].

Group velocity

Calculation of the group velocity vector is essential for determining the directions of rays carrying the wave energy. Many authors calculated the group velocity for one- and two-dimensional systems using various FE-based approaches. For example, Mace et al. [80], and later Ichchou et al. [135] considered energy velocity, the ratio between time-averaged energy flow and total energy density, utilising the fact that energy velocity is equal to group velocity in undamped systems. They expressed energy velocity in terms of wave modes, stiffness and mass matrices and the length of a periodic unit cell of one-dimensional systems. Ichchou et al. [135] also computed group velocity by applying a finite central difference scheme to the definition $\frac{\partial \omega}{\partial k}$. It is required that dispersion curves are categorised into distinct propagating branches. This drawback was outlined as well by Finnveden [136]. In his work, the group velocity is expressed in terms of the derivatives of the spectral form of the equations of motion. A similar approach has been recently proposed by Cicirello et al. [137]. They have analysed the first and second-order sensitivity of the general eigenvalue problems, including ones as in Equations (4.6) and (4.8), thus expressing group velocity in terms of wave number sensitivities. However, all referenced works considered one-dimensional systems, and the group velocity scalars are only obtained along the x or y directions, which is not sufficient to define the group velocity field in two-dimensional systems. Langley [138], and later Wang et al. [139] derived the same expressions for group velocity vector components for two-dimensional curved shells and laminated plates, respectively. The energy skew angle, that is, the angular divergence between group and phase velocities, was calculated using geometric considerations of wave vector curves and sensitivity analysis of governing equations of motion in [129]. Chronopoulos et al. [140] studied the sensitivity of propagating waves in two-dimensional composite plates and computed $\frac{\partial \omega}{\partial k}$ for a given set of wave number component (k_x, k_y) . However, the component $\frac{\partial \omega}{\partial k}$ is only the projection of the group velocity vector on the wave vector [19, 21, 138]. Zhao et al. [141] computed group velocity vectors of Lamb waves in unidirectional and angle-ply composite plates using the 3D elasticity theory.

Scattering coefficients

Most of the research on the calculation of scattering coefficients focused on isotropic plates so far, such as early work done by Cremer et al. [142] considering the structure-borne sound transmission of a flexural wave for rightangled joints of thin plates. Later, Craven and Gibbs [143, 144] and Wöhle et al. [145, 146] included the in-plane wave modes in their analyses together with considering up to four plates joined together. Langley and Heron [147] computed scattering coefficients for structural junctions connecting an arbitrary number of thin, isotropic plates along a rigid beam. A simplified treatment using a line-junction approximation is also described in [147]; that is, boundary conditions and force-balance equations are considered along a 1D line at the centre of the junction only. Mace [148] demonstrated important properties of the scattering coefficients such as reciprocity and conservation of energy on the example of several rods attached to a junction member. Mees and Vermeir [149] analysed the bending wave transmission loss in the system of plates connected by a hinge or by an elastic interlayer. McCollum and Cuschieri [150] studied the flexural behaviour of right-angled thick finite plates using a mobility power flow approach. They also considered both in-plane and out-of-plane wave scattering in right-angled thick semi-infinite plates [151]. Langley investigated wave reflection and transmission coefficients for structural junctions between curved panels and beams [138]. The occurrence of the negative group velocity phenomenon in cylindrical structures has been outlined in this work. Skeen and Kessissoglou [152] computed transmission coefficients for finite and semi-infinite coupled plate structures. Mencik and Ichchou introduced the hybrid FE/WFE method for calculating reflection and transmission coefficients for one-dimensional waveguides coupled longitudinally [81]. In recent years,
this method has been developed and extended to other types of junctions [153–158] and to two-dimensional waveguides [159, 160] However, structures considered in those works mainly were isotropic or were consisting of layers of isotropic materials.

Beyond the case of wave propagation in isotropic materials, Bosmans et al. [93] studied the scattering properties of orthotropic plate junctions with principal material axes aligned with the plate coordinates, that is, so-called *specially* orthotropic plates. However, no details on the derivation are given, and results are presented only for the particular case of bending wave transmission loss in right-angled plates, so-called L-junctions. It is not clear whether the approach derived in [93] is limited to specially orthotropic plates or can be extended to an arbitrary number of plates meeting at the junction with different orientation of the principal material axes. Moreover, detailed information on the reflection/transmission behaviour of all propagating modes at complex junctions is needed for a DEA treatment. This includes information about the angle-of-incidence dependence of scattering coefficients and mode conversion. Karunasena and Shah [161] studied reflection of guided waves at the region of bonding material between two composite plates using the hybrid FE and semi-analytical FE method. Lee et al. [162] have presented the scattering coefficients of coupled composite plates with joint compliance and damping using the First-Order laminated plate theory [67]. Again, the principal material axes of laminates considered in these works are aligned with the plate coordinates, effectively reducing the complexity of underlying governing equations. Furthermore, in [162], the shear correction factor is introduced to correct transverse shear stiffness in the laminate, which must be defined for each laminate separately. Chronopoulos [163] computed scattering coefficients at the damaged junction between two composite beams. Later, Apolowo and Chronopoulos [164] computed the scattering coefficients of two multi-layer composite plates coupled longitudinally to localise the structural damage in the context of Structural Health Monitoring. An attempt of extending the work of Mitrou et al.

in [153] to composite plates has been made by Mitrou and Renno in [165]. The results were not reliable as the energy scattering coefficients did not sum to unity as expected in lossless systems [153, 160].

Therefore, there is a need for a detailed derivation of reflection and transmission matrices for waves travelling in structural junctions connecting composite plates at arbitrary angles and without any restrictions on the orientation of principal material axes both with respect to the orientation of the junction and with respect to the orientation in different plates. This is done following two approaches. The first one is extending the work of Langley and Heron [147] to composite laminated plates utilising the CLP theory. The second one is extending the work of Mitrou et al. [160] to arbitrarily layered composite plates. It is important to note that in both approaches, the scattering coefficients are obtained by working in the infinite junction approximation, i.e., solving the wave problem for an incident plane wave assuming that the plates extend to infinity along the junction. This approximation is justified in the high-frequency range by the *locality principle*, that is, local vibrational behaviour depends on local properties such as geometry and material properties [39].

1.2 Aims and objectives

This thesis aims to develop an extension of the DEA method for composite structures that can be represented by two-dimensional finite shell elements. The following specific goals are put in place to achieve this aim, namely:

- to develop a semi-analytical method for calculating scattering coefficients at line junctions between composite laminated plates.
- to form a hybrid FE and WFE method for computing reflection and transmission matrices of general junctions of composite laminated plates.
- to analyse and compare the effectiveness and applicability of the two methods above.

• to extend the DEA method for composite structures and compare results with FE method simulations.

1.3 Thesis outline

Chapter 2 aims to present the main theoretical basics of the DEA method for two-dimensional isotropic structures. The solution of Helmholtz and biharmonic wave equations are presented. Furthermore, we identify and discuss the main modifications needed to allow for DEA application on composite structures. Also, we describe the CLP theory for thin laminated plates, which is used in Chapter 3 to set up the governing equations of motion for composite plates. Finally, we review the main equations of the WFE method for composite plates. These are used further in Chapter 4 to derive expressions for scattering coefficients using the hybrid FE/WFE method.

Chapter 3 describes in all generality how to compute energy scattering coefficients of structural junctions made up of thin composite laminated plates in the line junction approximation. Expressions quantifying transmission and reflection coefficients as a function of the frequency and the wave number component k_x are derived. Interesting phenomena such as negative refraction and negative group velocity are observed and analysed. Furthermore, an effective scattering matrix for a plate with multiple finite stiffeners attached to it is computed. The scattering coefficients are computed explicitly for examples of two and three composite plates joined together in an L and T geometry.

A hybrid FE/WFE model that predicts the scattering properties for different junctions of two-dimensional anisotropic composite plates is developed in Chapter 4. The influence of the angle of incidence and the frequency on the distribution of the power flow of incident bending, shear and longitudinal type waves is investigated. A detailed comparison with semi-analytical evaluations of scattering coefficients derived in Chapter 3 is presented. The method gives for the first time a detailed recipe for computing scattering coefficients for the generic case of an arbitrary number of composite plates connected at a junction without restrictions on the angles at which the plate meet or the orientation of the principal axis of individual plates.

In Chapter 5, we present the modified theoretical base of DEA for composite structures. The findings of Chapters 3 and 4 are used to derive the stationary wave energy density arising in the structure due to a point or an edge sources. Numerical results for the cases of a polygonally shaped plate, an electric vehicle gearbox and an L-shaped composite plate are presented.

Finally, Chapter 6 presents concluding remarks with revision of contributions of the work and suggests further potential research.

Chapter 2 Background

In this chapter, we present the background theory needed to proceed with an understanding of the main findings of this thesis presented in Chapters 3, 4 and 5. Section 2.1 describes the basics of the DEA method for two-dimensional isotropic structures. Also, the wave problems that are considered throughout this thesis are reviewed. Furthermore, the main parts of the DEA method that need to be modified to allow for its application to composite structures are identified and discussed. In Section 2.2, we describe the CLP theory for thin laminated plates. It is used in Chapter 3 to set up the governing equations of motion for composite plates, which eventually lead to the computation of the scattering coefficients of various junctions. Finally, Section 2.3 considers the main principles and equations in the WFE method for composite plates modelled using two-dimensional and three-dimensional finite elements. The equations presented in this section will be used further in Chapter 4 to derive expressions for scattering coefficients using the hybrid FE/WFE method.

2.1 Dynamical Energy Analysis for isotropic structures

In this section, we derive the main equations in relation to the DEA method. First, we consider general linear wave equations and show how they can be solved using a sum of the free-space Green functions and the solution of the homogeneous equations in subsection 2.1.1. In subsection 2.1.2, the free-space Green's functions for the Helmholtz and biharmonic wave equations are computed. In subsection 2.1.3, the solution of the homogeneous wave equation is obtained in terms of ray trajectories via high-frequency or so-called *Eikonal* approximation. Subsection 2.1.4 shows that the resulting mean wave energy density can be approximated by the Liouville equation, which is rewritten in terms of the boundary integral operator. The general algorithm of the DEA method is discussed at the end of the section.

2.1.1 Wave equation

We consider stationary problems with continuous monochromatic energy sources. In other words, we assume that linear wave equations in question are not explicitly time-dependent, and driving terms are time-harmonic with a fixed angular frequency ω . This allows us to operate in the frequency domain rather than in the time domain. The general problem of determining the response of a two-dimensional system to a time-harmonic force with the amplitude F_0 at a source point $\mathbf{r}_0 \in \Omega$ can be written as

$$\left(\hat{H} + \omega^2\right) G(\mathbf{r}, \mathbf{r}_0, \omega) = -F_0 \,\delta(\mathbf{r} - \mathbf{r}_0) \,, \quad \mathbf{r} \in \Omega \subset \mathbb{R}^2 \,, \tag{2.1}$$

where \hat{H} is the linear combination of partial-differential operators with constant coefficients, G is the Green function, and δ is the Dirac delta function [50, 51]. We seek solutions $G(\mathbf{r}, \mathbf{r}_0, \omega)$ in the domain Ω with the boundary Γ . If the linear wave operator \hat{H} is equal to $c^2\Delta$ with Laplacian operator Δ and positive c denoting the wave velocity, also called the phase velocity, then Equation (2.1) is the inhomogeneous Helmholtz equation. With the appropriate form of the wave velocity c, this equation models vibrations of thin membranes with clamped edges and in-plane deformations of plates due to point source [142, 166, 167]. In such cases, Equation (2.1) can be rewritten as

$$\left(\Delta + k^2\right) G(\mathbf{r}, \mathbf{r}_0, \omega) = f_0 \,\delta(\mathbf{r} - \mathbf{r}_0) \,, \quad \mathbf{r} \in \Omega \,, \tag{2.2}$$

where $k = \frac{\omega}{c}$ denotes the wave number, and $f_0 = -\frac{F_0}{c^2}$ is the corresponding forcing term.

Another important application of Equation (2.1) is the bending motion of a thin isotropic plate in the Kirchhoff-Love plate theory context. The corresponding linear wave operator \hat{H} is equal to $-\frac{D}{\rho h}\Delta^2$, and the equation takes the following form

$$\left(\Delta^2 - k^4\right) G(\mathbf{r}, \mathbf{r}_0, \omega) = f_0 \,\delta(\mathbf{r} - \mathbf{r}_0) \,, \quad \mathbf{r} \in \Omega \,, \tag{2.3}$$

where the Green function G now denotes the out-of-plane displacement amplitude, $k^4 = \frac{\rho h}{D}\omega^2$ and $f_0 = \frac{F_0\rho h}{D}$. Here, ρ is the density of the plate, h is the plate thickness, and $D = \frac{Eh^3}{12(1-\nu^2)}$ is the bending stiffness with Young's modulus E and Poisson ratio ν [1, 142].

Equations (2.2) and (2.3) are supplied with boundary conditions at Γ . These can be Dirichlet, Neumann or mixed Robin conditions for the former case and clamped, free and simply supported boundary conditions for the latter case. The solution of these problems can be found by splitting *G* into a homogeneous and an inhomogeneous part as

$$G(\mathbf{r}, \mathbf{r}_0, \omega) = G_0(\mathbf{r}, \mathbf{r}_0, \omega) + G_h(\mathbf{r}, \mathbf{r}_0, \omega), \qquad (2.4)$$

where G_0 is the free-space Green function, and G_h is a function which satisfies the homogeneous equation

$$\left(\hat{H} - \omega^2\right) G_h(\mathbf{r}, \mathbf{r}_0, \omega) = 0, \mathbf{r} \in \Omega, \qquad (2.5)$$

and its amplitude is determined by the boundary conditions [51, 168]. The function G_0 describes contributions arising directly from the source, whereas G_h represents the scattering at the boundary Γ . In the next section, we will compute the free-space Green functions for the Helmholtz and biharmonic wave equations.

2.1.2 The free-space Green function

The free-space Green function is defined within an unbounded domain, and it can be obtained by the application of Fourier transform techniques [168, 169]. Since we consider isotropic structures made of two-dimensional elements, Helmholtz and biharmonic wave equations (2.2) and (2.3) hold for r in the whole two-dimensional space \mathbb{R}^2 . Without loss of generality, we assume that the input forcing term f_0 is equal to unity. By imposing the Sommerfeld radiation condition to ensure that only outgoing waves are present at infinity [170], one can express the free-space Green function of the Helmholtz equation $G_0(\mathbf{r}, \mathbf{r}_0, \omega)$ as

$$G_H(\mathbf{r}, \mathbf{r}_0, \omega) = -\frac{i}{4} H_0^{(1)}(k|\mathbf{r} - \mathbf{r}_0|) \quad , \quad H_0^{(1)}(z) = \frac{1}{\pi i} \int_{-\infty}^{\infty} e^{iz \cosh u} du \,, \quad (2.6)$$

where i is the imaginary unit, and $H_0^{(1)}$ is the Hankel function of the first kind and zero order. The asymptotic behaviour for $\omega \to \infty$ is then given by

$$G_H(\mathbf{r}, \mathbf{r}_0, \omega) \sim -\sqrt{\frac{2}{\pi k |\mathbf{r} - \mathbf{r}_0|}} e^{i(k|\mathbf{r} - \mathbf{r}_0| - \frac{\pi}{4})}.$$
 (2.7)

For the case of the biharmonic wave equation, the free-space Green function satisfies

$$\left(\Delta^2 - k^4\right) G_B(\mathbf{r}, \mathbf{r}_0, \omega) = \left(\Delta - k^2\right) \left(\Delta + k^2\right) G_B(\mathbf{r}, \mathbf{r}_0, \omega) = \delta(\mathbf{r} - \mathbf{r}_0), \mathbf{r} \in \mathbb{R}^2.$$
(2.8)

Now, the Green function G_B can be written as $G_B = -\frac{1}{2k^2} (G_H - G_M)$ [167, 171], where G_M is the Green function of the modified Helmholtz equation

$$\left(\Delta - k^2\right) G_M(r, r_0, \omega) = \delta(\mathbf{r} - \mathbf{r}_0), \quad \mathbf{r} \in \mathbb{R}^2.$$
(2.9)

Its solution can be readily obtained from Equation (2.6) by replacing k with ik yielding

$$G_M(\mathbf{r}, \mathbf{r}_0, \omega) = -\frac{i}{4} H_0^{(1)}(ik|\mathbf{r} - \mathbf{r}_0|) = \frac{1}{2\pi} K_0(k|\mathbf{r} - \mathbf{r}_0|), \qquad (2.10)$$

with K_0 being the modified Bessel function of the second kind.

Finally, the free-space Green function of the biharmonic wave equation can be

written as

$$G_B(\mathbf{r}, \mathbf{r}_0, \omega) = \frac{1}{2k^2} \left(\frac{i}{4} H_0^{(1)}(k|\mathbf{r} - \mathbf{r}_0|) - \frac{1}{2\pi} K_0(k|\mathbf{r} - \mathbf{r}_0|) \right) , \qquad (2.11)$$

and the corresponding high-frequency asymptotic form reads

$$G_B(\mathbf{r}, \mathbf{r}_0, \omega) \sim \frac{1}{8k^2} \sqrt{\frac{2}{\pi k |\mathbf{r} - \mathbf{r}_0|}} \left(e^{i(k|\mathbf{r} - \mathbf{r}_0| - \frac{\pi}{4})} - e^{-k|\mathbf{r} - \mathbf{r}_0|} \right) .$$
(2.12)

2.1.3 Eikonal approximation and ray tracing

Having computed the free-space Green functions, we can concentrate on the homogeneous part G_h . For the sake of simplicity, we operate with the homogeneous Helmholtz equation

$$(\Delta + k^2) G_h(\mathbf{r}, \omega) = 0, \mathbf{r} \in \Omega,$$

$$G_h(\mathbf{r}_s, \omega) = -G_0(\mathbf{r}_s, \omega), \mathbf{r}_s \in \Gamma$$
(2.13)

where we assume Dirichlet boundary conditions $G(\mathbf{r}_s, \omega) = 0$. Generalising the approach to the biharmonic and other wave equations is straightforward [50]. In the high frequency approximation, one can represent G_h in terms of slow varying amplitude A and general phase function ϕ as

$$G_h(\mathbf{r},\omega) = A(\mathbf{r})e^{\mathbf{i}\omega\phi(\mathbf{r})}, \mathbf{r}\in\Omega.$$
 (2.14)

Substituting this expression into Equation (2.5) and separating the real from the imaginary part yields the following two equations

$$|\nabla \phi|^2 = \frac{1}{c^2} + \frac{1}{\omega^2} \frac{\Delta A}{A}$$

$$A \Delta \phi + 2\nabla A \nabla \phi = \nabla \cdot \left(A^2 \nabla \phi\right) = 0$$
(2.15)

Now, assuming that $\frac{\Delta A}{A} \ll k^2$, which is consistent with the slow variation of the amplitude, we can drop the ω^{-2} term in the first equation in (2.15) as $\omega \to \infty$ and get the *eikonal* equation for the phase function alone

$$|\nabla\phi| = \frac{1}{c} = \frac{k}{\omega} \tag{2.16}$$

Now, defining $\mathbf{p} \equiv \nabla \phi$ as the *momentum vector* and setting the Hamilton function $H(\mathbf{r}, \mathbf{p}) = c|\mathbf{p}|$ yields the Hamilton-Jacobi equation

$$H(\mathbf{r}, \mathbf{p}) = 1. \tag{2.17}$$

This equation can be solved by the *method of characteristics* [45, 172, 173]. First, we define ray trajectories $\mathbf{r} = \mathbf{r}(\tau)$ by the following ODEs

$$\frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} = \nabla_{\mathbf{p}} H \,. \tag{2.18}$$

The system of equations in (2.18) is under-determined, therefore we need equations for $\mathbf{p} = \mathbf{p}(\mathbf{r}(\tau))$. To accomplish that, we differentiate (2.17) with respect to \mathbf{r} as

$$\nabla_{\mathbf{r}} H(\mathbf{r}, \mathbf{p}) + \nabla_{\mathbf{p}} H(\mathbf{r}, \mathbf{p}) \nabla \mathbf{p} = 0, \qquad (2.19)$$

and \mathbf{p} — with respect to τ as

$$\frac{\mathrm{d}\mathbf{p}}{\mathrm{d}\tau} = \nabla \mathbf{p} \frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} \,. \tag{2.20}$$

Now, combining Equations (2.18), (2.19) and (2.20) gives Hamilton's equations for ray-trajectories $(\mathbf{r}(\tau), \mathbf{p}(\tau))$

$$\frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} = \nabla_{\mathbf{p}} H = c \frac{\mathbf{p}}{|\mathbf{p}|},$$

$$\frac{\mathrm{d}\mathbf{p}}{\mathrm{d}\tau} = -\nabla_{\mathbf{r}} H.$$
(2.21)

Note that if the phase velocity c is constant in the domain Ω , then $\frac{d\mathbf{p}}{d\tau}$ is zero, and consequently, the ray trajectories are straight lines. Now, if we write the boundary conditions $G_h(\mathbf{r}_s, \omega)$ in the form

$$G_h(\mathbf{r}_s,\omega) = A_0(\mathbf{r}_s)e^{i\omega\phi_0(\mathbf{r}_s)} \quad \text{for} \quad \mathbf{r}_s \in \Gamma,$$
(2.22)

then initial conditions for the system of ODEs in (2.21) can be stated as

$$\mathbf{r}(0) = \mathbf{r}_s$$

for $\mathbf{r}_s \in \Gamma$ (2.23)
$$\mathbf{p}(0) = \nabla \phi(\mathbf{r}(0)) = \nabla \phi_0(\mathbf{r}_s)$$

The solution of Equation (2.21) with the mentioned initial conditions are bicharacteristics (\mathbf{r}, \mathbf{p}) which are curves in the 4-dimensional (4D) phase space $\mathbb{R}^{2\times 2}$. Now, in order to find a solution $G_h(\mathbf{r}', \omega)$ at a point \mathbf{r}' , we need to find all bicharacteristic curves (\mathbf{r}, \mathbf{p}) that pass through the point \mathbf{r}' starting on the boundary points $\mathbf{r}_s \in \Gamma$. In general, there can be infinitely many such bicharacteristic curves, and we can compute the phase function $\phi_j(\mathbf{r}')$ obtained from the ray trajectory starting at the boundary point $\mathbf{r}_{s,j}$ in the following way

$$\frac{\mathrm{d}\phi}{\mathrm{d}\tau} = \nabla \phi \frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} = \mathbf{p} \frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} \to \phi(\mathbf{r}',\tau) = \phi(\mathbf{r},0) + \int_0^\tau \mathbf{p} \frac{\mathrm{d}\mathbf{r}}{\mathrm{d}\tau} \to \phi_j(\mathbf{r}') = \phi_0(\mathbf{r}_{s,j}) + \int_{\mathbf{r}_{s,j}}^{\mathbf{r}'} \mathbf{p} \,\mathrm{d}\mathbf{r} \,.$$
(2.24)

Once the phase function ϕ_j is known, the amplitude A_j is the solution of the *transport* equation

$$A_{j} \Delta \phi + 2\nabla A_{j} \nabla \phi_{j} = \nabla \cdot \left(A_{j}^{2}(\mathbf{r}) \nabla \phi_{j}(\mathbf{r}) \right) = 0, \quad \mathbf{r} \in \Omega,$$

$$J_{0}(\mathbf{r}_{s,j}) = A_{0}^{2}(\mathbf{r}_{s,j}) \frac{\partial \phi_{0}(\mathbf{r}_{s,j})}{\partial n}, \quad \mathbf{r}_{s,j} \in \Gamma,$$
(2.25)

where J_0 is the incoming flux normal to the boundary Γ [51, 174]. Finally, the solution $G_h(\mathbf{r}, \omega)$ to the problem (2.13) can be approximated as

$$G_h(\mathbf{r},\omega) = \sum_j |A_j(\mathbf{r})| e^{i\omega\phi_j(\mathbf{r}) - i\frac{\pi}{2}m_j}, \quad \mathbf{r} \in \Omega, \qquad (2.26)$$

where m_j are the Maslow indices [172, 173]. Since the phase information is not of interest in the DEA method, the details about the Maslow indices are omitted. In what follows, we will be interested in the wave energy density $\epsilon(\mathbf{r}, \omega)$ arising from the reflections off the boundary. It can be approximated as

$$\epsilon(\mathbf{r},\omega) \propto |G_{h}(\mathbf{r},\omega)|^{2} = \sum_{j,j'} |A_{j}A_{j'}| e^{i\omega(\phi_{j}-\phi_{j'})-i\frac{\pi}{2}(m_{j}-m_{j'})} =$$

$$= \sum_{j} |A_{j}|^{2} + \sum_{j \neq j'} |A_{j}A_{j'}| e^{i\omega(\phi_{j}-\phi_{j'})-i\frac{\pi}{2}(m_{j}-m_{j'})}.$$
(2.27)

Here, the first term represents a smooth part of the wave energy density, while the second term is associated with oscillations on the scale of the wavelength. The latter can be made negligibly small after averaging over a frequency band centred on ω . Therefore, the total mean wave energy density can be approximated as

$$\epsilon_{tot}(\mathbf{r},\omega) \propto |G_0(\mathbf{r},\mathbf{r}_0,\omega)|^2 + \rho(\mathbf{r},\omega), \quad \rho(\mathbf{r},\omega) = \sum_j |\bar{A}_j|^2.$$
 (2.28)

In the next section, we present the connection between the smooth part of the wave energy density $\rho(\mathbf{r}, \omega)$ and the boundary integral representation of the phase space density.

2.1.4 Liouville equation and boundary mapping

The mean wave energy density $\rho(\mathbf{r}, \omega)$ can be well approximated by the phase space density $\rho(\mathbf{r}, \mathbf{p}, \omega)$ passing through a point \mathbf{r} as

$$\rho(\mathbf{r},\omega) = \int_{\mathbb{R}^2} \rho(\mathbf{r},\mathbf{p},\omega) d\mathbf{p}. \qquad (2.29)$$

The phase space density $\rho(\mathbf{r}, \mathbf{p}, \omega)$ represents the density of non-interacting particles following trajectories governed by Hamilton equations (2.21) [50]. It satisfies the stationary *Liouville equation* [45, 51] that has the following form

$$\{H, \rho\} = 0, \qquad (2.30)$$

where $\{\cdot, \cdot\}$ are the Poisson brackets which have the following definition

$$\{f(\mathbf{r}, \mathbf{p}), g(\mathbf{r}, \mathbf{p})\} = \nabla_{\mathbf{r}} f \nabla_{\mathbf{p}} g - \nabla_{\mathbf{p}} f \nabla_{\mathbf{r}} g.$$
(2.31)

The solution of Equation (2.30) can be written as

$$\rho(\mathbf{r}, \mathbf{p}, \omega) = \sum_{j} |A_j(\mathbf{r}, \omega)|^2 \delta(\mathbf{p} - \nabla \phi_j(\mathbf{r}, \omega)), \qquad (2.32)$$

where $A_j(\mathbf{r}, \omega)$ and $\phi_j(\mathbf{r}, \omega)$ are solutions of the eikonal and transport equations (2.16) and (2.25), respectively [45, 174]. It is clear that the phase space density $\rho(\mathbf{r}, \mathbf{p}, \omega)$ given by Equation (2.32) satisfies Equation (2.29). This is also called the *Wigner measure* [45, 175], which represents the weak limit of the Wigner distribution as $\omega \to \infty$. Since the summation in (2.28) and (2.32) often consists of infinitely many terms especially when considering long-term dynamics, the formulation can be cumbersome and prone to poor convergence [167]. Instead, we will construct the stationary or long-time limit phase space density $\rho(\mathbf{r}, \mathbf{p}, \omega)$ using a boundary mapping technique. First, we map the initial energy density $\rho_0(\mathbf{r}, \mathbf{r}_0, \omega)$ from the point source at \mathbf{r}_0 to the boundary Γ using the ray-tracing formula (2.17) and (2.21). The initial energy density at a point source can be written as

$$\rho_0(\mathbf{r},\omega) = \frac{R}{2\pi c_g} \frac{e^{-\mu|\mathbf{r}-\mathbf{r}_0|}}{|\mathbf{r}-\mathbf{r}_0|}, \quad \mathbf{r} \in \Omega \subset \mathbb{R}^2,$$
(2.33)

where R is the source power, c_g is the group velocity and μ is the attenuation factor [53, 176]. The attenuation factor μ is related to the damping loss factor η and the group velocity c_g by $\mu = \eta \frac{\omega}{c_g}$. For instance, the attenuation factor μ is equal to ηk for in-plane waves and $0.5 \eta k$ for bending waves in thin isotropic plates. This is because the group velocity c_g is equal to the phase velocity c in the former case and twice the phase velocity 2c in the latter case [13, 142]. Now, the initial phase space density $\rho_0(\mathbf{r}, \mathbf{p}, \omega)$ can be represented as

$$\rho_0(\mathbf{r}, \mathbf{p}, \omega) = \frac{R}{2\pi c_g} \frac{e^{-\mu |\mathbf{r} - \mathbf{r}_0|}}{|\mathbf{r} - \mathbf{r}_0|} \delta(\mathbf{p} - \mathbf{p}_0), \quad \mathbf{p}_0 = |\mathbf{p}| \frac{\mathbf{r} - \mathbf{r}_0}{|\mathbf{r} - \mathbf{r}_0|}, \quad (2.34)$$

where \mathbf{p}_0 is the momentum vector of the ray connecting points \mathbf{r}_0 and \mathbf{r} . Note that the ray travels along the gradient of the Hamilton function $\nabla_{\mathbf{p}} H$ according to Equation (2.21), and in isotropic structures, the ray trajectory is aligned with the momentum vector direction, thus yielding the form of the \mathbf{p}_0 as in (2.34). However, as we will see in Section 3.3, Section 4.3 and Chapter 5, in composite structures, the ray trajectories are not parallel to the momentum vector \mathbf{p} , but to the group velocity vector \mathbf{c}_g , hence, the delta function in Equation (2.34) will have a different form.

It is worth noting that the phase space densities can describe much more general driving sources than just monochromatic point sources. For instance, in the electromagnetic context, phase space densities in the form of Wigner



Figure 2.1: The set up of the local coordinate system at the boundary $X = (s, p_{11})$. A ray emitting from the point \mathbf{r}_0 reach the point \mathbf{r}_s , which has the local position son the boundary Γ . The momentum vector \mathbf{p} has normal and tangential components p_{\perp} and p_{11} at the point s, and θ denotes the angle between the ray connecting \mathbf{r}_0 and s and the normal to Γ .

distributions can represent partially coherent near-homogeneous finite sources [177–181].

Rays emitted isotropically from the point source reach the boundary Γ and produce the phase space density on the boundary $\rho(\mathbf{r}_s, \mathbf{p})$, $\mathbf{r}_s \in \Gamma$. To proceed with the boundary mapping, we need to introduce a coordinate system $X = (s, p_{\parallel})$ on the boundary, where s parametrizes Γ , and p_{\parallel} is the component of the momentum vector \mathbf{p} tangential to Γ at the position s, see Figure 2.1. The component p_{\perp} of the momentum vector \mathbf{p} is normal to the boundary Γ at the position s. Also, we introduce the two-dimensional (2D) boundary phase space density $\rho_{\Gamma}(s, p_{\parallel})$, which will be used to construct the stationary phase space density $\rho_{\Gamma}(s, p_{\parallel})$ and the density on the full 4D phase space $\rho(\mathbf{r}_s, \mathbf{p})$. The former yields the energy flux through the boundary Γ at position s when integrated over the momentum variables, while the latter produces the energy density at the point \mathbf{r}_s [182]. They can be related as follows

$$\rho(\mathbf{r}_s, \mathbf{p}) = \rho_{\Gamma}(s, p_{\parallel}) \delta\left(\tilde{H}(\mathbf{r}_s, \mathbf{p})\right) , \qquad (2.35)$$

where $\tilde{H}(\mathbf{r}, \mathbf{p})$ is the general Hamilton function [50, 51]. In the case of the Helmholtz equation (2.13), $\tilde{H}(\mathbf{r}, \mathbf{p})$ is equal to $c|\mathbf{p}| - 1$ according to Equation (2.17). Now, we apply the delta function substitution rule to the following product of delta functions

$$\frac{cp_{\perp}}{|\mathbf{p}|}\delta(c|\mathbf{p}|-1)\delta(p_{\parallel}-p_{\parallel,0}) = \delta(\mathbf{p}-\mathbf{p}_0), \qquad (2.36)$$

where we combine $\delta(p_{\parallel} - p_{\parallel,0})\delta(p_{\perp} - p_{\perp,0})$ as $\delta(\mathbf{p} - \mathbf{p}_0)$. Inserting the expression (2.34) for the 4D phase space density into Equation (2.35) and using the relation (2.36) yield the initial 2D phase space density on the boundary as

$$\rho_{\Gamma,0}(s,p_{\scriptscriptstyle ||}) = \frac{cR}{2\pi c_g} \frac{e^{-\mu|\mathbf{r}_s-\mathbf{r}_0|}}{|\mathbf{r}_s-\mathbf{r}_0|} \cos\left(\theta(\mathbf{r}_s,\mathbf{r}_0)\right) \delta(p_{\scriptscriptstyle ||}-p_{\scriptscriptstyle ||,0}), \qquad (2.37)$$

where $\theta(\mathbf{r}_s, \mathbf{r}_0)$ is the angle between the ray connecting points \mathbf{r}_0 and \mathbf{r}_s and the normal to Γ , see Figure 2.1. Rays starting with phase space coordinates $X = (s, p_{\scriptscriptstyle \parallel})$ reach the boundary Γ again with phase space coordinates $X' = (s', p'_{\scriptscriptstyle \parallel})$, see Figure 2.1.

Now, the idea is to relate the 2D phase space density $\rho_{\Gamma}(X)$ with the new phase space density $\rho'_{\Gamma}(X')$, which is produced after one scattering event of the rays leaving the boundary Γ . This can be achieved by introducing the boundary integral operator T as

$$\rho_{\Gamma}'(X') = \{T\rho_{\Gamma}\}(X') = \int \lambda(X')e^{-\mu D(X,X')}\delta(X' - \Phi(X))\rho_{\Gamma}(X)dX. \quad (2.38)$$

Here, the function $\Phi(X)$ maps the phase space coordinates $X = (s, p_{\parallel})$ to $X' = (s', p'_{\parallel})$ according to the ray-tracing formula given in (2.21). The term $\lambda(X')$ represents the energy scattering coefficients, that is, the ratios of incoming and outgoing energy fluxes, whereas D(X, X') is the length of the ray trajectory connecting the phase space coordinates X and X'. Equation (2.38) means that the new phase space density ρ'_{Γ} at the phase space point X' is composed of the ray-particles starting at the phase space points X that reach X' whilst being damped along the way with attenuation factor μ and scattered upon hitting the boundary Γ . Note that in undamped systems with perfect

reflection/transmission, that is, when $\mu = 0$ and $\lambda(X') = 1$, the phase space density $\rho'_{\Gamma}(X')$ is equal to $\rho_{\Gamma}(\Phi(X))$, which is consistent with phase space volume conservation [45, 172, 173]. The correspondent integral operator is known as the *Perron-Frobenius* operator [50, 53, 172].

Finally, we compute the stationary or long-time limit 2D phase space density $\rho_{\Gamma,\infty}(X)$ as

$$\rho_{\Gamma,\infty}(X) = \sum_{n=1}^{\infty} T^n \rho_{\Gamma,0}(X) = (\mathbf{I} - T)^{-1} \rho_{\Gamma,0}(X) , \qquad (2.39)$$

where $\rho_{\Gamma,0}$ is the initial 2D phase space density on the boundary given by Equation (2.37). Note that for non-vanishing attenuation, the series converges as $n \to \infty$ [50]. The resulting stationary 4D phase space density $\rho_{\infty}(\mathbf{r}_s, \mathbf{p}), \mathbf{r}_s \in$ Γ can be recovered using Equation (2.35) as

$$\rho_{\infty}(\mathbf{r}_{s},\mathbf{p}) = \rho_{\Gamma,\infty}(s,p_{\text{H}})\delta(\tilde{H}(\mathbf{r}_{s},\mathbf{p})) = \rho_{\Gamma,\infty}(s,p_{\text{H}})\delta(c|\mathbf{p}|-1).$$
(2.40)

Now, one can express the solution of the Liouville equation (2.30) as

$$\rho(\mathbf{r}, \mathbf{p}, \omega) = e^{-\mu |\mathbf{r} - \mathbf{r}_s|} \rho_{\infty}(\mathbf{r}_s(\mathbf{r}, \mathbf{p}), \mathbf{p}), \qquad (2.41)$$

where \mathbf{r}_s is the point of boundary intersection of the ray emanating from the point \mathbf{r} in the direction $-\mathbf{p}$ [53]. Consequently, the mean wave energy density $\rho(\mathbf{r}, \omega)$ at the point \mathbf{r} can be obtained using Equation (2.29).

Having defined the main steps of the DEA method, we still need to address the problem of defining the solution of Equation (2.39). One needs to express the boundary integral operator T in a finite set of basis functions, thus constructing its discrete version. Several basis function sets have been used in the literature. In particular, Tanner [50, 52] used Fourier basis functions both in position and momentum space to represent the operator T. Due to several difficulties such as non-periodic boundary conditions and slow convergence of the associated quadrature rules, Chebyshev polynomial functions with the Gauss-Chebyshev quadrature rule were suggested as an alternative in [183]. It was found in [51, 184] that utilising a basis of functions that is orthogonal in the L^2 space inner

product is useful when considering multi-component fine-meshed systems. The Legendre polynomial functions were used instead of Chebyshev polynomials due to better computational efficiency whilst maintaining fairly good solution approximations.

In this thesis, the Legendre polynomial functions are used to discretise the boundary operator T. However, the implementation of this process and generally of the DEA method will be presented directly for composite structures in Chapter 5. To summarise, the algorithm on how to proceed with the DEA method can be put as

- 1. Define the Hamilton function (2.17) and the ray-tracing equations (2.21).
- 2. Compute the initial 2D phase space density using Equation (2.37).
- 3. Construct the discrete version of the operator T defined in (2.38).
- 4. Solve Equation (2.39) to obtain the stationary phase space density $\rho_{\Gamma,\infty}$.
- 5. Compute the stationary energy density using Equation (2.29).

(2.42)

To extend the applicability of the DEA method for composite structures, several major steps in this algorithm must be redefined accordingly. For instance, the Hamilton function and ray-tracing formula that define the boundary mapping $\Phi : X \to X'$ in the definition of the operator T will take more sophisticated forms in the case of composite structures as we shall see throughout this thesis. The scattering coefficients $\lambda(X')$ in (2.38) play a significant role in defining the long-term dynamics of the phase space densities. As outlined in the introduction, Chapters 3 and 4 are devoted to calculating these coefficients for the various junctions of composite laminated plates. Finally, the numerical implementation algorithm, whilst being based on the works of Chappell et al. [51] and Hartmann et al.[53], needs to be changed significantly to suit the case of composite structures. In the following section, the background theory needed for Chapter 3 is presented.

2.2 Classical Laminated Plate theory

In this section, we describe the Classical Laminated Plate theory for composite laminated plates. It is used in Chapter 3 to set up governing equations of motion of composite plates, which eventually leads to the computation of the scattering coefficients of various composite plate junctions.

The CLP theory is one of the Equivalent Single Layer (ESL) theory types, i.e. we model a composite laminated plate with an equivalent single layer [1, 67]. It is an extension of the Kirchhoff-Love plate theory to laminated plates. The following assumptions are made

- 1. The plate is thin, e.g. thickness is much smaller than length and width dimensions.
- 2. The plate is made of n linear elastic orthotropic laminas that are perfectly bonded together (see Figure 2.2).
- 3. Transverse shear strains ε_{xz} , ε_{yz} and transverse normal strain ε_{zz} are negligible.
- 4. Rotary inertia terms are negligible.
- 5. The effect of in-plane forces on bending is negligible.

Note that in infinite plates, a plate can be considered thin as long as

$$\lambda_B^{\min} \ge 6h \quad \text{or} \quad k_B^{\max}h < 1 \tag{2.43}$$

where h, k_B^{max} and $\lambda_B^{\text{min}} = \frac{2\pi}{k_B^{\text{max}}}$ are the laminate thickness, the maximum wave number and the minimum wave length of the bending mode, respectively [142]. According to the assumptions made, the displacement field can be described



Figure 2.2: The lay-up configuration of a composite laminate.

as

$$u(x, y, z, t) = u_0(x, y, t) - z \frac{\partial w_0}{\partial x}$$

$$v(x, y, z, t) = v_0(x, y, t) - z \frac{\partial w_0}{\partial y}$$

$$w(x, y, z, t) = w_0(x, y, t)$$

(2.44)

where u_0 , v_0 and w_0 are the displacements along the correspondent axes of a point (x, y, 0) on the mid-plane. Therefore, the infinitesimal strain tensor can be written as

$$\begin{cases} \varepsilon_{xx} \\ \varepsilon_{yy} \\ \varepsilon_{xy} \end{cases} = \begin{cases} \frac{\partial u_0}{\partial x} \\ \frac{\partial v_0}{\partial y} \\ \frac{\partial u_0}{\partial y} + \frac{\partial v_0}{\partial x} \end{cases} - z \begin{cases} \frac{\partial^2 w_0}{\partial x^2} \\ \frac{\partial^2 w_0}{\partial y^2} \\ 2\frac{\partial^2 w_0}{\partial x \partial y} \end{cases} .$$
(2.45)

Stress components of the kth lamina can be found from the following constitutive relations:

$$\begin{cases} \sigma_{xx} \\ \sigma_{yy} \\ \sigma_{xy} \end{cases}^{k} = \begin{bmatrix} Q_{11} & Q_{12} & Q_{16} \\ Q_{12} & Q_{22} & Q_{26} \\ Q_{26} & Q_{26} & Q_{66} \end{bmatrix}^{k} \begin{cases} \varepsilon_{xx} \\ \varepsilon_{yy} \\ \varepsilon_{xy} \end{cases}$$
(2.46)

where the Q_{ij} 's are plane-stress reduced stiffnesses of an orthotropic lamina. If the principle material axes of the kth lamina are not aligned with the natural coordinates of the plate, then all $Q_{ij}^k \neq 0$, and the lamina is generally orthotropic [1]. If they are aligned, then $Q_{16}^k = Q_{26}^k = 0$, and the lamina is specially orthotropic. Details about the relations between the Q_{ij} 's and the more familiar coefficients for specially orthotropic plates as, for example, found in [185], together with the relations to material constants can be found in Appendix B. In-plane stress and moment resultants can be computed via an integration of stresses along the thickness direction as follows

$$\begin{cases}
 N_{xx} \\
 N_{yy} \\
 N_{xy}
 \end{cases} = \int_{-\frac{h}{2}}^{\frac{h}{2}} \begin{cases}
 \sigma_{xx} \\
 \sigma_{yy} \\
 \sigma_{xy}
 \end{cases} dz = \sum_{k=1}^{n} \int_{z_{k}}^{z_{k+1}} \begin{cases}
 \sigma_{xx} \\
 \sigma_{yy} \\
 \sigma_{xy}
 \end{cases}^{k} dz \\
 \sigma_{xy}
 \end{cases} dz \\
\begin{cases}
 M_{xx} \\
 M_{yy} \\
 M_{xy}
 \end{cases} = -\int_{-\frac{h}{2}}^{\frac{h}{2}} \begin{cases}
 \sigma_{xx} \\
 \sigma_{yy} \\
 \sigma_{xy}
 \end{cases} dz = -\sum_{k=1}^{n} \int_{z_{k}}^{z_{k+1}} \begin{cases}
 \sigma_{xx} \\
 \sigma_{yy} \\
 \sigma_{xy}
 \end{cases}^{k} zdz$$
(2.47)

Now, we can use relations (2.45) and (2.46) to represent the in-plane stress and moment resultants in terms of the displacement field as

$$\begin{cases} N_{xx} \\ N_{yy} \\ N_{xy} \end{cases} = \begin{bmatrix} A_{11} & A_{12} & A_{16} \\ A_{12} & A_{22} & A_{26} \\ A_{26} & A_{22} & A_{66} \end{bmatrix} \begin{cases} \frac{\partial u_0}{\partial x} \\ \frac{\partial v_0}{\partial y} \\ \frac{\partial u_0}{\partial y} + \frac{\partial v_0}{\partial x} \end{cases} - \begin{bmatrix} B_{11} & B_{12} & B_{16} \\ B_{12} & B_{22} & B_{26} \\ B_{26} & B_{22} & B_{66} \end{bmatrix} \begin{cases} \frac{\partial^2 w_0}{\partial x^2} \\ \frac{\partial^2 w_0}{\partial x \partial y} \end{cases}$$
$$\begin{cases} M_{xx} \\ M_{yy} \\ M_{xy} \end{cases} = \begin{bmatrix} D_{11} & D_{12} & D_{16} \\ D_{12} & D_{22} & D_{26} \\ D_{26} & D_{22} & D_{66} \end{bmatrix} \begin{cases} \frac{\partial^2 w_0}{\partial x^2} \\ \frac{\partial^2 w_0}{\partial x^2}$$

where A_{ij} , B_{ij} and D_{ij} are called extensional, bending-extensional coupling and bending stiffnesses, defined as follows

$$(A_{ij}, B_{ij}, D_{ij}) = \int_{-\frac{h}{2}}^{\frac{h}{2}} Q_{ij} \left(1, z, z^2\right) dz = \sum_{k=1}^{n} \int_{z_k}^{z_{k+1}} Q_{ij}^k \left(1, z, z^2\right) dz.$$
(2.49)

If individual layers are homogeneous, the integrations in Equation 2.49 can be

replaced by summations in the following way:

$$A_{ij} = \sum_{k=1}^{n} Q_{ij}^{k} (z_{k+1} - z_{k})$$

$$B_{ij} = \frac{1}{2} \sum_{k=1}^{n} Q_{ij}^{k} (z_{k+1}^{2} - z_{k}^{2}).$$

$$D_{ij} = \frac{1}{3} \sum_{k=1}^{n} Q_{ij}^{k} (z_{k+1}^{3} - z_{k}^{3})$$

(2.50)

Finally, the governing equations of free motion can be written in terms of in-plane stress and moment resultants as follows

$$\frac{\partial N_{xx}}{\partial x} + \frac{\partial N_{xy}}{\partial y} = I_0 \frac{\partial^2 u_0}{\partial t^2} - I_1 \frac{\partial^2}{\partial^2 t} \left(\frac{\partial w_0}{\partial x}\right)$$
$$\frac{\partial N_{xy}}{\partial x} + \frac{\partial N_{yy}}{\partial y} = I_0 \frac{\partial^2 v_0}{\partial t^2} - I_1 \frac{\partial^2}{\partial^2 t} \left(\frac{\partial w_0}{\partial y}\right) \qquad , \qquad (2.51)$$
$$\frac{\partial^2 M_{xx}}{\partial x^2} + 2 \frac{\partial^2 M_{xy}}{\partial x \partial y} + \frac{\partial^2 M_{yy}}{\partial y^2} = -I_0 \frac{\partial^2 w_0}{\partial t^2} + I_1 \frac{\partial^2}{\partial^2 t} \left(\frac{\partial u_0}{\partial y} + \frac{\partial v_0}{\partial x}\right)$$

where $[I_0, I_1] = \int_{-\frac{h}{2}}^{\frac{h}{2}} \rho_0 [1, z] dz$ - the mass moments of inertia with ρ_0 being the material density. Transversal tractions Q_x and Q_y which do not appear in the equations of motion are important when formulating the boundary conditions. They can be expressed in terms of moments as follows

$$Q_x = -\frac{\partial M_{xx}}{\partial x} - \frac{\partial M_{xy}}{\partial y}$$

$$Q_y = -\frac{\partial M_{yy}}{\partial y} - \frac{\partial N_{xy}}{\partial x}.$$
(2.52)

These tractions together with the twisting moment contributions $-\frac{\partial M_{xy}}{\partial x}$ and $-\frac{\partial M_{xy}}{\partial y}$ define the effective shear forces V_x , V_y as follows:

$$V_x = Q_x - \frac{\partial M_{xy}}{\partial y}$$

$$V_y = Q_y - \frac{\partial M_{xy}}{\partial x}.$$
(2.53)

Note that for laminates that consist of an odd number of laminas and are symmetric, that is, when the ply stacking sequence, material, and ply thicknesses are symmetric around the mid-plane, all bending-extension stiffnesses $B_{ij} = 0$, and the mass moment of inertia $I_1 = 0$. Therefore, for such laminates, the equations of motion (2.51) become uncoupled, i.e. the first two equations gov-



(b) with SHELL181 elements

Figure 2.3: A periodic segment of a composite plate with two different alternating plies modelled with SOLID185 (a) and SHELL181 elements (b) in Ansys. The degrees of freedom are grouped into internal \mathbf{q}_I , edge \mathbf{q}_L , \mathbf{q}_R , \mathbf{q}_B , \mathbf{q}_T and corner \mathbf{q}_{LB} , \mathbf{q}_{RB} , \mathbf{q}_{LT} , \mathbf{q}_{RT} degrees of freedom.

ern the in-plane motion of the laminate, whereas the last equation describes the out-of-plane motion of the laminate. This concludes the review of the CLP theory for thin composite laminates. In the next section, the WFE method for composite two-dimensional waveguides is reviewed.

2.3 The Wave Finite Element method for composite plates

2.3.1 Statement of the problem

We consider a unit cell of a composite plate with arbitrary lay-up through the thickness direction and plane dimensions d_x and d_y . It can be modelled using two-dimensional shell elements with a composite lay-up or three-

dimensional solid elements stacked up one on top of the other, representing different composite layers. For instance, in ANSYS, the former can be represented by the linear element type SHELL181, and the latter by the linear element type SOLID185. Figure 2.3 presents a unit cell of a five-layer plate meshed with SOLID185 and SHELL181 elements and a nodal displacements vector labelling convention. A SOLID185 element consists of eight nodes with three translational degrees of freedom per node, whereas a SHELL181 element consists of four nodes with three translational and three rotational degrees of freedom per node. The nodal degrees of freedom are grouped into internal \mathbf{q}_I , edge \mathbf{q}_L , \mathbf{q}_R , \mathbf{q}_B , \mathbf{q}_T and corner \mathbf{q}_{LB} , \mathbf{q}_{RB} , \mathbf{q}_{LT} , \mathbf{q}_{RT} degrees of freedom. Accordingly, the nodal displacements vector \mathbf{q} can be organised as $\mathbf{q} = \left\{ \mathbf{q}_{LB} \ \mathbf{q}_{RB} \ \mathbf{q}_{B} \ \mathbf{q}_{L} \ \mathbf{q}_{R} \ \mathbf{q}_{I} \ \mathbf{q}_{LT} \ \mathbf{q}_{RT} \ \mathbf{q}_{T} \right\}^{\mathrm{T}}.$ The nodal forces vector \mathbf{f} is arranged in the same manner. The number of degrees of freedom must be the same for each pair of edges on opposite faces. The number of mesh cells in the x, y and z direction are labelled by n_x , n_y and n_z . The number of degrees of freedom per edge is labelled as m; for plates modelled with SOLID185 elements $m = 3(n_z + 1)$, whereas SHELL181 elements based plates have m = 6since there is only one node in the z direction. Consequently, the sizes of nodal displacement sub-vectors can be represented as

$$|\mathbf{q}_{L(R)B(T)}| = m, \ |\mathbf{q}_{L(R)}| = m(n_x - 1),$$

$$|\mathbf{q}_{B(T)}| = m(n_y - 1), \ |\mathbf{q}_I| = m(n_x - 1)(n_y - 1).$$
(2.54)

The governing equation of motion of the unit cell can be written as

$$\mathbf{M}\ddot{\mathbf{q}}(t) + \mathbf{C}\dot{\mathbf{q}}(t) + \mathbf{K}\mathbf{q}(t) = \mathbf{f}, \qquad (2.55)$$

where **M**, **C** and **K** are mass, damping and stiffness matrices, respectively. Note that these matrices are symmetric. The dimension of Equation (2.55) is $m(n_x+1)(n_y+1)$. We assume that the structure undergoes a harmonic motion at angular frequency ω , therefore, $\ddot{\mathbf{q}} = -\omega^2 \mathbf{q}$ and $\dot{\mathbf{q}} = i\omega \mathbf{q}$. Introducing this expression into Equation (2.55) yields

$$\left[\mathbf{K} + \mathrm{i}\omega\mathbf{C} - \omega^2\mathbf{M}\right]\mathbf{q} = \mathbf{f}.$$
 (2.56)

The matrix **C** can be composed of different components including layer- and material-wise viscous and structural damping matrices, but in this work, we assume that $\mathbf{C} = \frac{\eta}{\omega} \mathbf{K}$, where η is a uniform structural damping coefficient. Therefore, Equation (2.56) can be written as

$$\left[\mathbf{K}\left(1+\mathrm{i}\eta\right)-\omega^{2}\mathbf{M}\right]\mathbf{q}=\mathbf{f}.$$
(2.57)

2.3.2 Periodic structure theory and eigenvalue problem

We consider a plane wave travelling across the plate to be of the form $e^{-ik_x x - ik_y y + i\omega t}$, where k_x and k_y are x and y components of the wave vector \mathbf{k} . Periodic structure theory [186, 187] requires that the displacement vector between adjacent nodes and opposite sides differs only by a propagation factor λ . For the nodes placed along the x direction (see Figure 2.3), we obtain

$$\begin{cases} \mathbf{q}_{LT} \\ \mathbf{q}_{RT} \\ \mathbf{q}_{B} \end{cases} = \lambda_{x} \begin{cases} \mathbf{q}_{LB} \\ \mathbf{q}_{RB} \\ \mathbf{q}_{T} \end{cases} , \qquad (2.58)$$

where $\lambda_x = e^{-ik_x d_x}$ is the propagation factor in the *x* direction. Now, we denote as \mathbf{q}_X^j , X = L, R, I displacement sub-vectors of nodes placed at $x_j = d_x j/n_x$, $j = 1, 2, \ldots, n_x - 1$. Consequently, as in Equation (2.58), one can relate internal and edge degrees of freedom to bottom ones as

$$\begin{cases} \mathbf{q}_{L}^{j} \\ \mathbf{q}_{R}^{j} \\ \mathbf{q}_{I}^{j} \end{cases} = e^{-\mathbf{i}k_{x}d_{x}j/n_{x}} \begin{cases} \mathbf{q}_{LB} \\ \mathbf{q}_{RB} \\ \mathbf{q}_{B} \end{cases} = \lambda_{x}^{j/n_{x}} \begin{cases} \mathbf{q}_{LB} \\ \mathbf{q}_{RB} \\ \mathbf{q}_{B} \end{cases} .$$
(2.59)

Using Equations (2.58) and (2.59), we can express the nodal displacements vector \mathbf{q} in terms of displacement sub-vectors of nodes $\mathbf{q}_{red} = \left\{ \mathbf{q}_{LB} \quad \mathbf{q}_{RB} \quad \mathbf{q}_B \right\}^{\mathrm{T}}$

as

$$\mathbf{q} = \mathbf{T}\mathbf{q}_{red}, \quad \mathbf{T} = \begin{bmatrix} \mathbf{I}_{1} & 0 & 0 \\ 0 & \mathbf{I}_{1} & 0 \\ 0 & 0 & \mathbf{I}_{2} \\ \vdots & & & \\ \lambda_{x}^{j/n_{x}} \mathbf{I}_{1} & 0 & 0 \\ 0 & \lambda_{x}^{j/n_{x}} \mathbf{I}_{2} \\ \vdots & & & \\ \lambda_{x} \mathbf{I}_{1} & 0 & 0 \\ 0 & \lambda_{x} \mathbf{I}_{2} \end{bmatrix}, \quad j = 1, 2, \dots, n_{x} - 1,$$

$$(2.60)$$

where \mathbf{I}_1 and \mathbf{I}_2 are identity matrices with dimensions m and $m(n_y - 1)$. The dimension of the matrix \mathbf{T} is $(m(n_x + 1)(n_y + 1), m(n_y + 1))$, and hence, Equation (2.60) can be used to reduce the dimension of Equation (2.57). In fact, premultiplying both sides of Equation (2.57) by conjugate transpose \mathbf{T}^H yields

$$\mathbf{T}^{H}[\mathbf{K}(1+\mathrm{i}\eta)-\omega^{2}\mathbf{M}]\mathbf{T}\mathbf{q}_{red}=\mathbf{f}_{red},\qquad(2.61)$$

and

$$\mathbf{f}_{red} = \left\{ \begin{aligned} \mathbf{\tilde{f}}_L \\ \mathbf{\tilde{f}}_R \\ \mathbf{\tilde{f}}_O \end{aligned} \right\} = \mathbf{T}^H \mathbf{f} = \left\{ \begin{aligned} \mathbf{f}_{LB} + \sum_{\substack{j=1\\n_x-1}}^{n_x-1} \lambda_x^{-j/n_x} \mathbf{f}_L^j + \lambda_x^{-1} \mathbf{f}_{LT} \\ \mathbf{f}_{RB} + \sum_{\substack{j=1\\n_x-1}}^{n_x-1} \lambda_x^{-j/n_x} \mathbf{f}_R^j + \lambda_x^{-1} \mathbf{f}_{RT} \\ \mathbf{f}_B + \sum_{j=1}^{n_x-1} \lambda_x^{-j/n_x} \mathbf{f}_I^j + \lambda_x^{-1} \mathbf{f}_T \end{aligned} \right\} , \qquad (2.62)$$

where \mathbf{f}_X^j , X = L, R, I denote force sub-vectors of nodes placed at $x_j = d_x j/n_x$, $j = 1, 2, ..., n_x - 1$. Internal nodal forces $\mathbf{f}_I = 0$ in the absence of external forces, and force equilibrium between opposite sides yields $\lambda_x \mathbf{f}_B + \mathbf{f}_T = 0$. Therefore, the reduced nodal forces vector $\mathbf{\tilde{f}}_O = 0$. Relabelling \mathbf{q}_{red} in Equation (2.61) as $\left\{ \tilde{\mathbf{q}}_{L} \ \tilde{\mathbf{q}}_{R} \ \tilde{\mathbf{q}}_{O} \right\}^{\mathrm{T}}$ yields the following equations of motion: $\begin{bmatrix} \mathbf{D}_{LL} \ \mathbf{D}_{LR} \ \mathbf{D}_{LO} \end{bmatrix} \begin{pmatrix} \tilde{\mathbf{q}}_{L} \end{pmatrix} \begin{pmatrix} \tilde{\mathbf{f}}_{L} \end{pmatrix}$

$$\begin{bmatrix} \mathbf{D}_{RL} & \mathbf{D}_{RR} & \mathbf{D}_{RO} \\ \mathbf{D}_{OL} & \mathbf{D}_{OR} & \mathbf{D}_{OO} \end{bmatrix} \begin{bmatrix} \mathbf{I}_{L} \\ \tilde{\mathbf{q}}_{R} \\ \tilde{\mathbf{q}}_{O} \end{bmatrix} = \begin{bmatrix} \mathbf{I}_{L} \\ \tilde{\mathbf{f}}_{R} \\ 0 \end{bmatrix}, \qquad (2.63)$$

where $\mathbf{D} = \mathbf{T}^{H}[\mathbf{K}(1+i\eta) - \omega^{2}\mathbf{M}]\mathbf{T}$ is the reduced dynamic stiffness matrix. The dimension of the square matrix \mathbf{D} is $m(n_{y} + 1)$. Now, we can reduce further the dimension of Equation (2.63) via dynamic condensation of internal degrees of freedom $\tilde{\mathbf{q}}_{O}$ and obtain the following equation

$$\begin{bmatrix} \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR} \\ \tilde{\mathbf{D}}_{RL} & \tilde{\mathbf{D}}_{RR} \end{bmatrix} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{cases} = \begin{cases} \tilde{\mathbf{f}}_L \\ \tilde{\mathbf{f}}_R \end{cases}, \qquad (2.64)$$

where

$$\tilde{\mathbf{D}}_{LL} = \mathbf{D}_{LL} - \mathbf{D}_{LO} \mathbf{D}_{OO}^{-1} \mathbf{D}_{OL}$$

$$\tilde{\mathbf{D}}_{LR} = \mathbf{D}_{LR} - \mathbf{D}_{RO} \mathbf{D}_{OO}^{-1} \mathbf{D}_{OR}$$

$$\tilde{\mathbf{D}}_{RL} = \mathbf{D}_{RL} - \mathbf{D}_{LO} \mathbf{D}_{OO}^{-1} \mathbf{D}_{OL}$$

$$\tilde{\mathbf{D}}_{RR} = \mathbf{D}_{RR} - \mathbf{D}_{RO} \mathbf{D}_{OO}^{-1} \mathbf{D}_{OR}$$
(2.65)

The term \mathbf{D}_{OO}^{-1} is computed for each frequency of interest, thus increasing the computational cost. To reduce it, one can approximate \mathbf{D}_{OO}^{-1} by expanding in a power series in ω^2 [127] as

$$\mathbf{D}_{OO}^{-1} = \left(\mathbf{I} + \frac{\omega^2}{1+i\eta} \tilde{\mathbf{K}}_{OO}^{-1} \tilde{\mathbf{M}}_{OO} + O\left(\left(\frac{\omega^2}{1+i\eta} \tilde{\mathbf{K}}_{OO}^{-1} \tilde{\mathbf{M}}_{OO}\right)^2\right)\right) \tilde{\mathbf{K}}_{OO}^{-1} \frac{1}{1+i\eta} \approx \\ \approx \left(\mathbf{I} + \frac{\omega^2}{1+i\eta} \tilde{\mathbf{K}}_{OO}^{-1} \tilde{\mathbf{M}}_{OO}\right) \tilde{\mathbf{K}}_{OO}^{-1} \frac{1}{1+i\eta}$$

$$(2.66)$$

Now, substituting this expression into Equations (2.65) gives

$$\begin{split} \tilde{\mathbf{D}} &= \left(\begin{bmatrix} \tilde{\mathbf{K}}_{LL} & \tilde{\mathbf{K}}_{LR} \\ \tilde{\mathbf{K}}_{RL} & \tilde{\mathbf{K}}_{RR} \end{bmatrix} - \begin{bmatrix} \tilde{\mathbf{K}}_{LO} \\ \tilde{\mathbf{K}}_{RO} \end{bmatrix} \tilde{\mathbf{K}}_{OO}^{-1} \begin{bmatrix} \tilde{\mathbf{K}}_{OL} \\ \tilde{\mathbf{K}}_{OR} \end{bmatrix} \right) (1 + \mathrm{i}\eta) - \omega^{2} \left(\begin{bmatrix} \tilde{\mathbf{M}}_{LL} & \tilde{\mathbf{M}}_{LR} \\ \tilde{\mathbf{M}}_{RL} & \tilde{\mathbf{M}}_{RR} \end{bmatrix} - \\ &- \begin{bmatrix} \tilde{\mathbf{K}}_{LO} \\ \tilde{\mathbf{K}}_{RO} \end{bmatrix} \tilde{\mathbf{K}}_{OO}^{-1} \begin{bmatrix} \tilde{\mathbf{M}}_{OL} \\ \tilde{\mathbf{M}}_{OR} \end{bmatrix} - \begin{bmatrix} \tilde{\mathbf{M}}_{LO} \\ \tilde{\mathbf{M}}_{RO} \end{bmatrix} \tilde{\mathbf{K}}_{OO}^{-1} \begin{bmatrix} \tilde{\mathbf{K}}_{OL} \\ \tilde{\mathbf{K}}_{OO} \end{bmatrix} + \begin{bmatrix} \tilde{\mathbf{K}}_{LO} \\ \tilde{\mathbf{K}}_{RO} \end{bmatrix} \tilde{\mathbf{K}}_{OO}^{-1} \tilde{\mathbf{M}}_{OO} \begin{bmatrix} \tilde{\mathbf{K}}_{OL} \\ \tilde{\mathbf{K}}_{OR} \end{bmatrix} \right), \end{split}$$
(2.67)

where $\tilde{\mathbf{K}}_{XY} = \mathbf{T}^* \mathbf{K}_{XY} \mathbf{T}$ and $\tilde{\mathbf{M}}_{XY} = \mathbf{T}^* \mathbf{M}_{XY} \mathbf{T}$ for X, Y = L, R, O. The component $\tilde{\mathbf{K}}_{OO}^{-1}$ can be precomputed for the range of frequencies considered, thus decreasing the computational cost. Finally, by applying the periodic structure theory and force equilibrium in the y direction, which can be written as

$$\tilde{\mathbf{q}}_R = \lambda_y \, \tilde{\mathbf{q}}_L \,, \quad \tilde{\mathbf{f}}_R = -\lambda_y \, \tilde{\mathbf{f}}_L \,, \quad \lambda_y = e^{-\mathrm{i}k_y d_y} \,, \tag{2.68}$$

one can get from Equation (2.64) the following eigenvalue problem for the propagation factor λ_y

$$\mathbf{S} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} = \lambda_y \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} \quad \text{with} \quad \mathbf{S} = \begin{bmatrix} -\tilde{\mathbf{D}}_{LR}^{-1} \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR}^{-1} \\ -\tilde{\mathbf{D}}_{RL} + \tilde{\mathbf{D}}_{RR} \tilde{\mathbf{D}}_{LR}^{-1} \tilde{\mathbf{D}}_{LL} & -\tilde{\mathbf{D}}_{RR} \tilde{\mathbf{D}}_{LR}^{-1} \end{bmatrix}.$$
(2.69)

Solving Equation (2.69) yields the propagation factors λ_y and, eventually, the wave number components k_y .

This concludes the review of the WFE method for composite plates.

2.4 Conclusion

This chapter aimed to review the background theory needed to understand the main findings of this thesis presented in the following chapters. In Section 2.1, we have described the basics of the DEA method for two-dimensional isotropic structures. Furthermore, we have reviewed the wave problems that are considered throughout this thesis. Finally, we have identified and discussed the main parts of the DEA method that need to be modified to allow for its application to composite structures.

One of the main parts to be changed is the calculation of scattering coefficients for composite plates. These coefficients are obtained via two approaches the semi-analytical approach based on the CLP theory for thin laminated plates and the hybrid FE/WFE approach for general composite plates. In Section 2.2, we have described the CLP theory for thin laminated plates. It will be used in Chapter 3 to set up the governing equations of motion for composite plates, which eventually lead to the computation of the scattering coefficients of various junctions. Finally, in Section 2.3, we have considered the main principles and equations in the WFE method for composite plates modelled using two-dimensional and three-dimensional finite elements. The equations presented in this section will be used further in Chapter 4 to derive expressions for scattering coefficients using the hybrid FE/WFE method.

Chapter 3

Energy scattering properties of line joints connecting composite plates

3.1 Introduction

In this chapter, the semi-analytical method for computing scattering coefficients of line joints connecting composite laminated plates is derived. Composite laminates are modelled in the context of the CLP theory introduced in Section 2.2. We restrict our study to symmetric laminates; that is, the laminas are symmetrically placed around the mid-plane of the laminate. As shown in Section 2.2, in such a case, the governing equations of motion (2.51) become uncoupled, thus reducing the computational complexity of the model. There are two specific cases which we will have a look at that are described in Appendix A, namely, regular cross- and angle-ply laminates with alternating ply direction angles $(0^{\circ}/90^{\circ})$ and $(-\alpha/\alpha)$, $0^{\circ} < \alpha < 90^{\circ}$, respectively. The method presented here will be used to validate a wave finite element approach of extracting reflection and transmission coefficients in Chapter 4.

This chapter is organised as follows. In Section 3.2, the governing equations of motion for composite symmetric laminated plates are presented. Then, dispersion relations and group velocities for in-plane and out-of-plane waves are given in Section 3.3. The importance of modifications to Snell's law for composite plates is discussed using an example configuration. In Section 3.4, the wave dynamic stiffness matrix is then introduced, which relates displacements and



Figure 3.1: The set up of N plates joined together at a common interface along the x_g axis.

forces at the junction. In Section 3.5, the individual dynamic stiffness matrices for plates are then assembled into a global equation via the application of continuity and equilibrium conditions at the junction. Consequently, we obtain the scattering matrix formulation for a junction of several semi-infinite plates. As a final step, in Section 3.6, we derive an effective scattering matrix for a plate with multiple finite stiffeners attached to it. In Section 3.7, numerical case studies for two and three coupled composite plates are presented.

3.2 Statement of the problem

In this section, we outline the mathematical basis of the wave scattering problem. We consider N semi-infinite thin laminated plates connected along a lossless junction as shown in Figure 3.1. The plates consist of n generally orthotropic laminas. The shared edge of the plates is aligned with the x_g axis of the global coordinate system (x_g, y_g, z_g) . All plates are semi-infinite in the positive direction of their local y_j axis. The rotation angle ψ_j describes the position of the *j*th plate relative to the y_g axis. The position, displacements and tractions on the *j*th plate are defined with respect to the local coordinate system (x_j, y_j, z_j) , where the local and global x axis are identical, see Figure 3.2. Note that the positive direction of the y_j axis always points away from the junction.

3.2.1 Governing equations of motion

As outlined in Section 3.1, we consider symmetric laminated plates, and therefore, all bending-extension coefficients B_{ij} are equal to zero. Substituting (2.48) into (2.51) yields the governing equations of motion for the *j*th plate in terms of displacements u_j, v_j, w_j as

$$A_{11}\frac{\partial^2 u}{\partial x^2} + 2A_{16}\frac{\partial^2 u}{\partial x \partial y} + A_{66}\frac{\partial^2 u}{\partial y^2} + A_{16}\frac{\partial^2 v}{\partial x^2} + (A_{12} + A_{66})\frac{\partial^2 v}{\partial x \partial y} + A_{26}\frac{\partial^2 v}{\partial y^2} = \rho h \frac{\partial^2 u}{\partial t^2},$$

$$A_{22}\frac{\partial^2 v}{\partial y^2} + 2A_{26}\frac{\partial^2 v}{\partial x \partial y} + A_{66}\frac{\partial^2 v}{\partial x^2} + A_{16}\frac{\partial^2 u}{\partial x^2} + (A_{12} + A_{66})\frac{\partial^2 u}{\partial x \partial y} + A_{26}\frac{\partial^2 u}{\partial y^2} = \rho h \frac{\partial^2 v}{\partial t^2},$$

$$D_{11}\frac{\partial^4 w}{\partial x^4} + 4D_{16}\frac{\partial^3 w}{\partial x^3 \partial y} + 2(D_{12} + 2D_{66})\frac{\partial^4 w}{\partial x^2 \partial y^2} + 4D_{26}\frac{\partial^3 w}{\partial x \partial y^3} + D_{22}\frac{\partial^4 w}{\partial y^4} = -\rho h \frac{\partial^2 w}{\partial t^2}.$$

$$(3.1)$$

Recall that coefficients A_{ij} and D_{ij} are computed using Equation (2.49), and for special cases of cross- and angle-ply laminates these coefficients are obtained by formula given in Appendices (A.1) and (A.2). We omit the plate index j in Equations (3.1) and whenever we talk about a specific plate for ease of notation; we emphasise here that all quantities, including the material parameters, are plate dependent. The equations of motion for in-plane and out-of-plane motion in the plate are uncoupled and so are the relations between in-plane displacements u_e , v_e to in-plane tractions N_{xy} , N_{yy} and out-of-plane displace-



Figure 3.2: Local coordinate system, displacements and tractions on the common edge "e" of the *j*th plate.

ment w_e and rotation θ_e to effective shear force V_y and bending moment M_{yy} at a junction or edge "e" of that plate j, see Figure 3.2. In what follows, we will use the sub-index "e", whenever we want to stress that a quantity is taken at the edge of a plate j, that is, at $y_j = 0$, following the notation in [147]. By expanding Equation (2.48), the elastic tractions and moments can be written in the following way

$$N_{yy} = A_{22} \frac{\partial v}{\partial y} + A_{12} \frac{\partial u}{\partial x} + A_{26} \left(\frac{\partial u}{\partial y} + \frac{\partial v}{\partial x} \right)$$

$$N_{xy} = A_{16} \frac{\partial u}{\partial x} + A_{26} \frac{\partial v}{\partial y} + A_{66} \left(\frac{\partial u}{\partial y} + \frac{\partial v}{\partial x} \right)$$

$$V_y = -D_{22} \frac{\partial^3 w}{\partial y^3} - 4D_{26} \frac{\partial^3 w}{\partial x \partial y^2} - (D_{12} + 4D_{66}) \frac{\partial^3 w}{\partial x^2 \partial y} - 2D_{16} \frac{\partial^3 w}{\partial x^3}$$

$$M_{yy} = D_{22} \frac{\partial^2 w}{\partial y^2} + D_{12} \frac{\partial^2 w}{\partial x^2} + 2D_{26} \frac{\partial^2 w}{\partial x \partial y}.$$
(3.2)

In what follows, the twisting moment M_{xy} will be important in calculating the energy flow of a bending mode. It is expressed in terms of displacements as

$$M_{xy} = D_{16} \frac{\partial^2 w}{\partial x^2} + D_{26} \frac{\partial^2 w}{\partial y^2} + 2D_{66} \frac{\partial^2 w}{\partial x \partial y}.$$
(3.3)

The angle of rotation γ is approximated as $\frac{\partial w}{\partial y}$ (see Equation (2.44)). Finally, it is noted that Equations (3.1) and (3.2) with appropriate extensional and bending coefficients A_{ij} and D_{ij} describe the structural behaviour of a generally orthotropic plate, that is, an orthotropic plate with principal material axes not necessarily aligned with its local coordinate system. Therefore, we can also analyse the vibrational motion of individual orthotropic lamina of a composite plate and, consequently, the scattering properties of the junctions of orthotropic plates.

3.3 Dispersion relations, the group velocity and Snell's law

Before discussing reflection and transmission coefficients for incoming waves at a specific angle of incidence, it is worth considering the relations between the angle of incoming and outgoing waves at an interface, that is, the relation equivalent of Snell's law for composite plates.

3.3.1 Relations between incoming and outgoing waves at junctions for composite plates

Since the directions of the group and phase velocities do not necessarily coincide in non-isotropic media [166], formulating the relations between the angles of incoming and outgoing waves at interfaces between plates with different material properties and thus the generalisation of Snell's law is less straightforward. The connection between group and phase velocities is provided by the dispersion relation, which needs to be studied in detail. Given that energy is transported along the group velocity vectors, the relation between incoming and outgoing group velocity directions is essential for calculating ray trajectories in the context of DEA.

We consider a plane wave of the form $e^{-ik_x x - ik_y y + i\omega t}$ travelling across the plate, where k_x and k_y are the x and y components of the wave vector **k**, and ω is the angular frequency. Substituting this plane wave solution into the 4th order bending equation (3.1) yields the characteristic equation for bending waves

$$D_{11}k_x^4 + 4D_{16}k_x^3k_y + 2(D_{12} + 2D_{66})k_x^2k_y^2 + 4D_{26}k_xk_y^3 + D_{22}k_y^4 - \rho h\omega^2 = 0.$$
(3.4)

Following the same procedure using the first two equations in (3.1), the characteristic equation for in-plane waves can be expressed as

$$(A_{11}k_x^2 + 2A_{16}k_xk_y + A_{66}k_y^2 - \rho h\omega^2) (A_{66}k_x^2 + 2A_{26}k_xk_y + A_{22}k_y^2 - \rho h\omega^2) - - (A_{16}k_x^2 + (A_{12} + A_{66})k_xk_y + A_{26}k_y^2)^2 = 0.$$

$$(3.5)$$

For fixed ω , the solutions to Equations (3.4) and (3.5) give rise to a closed curve in (k_x, k_y) space describing the *wave vector* curves for bending, shear and longitudinal waves. We note that only real solutions (k_x, k_y) describe propagating waves.

We are interested in the dynamic response of a plate j to an incident plane wave

of the form $e^{-ik_x x + ik_y y + i\omega t}$ travelling towards the junction on plate j'. Compatibility conditions at the junction yield the response of the plate j in the form $e^{-ik_x x - i\mu y + i\omega t}$, that is, k_x and ω are common to all plates on the edge "e". The y component of the wave vector, here denoted μ , is then computed from the dispersion relations (3.4) and (3.5). For the out-of-plane motion, one obtains four solutions of Equation (3.4) which come in pairs of roots μ^{\pm} . One pair is either real or complex, corresponding to propagating or evanescent bending waves, respectively. We denote this pair as $\mu_{B_1}^{\pm}$ where the superscripts "+" and "-" represent outgoing and incoming waves. The other pair of roots denoted $\mu_{B_2}^{\pm}$ is always complex; thus, the corresponding bending waves are always evanescent. Evanescent waves are usually characterised with purely imaginary values of μ^{\pm} , so that the correspondent displacements increase or decay exponentially along its direction without oscillation [142, 147, 166]. In such waves, energy can only be carried through the interaction of a pair of waves [188, 189]. One might argue that for complex μ^{\pm} with $\operatorname{Re}(\mu^{\pm}) \neq 0$, the solution would oscillate while decaying or increasing exponentially. Therefore, the correspondent waves must be attenuating and carrying energy flux. However, in the absence of damping, such waves are still evanescent with the inclined angle of the decay with respect to the y axis - more on this in subsection 3.4.1.

Similarly, the dispersion relations (3.5) can be solved for the unknown μ yielding a characteristic equation with four roots $\mu_{L,S}^{\pm}$ that represent real or complex incoming and outgoing quasi-longitudinal and quasi-shear waves, here denoted as L and S, respectively. Valid plate responses, that is, outgoing waves with μ_X^+ , $X = B_1, B_2, L$ and S either oscillate with a positive energy flux along the y axis or decay exponentially with increasing y; again, we will discuss this in more detail in subsection 3.4.1. In the example to follow below, we will only consider the propagating branches and explain the relation between the directions of incoming and outgoing waves, which follows from the continuity condition $k_x = const$.

For this, we also need the group velocity vector \mathbf{c}_q , which gives the direction

of the wave energy flow [190]. The standard definition $\mathbf{c}_g = \partial \omega / \partial \mathbf{k}$ is not convenient for our purposes here, since the dependence of the angular frequency ω on the wave vector \mathbf{k} is only implicitly given through the dispersion relations (3.4) and (3.5). Instead, one can use

$$(c_{gx}, c_{gy}) = -\left(\frac{\partial \tilde{H}}{\partial k_x}, \frac{\partial \tilde{H}}{\partial k_y}\right) \div \frac{\partial \tilde{H}}{\partial \omega}$$
(3.6)

to find the components of the group velocities c_{gx} and c_{gy} [139, 190] for different modes. Here, $\tilde{H} = \tilde{H}(k_x, k_y, \omega) = 0$ represents the dispersion relations (3.4) or (3.5), respectively, and it is the general Hamilton function that defines the ray trajectories in composite plates; we have seen a form of it for isotropic plates in Equation (2.35).

It is important to note that after introducing the polar coordinate system as $k_x = k \sin \theta$, $k_y = k \cos \theta$, the dispersion relations can be recast as

$$H_l(k,\theta) = \beta_l(\theta)k^l = \omega, \quad l = 1, 2, \qquad (3.7)$$

where l = 1, 2 represents the cases of in-plane and out-of-plane waves, accordingly. Furthermore, the coefficients $\beta(\theta)$ are given as follows:

$$\beta_1^{L,S}(\theta) = \sqrt{\frac{2(C_1C_2 - C_3^2)}{\rho h(C_1 + C_2 \mp \sqrt{(C_1 - C_2)^2 + 4C_3^2}}} \quad , \quad \beta_2(\theta) = \sqrt{\frac{D}{\rho h}} \quad (3.8)$$

with

$$C_{1} = A_{11} \sin^{2} \theta + 2A_{16} \sin \theta \cos \theta + A_{66} \cos^{2} \theta$$

$$C_{2} = A_{66} \sin^{2} \theta + 2A_{26} \sin \theta \cos \theta + A_{22} \cos^{2} \theta$$

$$C_{3} = A_{16} \sin^{2} \theta + (A_{12} + A_{66}) \sin \theta \cos \theta + A_{26} \cos^{2} \theta$$

$$D = D_{11} \sin^{4} \theta + 4D_{16} \sin^{3} \theta \cos \theta + 2 (D_{12} + 2D_{66}) \sin^{2} \theta + \cos^{2} \theta + \cos^{2} \theta + 4D_{26} \sin \theta \cos^{3} \theta + D_{22} \cos^{4} \theta$$
(3.9)

While this form of the dispersion relations is simpler than ones in (3.4) and (3.5), one can only extract the propagating modes out of them. The evanescent modes are essential in the calculation of the scattering coefficients as we shall



Figure 3.3: Schematic of an L-junction of two orthotropic plates. The angle ψ_2 here is set to 90°. Red and blue lines represent the fiber direction of the plates. The local angles of orientations ϕ_1 and ϕ_2 are both set to 45°.

$E_1 \left(\mathrm{N/m^2} \right)$	$E_2 \left(\mathrm{N/m^2} \right)$	$E_3({ m N/m^2})$	$G_{12}(\mathrm{N/m^2})$	$G_{23}(\mathrm{N/m^2})$	$G_{13}(\mathrm{N/m^2})$
121×10^{9}	8.6×10^9	8.6×10^9	4.7×10^9	3.1×10^9	4.7×10^{9}
ν_{12}	ν_{23}	$ u_{13}$	$ ho({\rm kg/m^3})$		
0.27	0.4	0.27	1490		

Table 3.1: Engineering constants of Epoxy Carbon UD (230 GPa) material used for individual laminas of a composite plate.

see in the following sections; hence, Equations (3.4) and (3.5) will be used throughout this chapter. Still, as we will see in Chapter 5, Equation (3.7) is useful for relating the boundary and spatial phase space densities.

3.3.2 Snell's law for composite plates - an example

We will discuss some of the peculiarities of the interplay between incoming and outgoing wave directions at interfaces between composite plates by looking at a specific example. For the sake of clarity, we will consider first generally orthotropic plates meeting at an angle ψ_2 as shown in Figure 3.3. These plates can be viewed as composite laminates that have only one lamina with local ply direction angles ϕ_1 and ϕ_2 . The actual value of ψ_2 is not essential for the


Figure 3.4: Bending wave vector curve in a 45° rotated orthotropic plate at a frequency 3000 Hz (left) and a schematic representation of incoming/outgoing waves at the junction of two identical plates (right). Blue dots represent wave numbers related to outgoing waves while red squares correspond to incoming waves. Wave number components k_x^{max} and k_x^* equal to 62.96 m⁻¹ and 29.34 m⁻¹, respectively.

discussion in this section; it is set to $\psi_2 = 90^\circ$ in Figure 3.3.

The material characteristics of the plates are given in Table 3.1. The thicknesses of plates are both equal to 5 mm. Appendix B describes the relations connecting these material parameters to the stiffness coefficients in Equations (3.1). The orientation of the principal material axes of the two plates with respect to the interface and with respect to each other is important; here, we define the local angles of rotation of the material axes as $\phi_1 = \phi_2 = 45^{\circ}$ as shown in Figure 3.3. We work at a frequency of 3000 Hz here, a frequency value consistent with the thin plate assumptions described by condition (2.43)- at the chosen frequency $k_B^{\max}h = 0.31 < 1$. The left hand side of Figure 3.4 shows the bending wave vector curve at a frequency 3000 Hz obtained from Equation (3.4) in the local coordinate system displaying the 45° rotation. Note that the wave vector curve is the same for both plates in their local coordinate system. The range of k_x values which allow for propagating bending waves at 3000 Hz is between $(-k_x^{\max}, k_x^{\max})$. It is emphasised again that the angle ψ_2 between the two plates can take arbitrary values here; for the sake of clarity, the local coordinates of the plates, (x, y_i) with i = 1 or 2, are drawn in the same plane on the right-hand side of Figure 3.4 and in Figure 3.5 below.

As we are interested in the energy flow across interfaces, the relevant angles of incidence, reflection and transmission are those obtained from the group velocity vectors, Equation (3.6); they point along the gradient vectors to the wave vector curves (denoted by the angles α_i^{\pm} in Figure 3.4). Here, blue symbols correspond to velocity vectors with a positive y component describing waves transporting energy away from the junction, and red symbols describe incoming energy fluxes. The continuity condition $k_x = const$ now connects incoming with outgoing group velocity directions on the wave vector curve: red squares (incoming) with blue dots (outgoing). This leads to a peculiar effect here: in the range $k_x^* < |k_x| < k_x^{\text{max}}$, propagating waves, for example with angles of incidence $\alpha_{1,3}^-$, are transmitted or reflected keeping their direction of travel in the x direction, see the right side of Figure 3.4. This is what one usually finds at refracting interfaces. However, for $-k_x^* < k_x < k_x^*$, a wave, for example with incident angle α_2^- , is scattered reversing its direction in the x-direction, see α_2^+ in Figure 3.4. This gives rise to negative refraction for these values of k_x . The phenomenon is displayed in more detail in Figure 3.5 representing group velocity rays transmitting from one plate to another; the region in blue shows the negative refraction phenomenon.

Furthermore, note that at specific values of $k_x < -k_x^*$, the group velocity vector points in the opposite y direction compared to the corresponding wave vector, see Figure 3.4. This implies that the individual wavefronts travel away from the junction, whereas the wave energy travels towards it. Similar behaviour can be encountered for wave vectors in the right part of the wave vector curve; for specific values of $k_x > k_x^*$ the outgoing wavefronts travel towards the junction whereas the wave energy propagates away from it.

Interesting features can also be seen for in-plane wave vector curves, as shown in Figure 3.6. The shape of the longitudinal wave vector curve is similar to that of the bending wave. Therefore, it also exhibits negative refraction and opposition of phase and group velocity vector directions for specific fixed values



Figure 3.5: Group velocity ray picture for bending waves at junction of two identical 45° rotated orthotropic plates at a frequency 3000 Hz. Blue lines highlight the region of incoming rays, which represents the negative refractive index phenomenon.



Figure 3.6: Longitudinal and shear wave vector curves in a 45° rotated orthotropic plate at a frequency 3000 Hz. Dots represent wave numbers related to outgoing waves while squares - incoming waves. Wave number components k_L^* and k_S^* equal to 6 m^{-1} and 9.3 m^{-1} , respectively - the critical values for longitudinal and shear waves

of k_x . The shear wave vector curve shows additional features: there exists a set of values of k_x (labelled $k_{S_2}^*$), for which there are two pairs of incoming and outgoing shear waves for each k_x . The second pair of shear waves exhibit behaviour similar to what has been described for bending waves. In fact, individual wavefronts at $k_x \in k_{S_2}^*$ travel away from the junction, whereas the wave energy is travelling towards it. Conversely, the wave energy is travelling away from the junction with individual wavefronts travelling towards it at $k_x \in$ $-k_{S_2}^*$. Furthermore, the two pairs of shear waves can couple with each other so that an incoming wave can convert into two distinct shear waves. Finally,



Figure 3.7: Bending wave vector curve in a $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ composite plate at a frequency 3000 Hz (left) and a group velocity ray picture for bending waves at the junction of two identical plates (right). Blue dots represent wave numbers related to outgoing waves while red squares correspond to incoming waves. Wave number components k_x^{\max} and k_x^* equal to 53 m^{-1} and 10.43 m^{-1} , respectively. Blue lines highlight the region of incoming rays, which represents the negative refractive index phenomenon.

propagating longitudinal and shear waves can only exist for k_x values in the range $(-k_L^{\max}, k_L^{\max})$ and $(-k_S^{\max}, k_S^{\max})$, respectively. Beyond these intervals, the corresponding propagating waves become attenuating.



Figure 3.8: Longitudinal and shear wave vector curves in a $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ composite plate at a frequency 3000 Hz. Dots represent wave numbers related to outgoing waves while squares - incoming waves. Wave number components k_L^* and k_S^* equal to 3.71 m^{-1} and 7.8 m^{-1} , respectively - the critical values for longitudinal and shear waves

Now, we consider two identical angle-ply laminates with 5 orthotropic laminas of total thicknesses of 5 mm and with the correspondent lamination schemes $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$. The material of individual laminas is given in Table 3.1, and Section A.2 describes how to compute extensional and bending stiffness coefficients A_{ij} and D_{ij} for governing equations of motion (3.1). The bending wave vector curve is shown on the left-hand side of Figure 3.7. We note that negative refraction occurs in a smaller range $[-k_x^*, k_x^*]$ than in the previous case. This can also be seen on the right-hand side of Figure 3.7. Regarding the in-plane wave vector curves, the shape of the shear wave vector curve changes drastically, see Figure 3.8. Two pairs of incoming and outgoing shear waves are present for larger ranges $k_x \in -k_{S_2}^*$ and $k_x \in k_{S_2}^*$ than in the generally orthotropic plate case. We will come back to this example in subsection 3.7.2.

3.4 Derivation of the dynamic stiffness matrix

Next, we are interested in the local dynamic stiffness matrix for the plate j. This matrix relates boundary forces to boundary displacements produced by the waves emerging from a junction in the form $e^{-ik_x x - i\mu y + i\omega t}$. As discussed in subsection 3.3.1, k_x and ω are common to all plates meeting at an edge "e" and the y components of the wave vectors, μ , are obtained from the dispersion relations (3.4) and (3.5) with solutions forming real or complex pairs μ_x^{\pm} with $X = B_1, B_2, L$ or S.

3.4.1 Energy flow and the response of the plate

The bending wave time-averaged energy flow in the y direction can be written as [191]

$$J_{B} = \frac{1}{2} \operatorname{Re} \left(i\omega \left[w \ \gamma \ \frac{\partial w}{\partial x} \right]^{*} \begin{bmatrix} Q_{y} \\ M_{yy} \\ M_{xy} \end{bmatrix} \right), \qquad (3.10)$$



Figure 3.9: Decay direction angles of various evanescent waves as a function of a ply direction angle in an orthotropic plate with material parameters from Table 3.1.

where * denotes complex conjugation and $\gamma = \frac{\partial w}{\partial y}$ as mentioned in subsection 3.2.1. Now, introducing $w = e^{-ik_x x - i\mu_B y + i\omega t}$ into Equation (3.10) yields

$$J_{B_{1,2}}^{\pm} = \frac{1}{2} \operatorname{Re} \left(i\omega \left[1 - i\mu_{B_{1,2}}^{\pm} - ik_x \right]^* \begin{bmatrix} Q_{y_{1,2}}^{\pm} \\ M_{yy_{1,2}}^{\pm} \\ M_{xy_{1,2}}^{\pm} \end{bmatrix} \right).$$
(3.11)

Furthermore, $Q_{y_{1,2}}^{\pm}$, $M_{yy_{1,2}}^{\pm}$ and $M_{xy_{1,2}}^{\pm}$ are given by Equations (3.2) and (3.3) with $\frac{\partial}{\partial x}$ and $\frac{\partial}{\partial y}$ replaced by $-ik_x$ and $-i\mu_{B_{1,2}}^{\pm}$, respectively.

As stated in subsection 3.3.1, propagating incoming or outgoing waves are those with real wave number components $\mu_{B_1}^{\pm}$ giving rise to an energy flux to or away from the junction, that is, $J_{B_1}^{\pm} < 0$ or $J_{B_1}^{\pm} > 0$, respectively. If the $\mu_{B_{1,2}}^{\pm}$ are complex, then the out-of-plane displacements of outgoing or incoming waves are of the form

$$w^{\pm} = e^{-\mathrm{i}k_x x \mp \mathrm{i}\operatorname{Re}(\mu^{\pm})y \mp \operatorname{Im}(\mu^{\pm})y + \mathrm{i}\omega t} .$$
(3.12)

and hence, they appear to be oscillating while decaying away or increasing exponentially towards the junction. This seems to cause energy transport as well, since the corresponding energy flux $J_{B_{1,2}}^{\pm} \neq 0$. However, as we mentioned in subsection 3.3.1, these waves are still evanescent and can only exponentially



Figure 3.10: Contour plot of the real part of the out-of-plane displacement produced by the outgoing evanescent bending wave in a 45° -rotated orthotropic plate plotted in the (x, y) plane. Brighter colours represent positive z values, whilst darker ones denote negative z values.

decay away or increase towards the junction with the decay/increase direction axis being not aligned with the y axis, in general [192, 193]. The angle of decay/increase of evanescent waves depends on the angle of orientation of principal material axes of plies of the plate, i.e. on the ply direction angle. It can be computed as $\arctan(\operatorname{Re}(\mu^{\pm})/k_x)$. Figure 3.9 presents the decay direction angles of longitudinal, shear and bending evanescent waves with respect to the y axis as a function of the ply direction angle of an orthotropic plate with material parameters from Table 3.1. It can be noted that the decay of evanescent waves can be inclined with respect to the y axis with angles up to 60° ; for example, the line corresponding to the evanescent shear wave in Figure 3.9. We can see that the decay angles are zero for ply direction of angles 0° and $\pm 90^{\circ}$, that is, the evanescent waves decay/increase along the yaxis, and the correspondent $J_{B_{1,2}}^{\pm} = 0$, as expected. In such cases, the plate is *specially orthotropic* if it consists of only one layer (see Section A.1 for details) or balanced if it consists of multiple layers [1].

Figure 3.10 presents the contour plot of the real part of the out-of-plane displacement field created by the outgoing bending evanescent wave in a 45° rotated orthotropic plate. The displacement at y = 0 along the x axis is oscillating, as expected. In contrast, at y > 0 it decays away along the inclined null-lines, i.e., lines at which displacement is zero (black straight lines in Figure 3.10). This leads to the oscillating displacement shape projection along the y axis, which is why the energy flux along the y axis J_B^{\pm} appears to be non-zero.

Once the appropriate bending wave number roots $\mu_{B_{1,2}}^+$ are defined, the outof-plane response of the plate can be expressed as

$$w^{+} = \alpha_{B_{1}}^{+} e^{-ik_{x}x - i\mu_{B_{1}}^{+}y + i\omega t} + \alpha_{B_{2}}^{+} e^{-ik_{x}x - i\mu_{B_{2}}^{+}y + i\omega t}, \qquad (3.13)$$

where the constants $\alpha_{B_1}^+$ and $\alpha_{B_2}^+$ are amplitudes of the outgoing bending waves. For the in-plane motion, the response of the plate takes the form

$$v = \Phi_v e^{-ik_x x - i\mu y + i\omega t}$$

$$u = \Phi_u e^{-ik_x x - i\mu y + i\omega t}.$$
(3.14)

Equation (3.5) can be solved for the unknown μ yielding a characteristic equation with four roots $\mu_{L,s}^{\pm}$ that represent incoming and outgoing quasilongitudinal and quasi-shear waves, here denoted again as L and S, respectively. Similar to the out-of-plane case, a valid outgoing solution produces a positive energy flow in the y direction if the correspondent wave is propagating, or it is associated with complex-valued $\mu_{L,s}^+$ with $\operatorname{Im}(\mu_{L,s}^+) < 0$; that is, the correspondent wave is exponentially decaying as $y \to \infty$. Note that purely imaginary solutions of Equation (3.5) $\mu_{L,s}^+$, which are present in specially orthotropic plates, produce no energy flow in the y direction. On the other hand, complex-valued solutions can appear to produce energy flow along the y axis, as discussed previously in the out-of-plane case.

The time-averaged energy flow in the y direction of in-plane waves can be generally written as

$$J = \frac{1}{2} \operatorname{Re} \left(i\omega \begin{bmatrix} v & u \end{bmatrix}^* \begin{bmatrix} N_{yy} \\ N_{xy} \end{bmatrix} \right).$$
(3.15)

Introducing Equations (3.14) into Equation (3.15) yields the energy flow expressions for the longitudinal and shear wave modes, that is,

$$J_{L}^{\pm} = \frac{1}{2} \operatorname{Re} \left(i\omega \left[1 \ \Phi_{L}^{\pm} \right]^{*} \begin{bmatrix} N_{yy_{L}}^{\pm} \\ N_{xy_{L}}^{\pm} \end{bmatrix} \right) \quad , \quad J_{S}^{\pm} = \frac{1}{2} \operatorname{Re} \left(i\omega \left[\Phi_{S}^{\pm} \ 1 \right]^{*} \begin{bmatrix} N_{yy_{S}}^{\pm} \\ N_{xy_{S}}^{\pm} \end{bmatrix} \right)$$

$$(3.16)$$

with

$$\Phi_{L}^{\pm} = -\frac{A_{26} \mu_{L}^{\pm^{2}} + k_{x} \mu_{L}^{\pm} (A_{12} + A_{66}) + A_{16} k_{x}^{2}}{A_{66} \mu_{L}^{\pm^{2}} + 2 A_{16} k_{x} \mu_{L}^{\pm} + A_{11} k_{x}^{2} - \rho h \omega^{2}}$$

$$\Phi_{S}^{\pm} = -\frac{A_{26} \mu_{S}^{\pm^{2}} + k_{x} \mu_{S}^{\pm} (A_{12} + A_{66}) + A_{16} k_{x}^{2}}{A_{22} \mu_{S}^{\pm^{2}} + 2 A_{26} k_{x} \mu_{S}^{\pm} + A_{66} k_{x}^{2} - \rho h \omega^{2}},$$
(3.17)

where $\begin{bmatrix} 1 & \Phi_L^{\pm} \end{bmatrix}^T$ and $\begin{bmatrix} \Phi_S^{\pm} & 1 \end{bmatrix}^T$ are the eigenvectors in the Φ_v, Φ_u basis corresponding to the wave numbers μ_L^{\pm} and μ_S^{\pm} for incoming and outgoing modes. $N_{yy_{L,S}}^{\pm}$ and $N_{xy_{L,S}}^{\pm}$ are given by substituting $\frac{\partial}{\partial x}$ and $\frac{\partial}{\partial y}$ by $-ik_x$ and $-i\mu_{L,S}^{\pm}$ in Equations (3.2), respectively.

The in-plane response of the plate can be written as

$$\begin{bmatrix} v^+ \\ u^+ \end{bmatrix} = \alpha_L^+ \begin{bmatrix} 1 \\ \Phi_L^+ \end{bmatrix} e^{-ik_x x - i\mu_L^+ y + i\omega t} + \alpha_S^+ \begin{bmatrix} \Phi_S^+ \\ 1 \end{bmatrix} e^{-ik_x x - i\mu_S^+ y + i\omega t}, \qquad (3.18)$$

where α_L^+ and α_S^+ are the amplitudes of outgoing quasi-longitudinal and quasishear waves. This particular choice of eigenvectors ensures the correct representation of the displacement field in the case of $k_x = 0$ and $A_{16} = A_{26} = 0$, that is, when the incident plane wave vector is directed normal to the junction and the plate is *quasi*-specially orthotropic. Then $\Phi_L^+ = \Phi_S^+ = 0$, and the response of the plate can be expressed as

$$v^{+} = \alpha_{L}^{+} e^{-i\tilde{\mu}_{L}^{+}y + i\omega t}$$

$$u^{+} = \alpha_{s}^{+} e^{-i\tilde{\mu}_{S}^{+}y + i\omega t}.$$
(3.19)

The values $\tilde{\mu}_L^+ = \sqrt{\rho \omega^2 / Q_{22}}$ and $\tilde{\mu}_S^+ = \sqrt{\rho \omega^2 / Q_{66}}$ agree then with the wave numbers of purely longitudinal and shear waves, respectively, and the in-plane

response of the plate consists of purely longitudinal and shear displacements in the y and x directions.

3.4.2 The dynamic stiffness matrix

Using Equation (3.13), one can define the out-of-plane displacement w^+ and rotation γ^+ at the common intersection "e" at y = 0 in terms of the amplitudes $\alpha^+_{B_1}$ and $\alpha^+_{B_2}$, that is,

$$\begin{bmatrix} w_{\rm e}^+ \\ \gamma_{\rm e}^+ \end{bmatrix} = \begin{bmatrix} 1 & 1 \\ -i\mu_{B_1}^+ & -i\mu_{B_2}^+ \end{bmatrix} \begin{bmatrix} \alpha_{B_1}^+ \\ \alpha_{B_2}^+ \end{bmatrix} e^{-ik_x x + i\omega t}.$$
 (3.20)

The elastic tractions involving the out-of-plane displacement can be written in terms of the same amplitudes $\alpha_{B_1}^+$ and $\alpha_{B_2}^+$ by inserting (3.13) in (3.2), that is,

$$V_{y_{e}}^{+} = \sum_{m=1}^{2} \left(D_{22} \,\mathrm{i}\mu_{B_{m}}^{+3} - 4D_{26} \,\mu_{B_{m}}^{+2} \,\mathrm{i}k_{x} - (D_{12} + 4D_{66}) \,\mathrm{i}\mu_{B_{m}}^{+} k_{x}^{2} - 2D_{16} \,\mathrm{i}k_{x}^{3} \right) \alpha_{B_{m}}^{+} e^{-\mathrm{i}k_{x}x + \mathrm{i}\omega t}$$

$$M_{yy_{e}}^{+} = -\sum_{m=1}^{2} \left(D_{22} \,\mu_{B_{m}}^{+2} + D_{12} \,k_{x}^{2} + 2D_{26} \,\mu_{B_{m}}^{+} \,k_{x} \right) \alpha_{B_{m}}^{+} e^{-\mathrm{i}k_{x}x + \mathrm{i}\omega t}.$$
(3.21)

Finally, combining Equations (3.20) with Equations (3.21) yields a relation between the elastic tractions $V_{y_e}^+$, $M_{yy_e}^+$ and the edge displacement w_e^+ and rotation γ_e^+ at y = 0, that is,

$$\begin{bmatrix} V_{y_{e}}^{+} \\ M_{yy_{e}}^{+} \end{bmatrix} = K_{w\gamma}^{+} \begin{bmatrix} w_{e}^{+} \\ \gamma_{e}^{+} \end{bmatrix}$$
(3.22)

(3.23)

with

$$\begin{split} K_{w\gamma}^{+}(1,1) &= -D_{22} \,\mathrm{i}\mu_{B_{1}}^{+}\mu_{B_{2}}^{+} \left(\mu_{B_{1}}^{+} + \mu_{B_{2}}^{+}\right) + 4D_{26} \,\mu_{B_{1}}^{+}\mu_{B_{2}}^{+} \,\mathrm{i}k_{x} - 2D_{26} \,\mathrm{i}k_{x}^{3} \\ K_{w\gamma}^{+}(1,2) &= -D_{22} \left(\mu_{B_{1}}^{+2} + \mu_{B_{2}}^{+2} + \mu_{B_{1}}^{+}\mu_{B_{2}}^{+}\right) + 4D_{26} \left(\mu_{B_{1}}^{+} + \mu_{B_{2}}^{+}\right) k_{x} + \left(D_{12} + 4D_{66}\right) k_{x}^{2} \\ K_{w\gamma}^{+}(2,1) &= D_{22} \,\mu_{B_{1}}^{+}\mu_{B_{2}}^{+} - D_{12} k_{x}^{2} \\ K_{w\gamma}^{+}(2,2) &= -D_{22} \,\mathrm{i} \left(\mu_{B_{1}}^{+} + \mu_{B_{2}}^{+}\right) - 2D_{26} \mathrm{i}k_{x} \end{split}$$

 $K_{w\gamma}^+$ is a block part of the dynamic stiffness matrix for out-of-plane displacement w_e^+ and rotation γ_e^+ .

Next, we consider the corresponding part for in-plane motion. The in-plane displacements $v_{\rm e}^+$ and $u_{\rm e}^+$ at the shared edge are given as

$$\begin{bmatrix} v_{\rm e}^+ \\ u_{\rm e}^+ \end{bmatrix} = \begin{bmatrix} 1 & \Phi_s^+ \\ \Phi_L^+ & 1 \end{bmatrix} \begin{bmatrix} \alpha_L^+ \\ \alpha_s^+ \end{bmatrix} e^{-\mathrm{i}k_x x + \mathrm{i}\omega t} .$$
(3.24)

Now, inserting (3.18) into (3.2) as

$$N_{yy_{e}}^{+} = -i \left(A_{22} \mu_{L}^{+} + A_{12} \Phi_{L}^{+} k_{x} + A_{26} (\Phi_{L}^{+} \mu_{L}^{+} + k_{x}) \right) \alpha_{L}^{+} e^{-ik_{x}x + i\omega t}$$

$$- i \left(A_{22} \Phi_{S}^{+} \mu_{S}^{+} + A_{12} k_{x} + A_{26} (\mu_{S}^{+} + \Phi_{S}^{+} k_{x}) \right) \alpha_{S}^{+} e^{-ik_{x}x + i\omega t}$$

$$N_{xy_{e}}^{+} = -i \left(A_{26} \mu_{L}^{+} + A_{16} \Phi_{L}^{+} k_{x} + A_{66} (\Phi_{L}^{+} \mu_{L}^{+} + k_{x}) \right) \alpha_{L}^{+} e^{-ik_{x}x + i\omega t}$$

$$- i \left(A_{26} \Phi_{S}^{+} \mu_{S}^{+} + A_{16} k_{x} + A_{66} (\mu_{S}^{+} + \Phi_{S}^{+} k_{x}) \right) \alpha_{S}^{+} e^{-ik_{x}x + i\omega t}$$

(3.25)

and using (3.24) yield a relation between the in-plane tractions and the displacements of the outgoing in-plane waves, that is,

$$\begin{bmatrix} N_{yy_{e}}^{+} \\ N_{xy_{e}}^{+} \end{bmatrix} = K_{vu}^{+} \begin{bmatrix} v_{e}^{+} \\ u_{e}^{+} \end{bmatrix}$$
(3.26)

with

$$K_{vu}^{+}(1,1) = cA_{22}(\mu_{L}^{+} - \Phi_{L}^{+}\Phi_{S}^{+}\mu_{S}^{+}) + A_{26}(c\Phi_{L}^{+}(\mu_{L}^{+} - \mu_{S}^{+}) + k_{x})$$

$$K_{vu}^{+}(1,2) = cA_{26}(\mu_{S}^{+} - \Phi_{L}^{+}\Phi_{S}^{+}\mu_{L}^{+}) + cA_{22}\Phi_{S}^{+}(\mu_{S}^{+} - \mu_{L}^{+}) + A_{12}k_{x}$$

$$K_{vu}^{+}(2,1) = cA_{26}(\mu_{L}^{+} - \Phi_{L}^{+}\Phi_{S}^{+}\mu_{S}^{+}) + A_{66}(c\Phi_{L}^{+}(\mu_{L}^{+} - \mu_{S}^{+}) + k_{x})$$

$$K_{vu}^{+}(2,2) = cA_{66}(\mu_{S}^{+} - \Phi_{L}^{+}\Phi_{S}^{+}\mu_{L}^{+}) + cA_{26}\Phi_{S}^{+}(\mu_{S}^{+} - \mu_{L}^{+}) + A_{16}k_{x}$$
(3.27)

where $c = \frac{1}{i(1 - \Phi_L^+ \Phi_S^+)}$. The matrix K_{vu}^+ is the part of the dynamic stiffness matrix related to in-plane motion.

We now define the elastic tractions and displacements at the junction edge as $F = (N_{yy_e}, N_{xy_e}, V_{y_e}, M_{yy_e})^T \text{ and } U = (v_e, u_e, w_e, \gamma_e)^T \text{ and write}$

$$F_j^+ = K_j^+ U_j^+ (3.28)$$

for the relation between the displacements of an outgoing wave at the edge of plate j and the associate forces, where the 4×4 dynamic stiffness matrix K_j^+ is a block matrix defined through (3.22) and (3.26). An associated stiffness matrix K_j^- relating displacements of the incoming waves U_j^- with forces F_j^- can be obtained from K_j^+ by changing the corresponding wave number components μ^+ by μ^- in the sub-matrices related to in-plane and out-of-plane motion. In what follows, it will also be important to consider the relation between displacements and mode amplitudes. We denote $A^{\pm} = (\alpha_L^{\pm}, \alpha_S^{\pm}, \alpha_{B_1}^{\pm}, \alpha_{B_2}^{\pm})^T$ as the vector of amplitudes of incoming or outgoing modes, and write

$$U_j^{\pm} = H_j^{\pm} A_j^{\pm} \tag{3.29}$$

on the edge of plate j, where H_j^{\pm} is the block-diagonal matrix obtained from (3.20) and (3.24). Here, H_j^{-} is obtained from H_j^{+} by changing the wave number components μ^{+} by μ^{-} . In the next step, we will derive the global dynamic stiffness matrices of each plate junction and associated scattering matrices using force equilibrium conditions and continuity conditions at the junction.

3.5 Reflection and transmission at plate junctions

Assuming no external forces are applied at the junction, one can write the force equilibrium and continuity conditions at an edge shared between N plates as

$$\sum_{j=1}^{N} R_j F_j = 0, (3.30)$$

$$U_j = R_j^T U \quad \text{for all } j = 1, \dots, N,$$
(3.31)

where $F_j = F_j^+ + F_j^-$ and $U_j = U_j^+ + U_j^-$ is the total force and displacement at the edge of plate *j*. *U* denotes the displacement common to all plates (continuity) and R_j is the rotation matrix from the local coordinate system (x_j,y_j,z_j) to the global coordinate system $(x_{\rm g},y_{\rm g},z_{\rm g}),$ that is,

$$R_{j} = \begin{bmatrix} \cos \psi_{j} & 0 & -\sin \psi_{j} & 0 \\ 0 & 1 & 0 & 0 \\ \sin \psi_{j} & 0 & \cos \psi_{j} & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}.$$
 (3.32)

Rewriting (3.30) using (3.28), we obtain

$$\sum_{j=1}^{N} R_j K_j^+ U_j^+ = -\sum_{j=1}^{N} R_j K_j^- U_j^-$$
(3.33)

$$\Rightarrow \sum_{j=1}^{N} R_j K_j^+ U_j = \sum_{j=1}^{N} R_j \left(K_j^+ - K_j^- \right) U_j^-.$$
(3.34)

Using (3.31), we can now deduce the common displacement vector U as a function of the incoming waves, that is,

$$U = \left(\sum_{j=1}^{N} R_j K_j^+ R_j^T\right)^{-1} \sum_{n=1}^{N} R_n \left(K_n^+ - K_n^-\right) U_n^-.$$
(3.35)

Inserting (3.35) into the N matrix equations (3.31), one obtains for $m = 1, \ldots N$

$$U_m^+ = R_m^T \left(\sum_{j=1}^N R_j K_j^+ R_j^T\right)^{-1} \left(\sum_{n=1}^N R_n \left(K_n^+ - K_n^-\right) U_n^-\right) - U_m^-, \quad (3.36)$$

and writing this in terms of the mode amplitudes using (3.29), we obtain

$$A_{m}^{+} = \left(H_{m}^{+}\right)^{-1} R_{m}^{T} \left(\sum_{j=1}^{N} R_{j} K_{j}^{+} R_{j}^{T}\right)^{-1} \left(\sum_{n=1}^{N} R_{n} \left(K_{n}^{+} - K_{n}^{-}\right) H_{n}^{-} A_{n}^{-}\right) - \left(H_{m}^{+}\right)^{-1} H_{m}^{-} A_{m}^{-}.$$

$$(3.37)$$

Equation (3.37) gives relations between incoming and outgoing wave mode amplitudes and can be interpreted as defining the matrix elements of a $4N \times 4N$ scattering matrix S. We note that by setting N = 1 in Equation (3.37), we can compute wave reflection coefficients of different modes at the free edge of a composite plate. In fact, the elastic tractions $F_j = F_j^+ + F_j^-$ are equal to zero at the free edge of the *j*th plate. Using Equations (3.28) and (3.29), we can write

$$K_{j}^{+}H_{j}^{+}A_{j}^{+} + K_{j}^{-}H_{j}^{-}A_{j}^{-} = 0$$

$$A_{j}^{+} = -\left(K_{j}^{+}H_{j}^{+}\right)^{-1}K_{j}^{-}H_{j}^{-}A_{j}^{-}.$$
(3.38)

From the other side, setting N = 1 in Equation (3.37) yields

$$A_{j}^{+} = (H_{j}^{+})^{-1} (K_{j}^{+})^{-1} (K_{j}^{+} - K_{j}^{-}) H_{j}^{-} A_{j}^{-} - (H_{j}^{+})^{-1} H_{j}^{-} A_{j}^{-}$$

$$A_{j}^{+} = - (H_{j}^{+})^{-1} (K_{j}^{+})^{-1} K_{j}^{-} H_{j}^{-} A_{j}^{-}$$
(3.39)

which is exactly the same expression as in Equation (3.38).

Finally, we compute the energy scattering coefficients, that is, the ratio between outgoing and incident energy fluxes. Writing the matrix elements of Sin the form $s_{ij}^{nm}(\omega, k_x)$ for an incoming wave of type i in plate n and a reflected or transmitted wave of type j in plate m (at angular frequency ω and wave number component k_x), we obtain for the associated energy fluxes

$$t_{ij}^{nm}(\omega, k_x) = \begin{cases} \frac{J_{j,m}^+}{J_{i,n}^-} |s_{ij}^{nm}|^2 & \text{if wave } j \text{ is propagating.} \\ 0 & \text{otherwise.} \end{cases}$$
(3.40)

Here, $J_{i,n}^{-}(J_{j,m}^{+})$ is the incoming (outgoing) energy flux of type i (j) on plate n (m) given by either (3.16) for in-plane modes or (3.11) for out-of-plane motion. It is noted here that the sum of energy scattering coefficients over the outgoing modes equals one, that is,

$$\sum_{m=1}^{N} \sum_{j} t_{ij}^{nm} = 1 \tag{3.41}$$

due to energy conservation. We will use this relation as a check when considering numerical applications of the method developed.

3.6 Stiffened plate

This section considers several composite plates finite in their respective y coordinate directions mounted onto a ground composite plate from one or both sides. Figure 3.11 describes the configuration of a plate with m stiffeners at-



Figure 3.11: A ground composite plate with m stiffener plates attached onto it. The ground plate is considered infinite in both directions, whereas stiffeners are infinite only in their local x coordinate directions.



Figure 3.12: The schematic of incoming (shown in red) and outgoing (shown in blue) waves travelling in the stiffened plate. A_1^- and A_N^- are the incoming waves from infinity.

tached to the ground plate from its top side, thus forming T-shaped junctions locally. The lengths of stiffener plates are h_k , k = 1, ..., m, and distances between stiffeners are l_k , k = 1, ..., m-1. For the case of stiffeners attached from both sides of a ground plate, we assume that they are symmetrically placed around the ground plate and made of the same material - minor changes in the method's development would be needed to remove this assumption. We are interested in the scattering coefficients of incoming waves travelling from infinity towards the plate with stiffeners taking into account that the excitation can also enter into stiffeners and get reflected at their respective free ends as shown with red and blue arrows in Figure 3.12. This leads to resonance phenomena, that is, at certain wave number component and angular frequency values k_x and ω , a perfect transmission or reflection of incoming wave energy can occur. These effects were previously presented and analysed for beams with the symmetric constraint [194], for beams with periodic stiffeners [195] and for plates with periodic stiffeners [196]. We discuss the conditions at which resonances are attained in the last parts of subsections 3.6.1, 3.6.2 and 3.6.3. As previously stated, we want to compute an effective scattering matrix S_{eff} of the following form

$$\begin{bmatrix} A_1^+ \\ A_N^+ \end{bmatrix} = S_{\text{eff}} \begin{bmatrix} A_1^- \\ A_N^- \end{bmatrix}, \qquad (3.42)$$

where $A_{1,N}^{\pm}$ are the amplitudes of incoming/outgoing waves travelling from infinity towards the first junctions of the stiffened plate that the waves meet. Note that this particular setup is based on the short wavelength assumption. Specifically, the longest wavelengths of various modes that can propagate in the plate need to be smaller than the stiffener spacings and lengths. This assumption allows to neglect an always attenuating bending wave B_2 in the derivation of the effective scattering matrix since it rapidly decays along stiffener lengths and distances between them, thus carrying no wave energy - it is still used for the calculation of junction scattering coefficients as per Equation (3.37). We denote the propagating bending waves as B (instead of B_1) for simplicity.

In order to compute S_{eff} , we first derive an effective scattering matrix for a plate with one or a pair of stiffeners mounted on it in subsections 3.6.1 and 3.6.2. Using calculated effective scattering matrices, we derive an effective scattering matrix for a plate with two stiffeners or two pairs of stiffeners in subsection 3.6.3. Recursively, we compute an effective scattering matrix for a composite plate with several stiffeners. Finally, the effective energy scattering coefficients $t_{ij,\text{eff}}^{nm}$ can be computed as

$$t_{ij,\text{eff}}^{nm}(\omega,k_x) = \begin{cases} \frac{J_{j,m}^+}{J_{i,n}^-} |s_{ij,\text{eff}}^{nm}|^2 & \text{if wave } j \text{ is propagating.} \\ 0 & \text{otherwise.} \end{cases} \quad n,m = 1,N$$

$$(3.43)$$



Figure 3.13: A schematic view of a stiffened plate in the yz plane. Here, a_{ij}^- represent the amplitudes of incoming waves, and a_{ij}^+ , the amplitudes of outgoing waves.

3.6.1 A composite plate with one stiffener

We give here the details for computing the scattering coefficients for a plate with a stiffener k attached. The stiffener plate has a finite length of h_k . We are interested in the reflection and transmission coefficients between points 1 and 2 in Figure 3.13 considering that the excitation can also enter the stiffener at point 4 and being reflected at the free end at point 3. In order to compute the energy scattering coefficients (3.40) relating plate one and two, one needs to compute the associate scattering matrix s_{eff} from the scattering matrix at junction 4, but including waves travelling into the stiffener and being reflected at the end of the stiffener at 3.

Following the treatment in [197] extended to the elastodynamic case in [198], we now introduce the amplitudes of incoming and outgoing waves at each of the plate segments shown in Figure 3.13, that is,

$$a_{ij}^{\pm} = \begin{bmatrix} a_L^{\pm} \\ a_S^{\pm} \\ a_B^{\pm} \end{bmatrix}_{ij} \quad i, j \in (1, 2, 3, 4), \qquad (3.44)$$

where the subscript ij is related to a wave travelling from i to j. Following

this notation, we write the desired effective scattering matrix s_{eff} as

$$\begin{bmatrix} a_{41}^+ \\ a_{42}^+ \end{bmatrix} = s_{\text{eff}} \begin{bmatrix} a_{14}^- \\ a_{24}^- \end{bmatrix} .$$
(3.45)

From Equations (3.37) and (3.38), we can write the two scattering matrices at the junction 4 and free edge 3 as

$$\begin{bmatrix} a_{41}^+ \\ a_{42}^+ \\ a_{43}^+ \end{bmatrix} = S^{(4)} \begin{bmatrix} a_{14}^- \\ a_{24}^- \\ a_{34}^- \end{bmatrix} = \begin{bmatrix} \rho_{11} & \tau_{21} & \tau_{31} \\ \tau_{12} & \rho_{22} & \tau_{32} \\ \tau_{13} & \tau_{23} & \rho_{33} \end{bmatrix} \begin{bmatrix} a_{14}^- \\ a_{24}^- \\ a_{34}^- \end{bmatrix}$$
(3.46)

and

$$a_{34}^+ = S^{(3)}a_{43}^- = \tilde{\rho}_{33}a_{43}^- \,. \tag{3.47}$$

The sub-matrices in (3.46) contain the scattering coefficients s_{ij}^{nm} from plate n to m of mode type i to j in the form

$$\rho_{nn} = \begin{bmatrix} s_{LL}^{nn} & s_{LS}^{nn} & s_{BL}^{nn} \\ s_{LS}^{nn} & s_{SS}^{nn} & s_{BS}^{nn} \\ s_{LB}^{nn} & s_{SB}^{nn} & s_{BB}^{nn} \end{bmatrix}, \quad \tau_{nm} = \begin{bmatrix} s_{LL}^{nm} & s_{LS}^{nm} & s_{BL}^{nm} \\ s_{LS}^{nm} & s_{SS}^{nm} & s_{BS}^{nm} \\ s_{LB}^{nm} & s_{SB}^{nm} & s_{BB}^{nm} \end{bmatrix}. \quad (3.48)$$

Note that $\tilde{\rho}_{33}$ in Equation (3.47) denotes the reflection matrix at node 3 and is different from ρ_{33} corresponding to reflection of waves on node 4 incoming from plate 3.

The first two rows of matrix equation in (3.46) can render the effective scattering matrix s_{eff} if amplitudes a_{34}^- are expressed in terms of amplitudes a_{14}^- and a_{24}^- . To accomplish that, we first note that the amplitudes a_{43}^+ of outgoing waves are related to the amplitudes a_{43}^- of the incoming waves as

$$a_{43}^{-} = P a_{43}^{+}$$
, $P = \text{diag}\left(\exp\left(-i\mu^{+}h_{k}\right)\right)$, (3.49)

where $\mu^+ = \left[\mu_L^+ \ \mu_S^+ \ \mu_B^+\right]^T$ and diag $\left(\exp\left(-i\mu^+h_k\right)\right)$ represents the diagonal matrix with $\exp\left(-i\mu^+h_k\right)$ on its diagonal (see [197], [198]). The same applies



Figure 3.14: A schematic view of a stiffener plate in the xy plane. Blue lines represent the direction of the *i*th ply of the plate.

for the pair of amplitudes of outgoing and incoming waves a_{34}^+ and a_{34}^- , that is,

$$a_{34}^- = \tilde{P} a_{34}^+$$
, $\tilde{P} = \text{diag}\left(\exp\left(-i\tilde{\mu}^+ h_k\right)\right).$ (3.50)

We note that $\tilde{\mu}^+$ are computed in the local coordinate system of the semiinfinite plate with the free edge. If we label the local ply direction angles of the semi-infinite plate at point 4 as $\phi_1^{43}/\phi_2^{43}/\ldots/\phi_n^{43}$, then the local ply direction angles of the semi-infinite plate at point 3 must satisfy

$$\phi_i^{34} = -\phi_i^{43}, i = 1, \dots, n \tag{3.51}$$

to ensure that the stiffener plate is uniform and that waves with amplitudes $a_{43(34)}^-$ and $a_{43(34)}^+$ have the same angles of propagation, see Figure 3.14. This entails

$$\tilde{\mu}^+ \neq \mu^+ \Rightarrow \tilde{\mathbf{P}} \neq \mathbf{P} \quad \text{if} \quad \phi_i^{43(34)} \neq 0^\circ, 90^\circ \quad , i = 1, \dots, n \,.$$
 (3.52)

Now, to eliminate a_{34}^- in the matrix equation (3.46), we consider its third row,

that is,

$$a_{43}^{+} = \tau_{13}a_{14}^{-} + \tau_{23}a_{24}^{-} + \rho_{33}a_{34}^{-}.$$
(3.53)

By using (3.49), (3.50) and (3.47), one obtains

$$a_{34}^- = \dot{\mathbf{P}} \,\tilde{\rho}_{33} \mathbf{P} \,a_{43}^+, \tag{3.54}$$

which combined with (3.53) yields the following matrix equation for a_{34}^- :

$$\left(\mathbf{I} - \tilde{\mathbf{P}}\,\tilde{\rho}_{33}\mathbf{P}\,\rho_{33}\right)a_{34}^{-} = \tilde{\mathbf{P}}\,\tilde{\rho}_{33}\mathbf{P}\left(\tau_{13}a_{14}^{-} + \tau_{23}a_{24}^{-}\right)\,,\tag{3.55}$$

where I is a 3-by-3 identity matrix. Finally, by introducing the solution of Equation (3.55) into Equation (3.46), we can define the effective scattering matrix s_{eff} as

$$\begin{bmatrix} a_{41}^+ \\ a_{42}^+ \end{bmatrix} = s_{\text{eff}} \begin{bmatrix} a_{14}^- \\ a_{24}^- \end{bmatrix} = \begin{bmatrix} \rho_{11}^* & \tau_{21}^* \\ \tau_{12}^* & \rho_{22}^* \end{bmatrix} \begin{bmatrix} a_{14}^- \\ a_{24}^- \end{bmatrix}$$
(3.56)

with

$$\rho_{11}^{*} = \rho_{11} + \tau_{31} \left(\mathbf{I} - \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \tau_{13}$$

$$\tau_{21}^{*} = \tau_{21} + \tau_{31} \left(\mathbf{I} - \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \tau_{23}$$

$$\tau_{12}^{*} = \tau_{12} + \tau_{32} \left(\mathbf{I} - \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \tau_{13}$$

$$\rho_{22}^{*} = \rho_{22} + \tau_{32} \left(\mathbf{I} - \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \, \tilde{\rho}_{33} \mathbf{P} \, \tau_{23}$$
(3.57)

The effective energy scattering coefficients $t_{ij,\text{eff}}^{nm}$, $n, m \in \{1, 2\}$ can be computed from s_{eff} using (3.43). Note that $\tilde{P} \tilde{\rho}_{33} P \rho_{33}$ is sub-unitary due to the sub-unitarity of ρ_{33} , and s_{eff} is thus not singular.

A resonance condition can be formulated, that is, resonances are attained at wave numbers k_x and frequencies ω values giving rise to local minima of

$$\left|\det\left(\mathbf{I}-\tilde{\mathbf{P}}\,\tilde{\rho}_{33}\mathbf{P}\,\rho_{33}\right)\right|.\tag{3.58}$$



Figure 3.15: A schematic view of a plate with a pair of finite stiffener plates in the yz plane. Here, a_{ij}^- represent the amplitudes of incoming waves, and a_{ij}^+ , the amplitudes of outgoing waves.

3.6.2 A composite plate with a pair of stiffeners

In this subsection, we consider a composite ground plate stiffened with two stiffener plates symmetrically attached from both sides, forming a cross-shaped junction. This particular system prevents the generation of outgoing in-plane waves from an incoming bending wave [196], and therefore, the corresponding effective scattering matrix has null entries $s_{BL,eff}^{11}$ and $s_{BS,eff}^{12}$. Figure 3.15 describes the configuration of plates in the yz plane. As in the previous case, waves with amplitudes a_{15}^{-} impinging on the junction point 5 can be transmitted into plate 2 with amplitudes a_{52}^{+} , can be reflected with amplitudes a_{15}^{-} and can be transmitted to the stiffener plates with amplitudes a_{53}^{+} and a_{54}^{+} . At the free ends of the stiffeners, waves reflect and become incoming on the junction point 5 again, thus creating further scattering. Accordingly, we are interested in the reflection and transmission coefficients between points 1 and 2, that is,

$$\begin{bmatrix} a_{51}^+ \\ a_{52}^+ \end{bmatrix} = s_{\text{eff}} \begin{bmatrix} a_{15}^- \\ a_{25}^- \end{bmatrix} .$$
(3.59)

Following the approach described for the case of one stiffener plate in the previous subsection, we express the scattering matrices at the junction point 5 and free ends 3 and 4 as

$$\begin{bmatrix} a_{51}^{+} \\ a_{52}^{+} \\ a_{53}^{+} \\ a_{54}^{+} \end{bmatrix} = S^{(5)} \begin{bmatrix} a_{15}^{-} \\ a_{25}^{-} \\ a_{35}^{-} \\ a_{45}^{-} \end{bmatrix} = \begin{bmatrix} \rho_{11} & \tau_{21} & \tau_{31} & \tau_{41} \\ \tau_{12} & \rho_{22} & \tau_{32} & \tau_{42} \\ \tau_{13} & \tau_{23} & \rho_{33} & \tau_{43} \\ \tau_{14} & \tau_{24} & \tau_{34} & \rho_{44} \end{bmatrix} \begin{bmatrix} a_{15}^{-} \\ a_{25}^{-} \\ a_{35}^{-} \\ a_{45}^{-} \end{bmatrix}, \quad a_{45}^{+} = S^{(3)}a_{53}^{-} = \tilde{\rho}_{33}a_{53}^{-} \\ a_{45}^{-} \end{bmatrix}$$
(3.60)

Expressing amplitudes a_{35}^- and a_{45}^- in terms of amplitudes a_{15}^- and a_{25}^- would yield the scattering matrix s_{eff} from the first matrix equation in (3.60). To accomplish that, we first note that similarly as in Equations (3.49) and (3.50), we can relate the amplitudes $a_{53(35)}^-$ and $a_{54(45)}^-$ with $a_{53(35)}^+$ and $a_{54(45)}^+$ as follows

$$a_{53}^{-} = Pa_{53}^{+}$$

$$a_{35}^{-} = \tilde{P}a_{35}^{+}$$

$$a_{54}^{-} = \tilde{P}a_{54}^{+}$$

$$\tilde{P} = \operatorname{diag}\left(\exp\left(-i\mu^{+}h_{k}\right)\right)$$

$$\tilde{P} = \operatorname{diag}\left(\exp\left(-i\mu^{+}h_{k}\right)\right)$$

$$\tilde{P} = \operatorname{diag}\left(\exp\left(-i\mu^{+}h_{k}\right)\right)$$

$$(3.61)$$

$$a_{45}^{-} = Pa_{45}^{+}$$

The matrix P is used for the relations between pairs of amplitudes a_{53}^- , a_{53}^+ and pairs a_{45}^- , a_{45}^+ . This is because the stiffener plates are similar in material and geometrical parameters. The coordinate systems of the semi-infinite stiffener plates used to compute scattering matrices at the junction point 5 and the free end 4 are equivalent. Accordingly, the matrix \tilde{P} relate amplitudes of outgoing waves a_{35}^+ and a_{45}^+ with amplitudes of incoming waves a_{35}^- and a_{45}^- , respectively. By combining (3.61) with the second and third matrix equations in (3.60), one obtains

$$a_{35}^{-} = P \tilde{\rho}_{33} P a_{53}^{+} a_{45}^{-} = P \tilde{\rho}_{44} \tilde{P} a_{54}^{+}$$
(3.62)

Substituting these expressions to the third and fourth rows of the first equation in (3.60) yields the matrix equation for $\begin{bmatrix} a_{35}^- & a_{45}^- \end{bmatrix}^T$ as

$$\begin{bmatrix} I - \tilde{P}\tilde{\rho}_{33}P \rho_{33} & -\tilde{P}\tilde{\rho}_{33}P \tau_{43} \\ -P\tilde{\rho}_{44}\tilde{P} \tau_{34} & I - P\tilde{\rho}_{44}\tilde{P} \rho_{44} \end{bmatrix} \begin{bmatrix} a_{35}^- \\ a_{45}^- \end{bmatrix} = \begin{bmatrix} \tilde{P}\tilde{\rho}_{33}P \tau_{13} & \tilde{P}\tilde{\rho}_{33}P \tau_{23} \\ P\tilde{\rho}_{44}\tilde{P} \tau_{14} & P\tilde{\rho}_{44}\tilde{P} \tau_{24} \end{bmatrix} \begin{bmatrix} a_{15}^- \\ a_{25}^- \end{bmatrix} .$$
(3.63)

Finally, we can insert the solution of Equation (3.63) into (3.60) thus computing the effective scattering matrix s_{eff} as follows

$$s_{\text{eff}} = \begin{bmatrix} \rho_{11} \ \tau_{21} \\ \tau_{12} \ \rho_{22} \end{bmatrix} + \begin{bmatrix} \tau_{31} \ \tau_{41} \\ \tau_{32} \ \tau_{42} \end{bmatrix} \begin{bmatrix} I - \tilde{P}\tilde{\rho}_{33}P \ \rho_{33} \ -\tilde{P}\tilde{\rho}_{33}P \ \tau_{43} \\ -P\tilde{\rho}_{44}\tilde{P} \ \tau_{34} \ I - P\tilde{\rho}_{44}\tilde{P} \ \rho_{44} \end{bmatrix}^{-1} \begin{bmatrix} \tilde{P}\tilde{\rho}_{33}P \ \tau_{13} \ \tilde{P}\tilde{\rho}_{33}P \ \tau_{23} \\ P\tilde{\rho}_{44}\tilde{P} \ \tau_{14} \ P\tilde{\rho}_{44}\tilde{P} \ \tau_{24} \end{bmatrix}$$
(3.64)

The resonance conditions for which a perfect transmission or reflection of incoming waves can occur are defined by the wave number and angular frequency values k_x and ω at which local minima of

$$\left| \det \left(\begin{bmatrix} I - \tilde{P} \tilde{\rho}_{33} P \rho_{33} & -\tilde{P} \tilde{\rho}_{33} P \tau_{43} \\ -P \tilde{\rho}_{44} \tilde{P} \tau_{34} & I - P \tilde{\rho}_{44} \tilde{P} \rho_{44} \end{bmatrix} \right) \right|$$
(3.65)

are obtained.

3.6.3 A composite plate with multiple stiffeners

Now, having computed the scattering coefficients for a plate with one or a pair of stiffener plates attached to it, we can formulate the scattering matrix for a plate with two stiffener plates and, eventually, with m stiffeners using recursion. First, we consider a plate with two stiffeners k and k + 1 replaced by the correspondent effective matrices s_{eff}^k and s_{eff}^{k+1} as shown in Figure 3.16. The distance between points 2 and 3 is l_k . Following the same approach as in the previous subsection, we write the two effective scattering matrices as

$$\begin{bmatrix} a_{21}^+ \\ a_{23}^+ \end{bmatrix} = \begin{bmatrix} \rho_{11} & \tau_{31} \\ \tau_{13} & \rho_{33} \end{bmatrix} \begin{bmatrix} a_{12}^- \\ a_{32}^- \end{bmatrix} , \begin{bmatrix} a_{32}^+ \\ a_{34}^+ \end{bmatrix} = \begin{bmatrix} \rho_{22} & \tau_{42} \\ \tau_{24} & \rho_{44} \end{bmatrix} \begin{bmatrix} a_{-3}^- \\ a_{-43}^- \end{bmatrix} .$$
(3.66)

The amplitudes $a_{23(32)}^-$ are related to $a_{23(32)}^+$ in the similar way as in (3.49) and (3.50),



Figure 3.16: A schematic view of a stiffened plate with two stiffeners represented by effective scattering matrices s_{eff}^k and s_{eff}^{k+1} in the yz plane. Here, a_{ij}^- represent the amplitudes of incoming waves, and a_{ij}^+ , the amplitudes of outgoing waves.

that is,

$$\begin{cases} a_{23}^{-} = Pa_{23}^{+} , \quad P = diag(e^{-\mu^{+}l_{k}}) \\ a_{32}^{-} = \tilde{P}a_{32}^{+} , \quad \tilde{P} = diag(e^{-\tilde{\mu}^{+}l_{k}}) \end{cases}$$
(3.67)

As highlighted in subsection 3.6.1, P is not equal to \tilde{P} in general. Combining the second row of matrix equations (3.66) with (3.67), we find

$$a_{23}^{-} = P\tau_{13} a_{12}^{-} + P\rho_{33} a_{32}^{-} a_{32}^{-} = \tilde{P}\rho_{22} a_{23}^{-} + \tilde{P}\tau_{42} a_{43}^{-}$$
(3.68)

Now, solving (3.68) in terms of a_{12}^- and a_{43}^- yields the following:

$$a_{23}^{-} = \left(I - P\rho_{33}\tilde{P}\rho_{22}\right)^{-1} \left(P\tau_{13}a_{12}^{-} + P\rho_{33}\tilde{P}\tau_{42}a_{43}^{-}\right) a_{32}^{-} = \left(I - \tilde{P}\rho_{22}P\rho_{33}\right)^{-1} \left(\tilde{P}\rho_{22}P\tau_{13}a_{12}^{-} + \tilde{P}\tau_{42}a_{43}^{-}\right).$$
(3.69)

Finally, we substitute these equations into equations (3.66) and obtain the effective scattering matrix for a plate with two stiffeners attached as

$$\begin{bmatrix} a_{21}^+ \\ a_{34}^+ \end{bmatrix} = s_{\text{eff}}^* \begin{bmatrix} a_{12}^- \\ a_{43}^- \end{bmatrix} = \begin{bmatrix} \rho_{11}^* & \tau_{41}^* \\ \tau_{14}^* & \rho_{44}^* \end{bmatrix} \begin{bmatrix} a_{12}^- \\ a_{43}^- \end{bmatrix}$$
(3.70)

with

$$\rho_{11}^{*} = \rho_{11} + \tau_{31} \left(\mathbf{I} - \tilde{\mathbf{P}} \rho_{22} \mathbf{P} \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \rho_{22} \mathbf{P} \tau_{13}$$

$$\tau_{41}^{*} = \tau_{31} \left(\mathbf{I} - \tilde{\mathbf{P}} \rho_{22} \mathbf{P} \rho_{33} \right)^{-1} \tilde{\mathbf{P}} \tau_{42}$$

$$\tau_{14}^{*} = \tau_{24} \left(\mathbf{I} - \mathbf{P} \rho_{33} \tilde{\mathbf{P}} \rho_{22} \right)^{-1} \mathbf{P} \tau_{13}$$

$$\rho_{44}^{*} = \rho_{44} + \tau_{24} \left(\mathbf{I} - \mathbf{P} \rho_{33} \tilde{\mathbf{P}} \rho_{22} \right)^{-1} \mathbf{P} \rho_{33} \tilde{\mathbf{P}} \tau_{42}$$
(3.71)

We note that the values of wave numbers k_x and frequencies ω at which the resonances occur can be found by calculating local minima of

$$\left|\det\left(\mathbf{I}-\tilde{\mathbf{P}}\rho_{22}\mathbf{P}\rho_{33}\right)\right|$$
 and $\left|\det\left(\mathbf{I}-\mathbf{P}\rho_{33}\tilde{\mathbf{P}}\rho_{22}\right)\right|$. (3.72)

Finally, we can compute the total effective scattering matrix S_{eff} in Equation (3.42) via the following algorithm:

1. Compute local effective scattering matrices $s_{\text{eff}}^k, k = 1, \dots, m$

of T-type junctions using Equations (3.57) and (3.64).

- 2. Compute s_{eff}^* that replaces the local effective scattering matrices s_{eff}^1 and s_{eff}^2 using Equation (3.71).
- 3. Repeat step 2 for effective scattering matrices s_{eff}^* and s_{eff}^k , $k = 3, \ldots, m$.
- 4. Compute effective energy scattering coefficients using Equation (3.43).

(3.73)

3.7 Computational results

In this section, several numerical applications of the method discussed earlier are presented. We first consider the reflection of various incoming waves' power at the free edge of a composite plate, then we turn to the reflection and transmission coefficients of composite plates joined together in the form of L and T junctions. Finally, the effective energy scattering coefficients of a stiffened composite plate with one and multiple stiffeners are shown in the last example. Only the scattering coefficients for the propagating waves are shown. The attenuating contributions are, of course, considered in full in the actual computations.

3.7.1 Reflection at the free edge of a composite plate

We consider a semi-infinite regular symmetric laminated plate with the free edge along the x axis. The plate has five layers, and two lamination schemes are considered. Namely, the first one is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$, that is, a cross-ply lami-



Figure 3.17: Wave vector (left) and dispersion (right) curves for a cross-ply composite plate. The wave vector curves are plotted for a fixed frequency f = 3000 Hz, whereas the dispersion curves are plotted for a fixed wave number component $k_x = 5m^{-1}$. Squares and circles denote wave numbers of incoming and outgoing waves, respectively. The bending wave and dispersion curves are scaled with a factor of 1/3 for the sake of clarity.



Figure 3.18: Energy reflection coefficients of a regular cross-ply symmetric laminate at the free edge for various incident waves as a function of wave number component k_x at a frequency 3000 Hz. The lamination scheme is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

nated plate, and the second one is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$, that is, an angle-ply laminated plate. The material of the orthotropic lamina used in the laminate is given in Table 3.1. The total thickness of the laminate is 5 mm. The energy reflection coefficients can be computed using Equations (3.38) and (3.40). Figure 3.17 presents wave vector and dispersion curves of the cross-ply laminated plate for a fixed frequency f = 3000 Hz on the left side and a fixed wave number component $k_x = 5 \text{ m}^{-1}$ on the right side, respectively. Since the plate consists of laminas with ply direction angles of 0° and 90°, the extensional coefficients A_{16} , A_{26} and bending coefficients D_{16} , D_{26} are equal to zero (see Section A.1 for details). Therefore, according to Equations (3.4) and (3.5), the incoming and outgoing wave number components are equal in magnitude, i.e.



Figure 3.19: Energy reflection coefficients of a regular cross-ply symmetric laminate at the free edge for various incident waves as a function of frequency at a wave number component $k_x = 5m^{-1}$. The lamination scheme is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

 $k_y^+ = -k_y^-.$

We note that the longitudinal waves can propagate for $|k_x| \leq 2.6 \text{ m}^{-1}$ at a fixed frequency f = 3000 Hz and for $f \geq 5701 \text{ Hz}$ at a fixed wave number component $k_x = 5 \text{ m}^{-1}$. Regarding the shear waves, they can propagate for $|k_x| \leq 10.6 \text{ m}^{-1}$ at a fixed frequency f = 3000 Hz and for $f \geq 1416 \text{ Hz}$ at a fixed wave number component $k_x = 5 \text{ m}^{-1}$.

Figure 3.18 presents the energy reflection coefficients of the cross-ply laminated plate for incoming shear and longitudinal waves as a function of the wave number component k_x at a fixed frequency f = 3000 Hz. An incoming longitudinal wave is fully reflected without mode conversion for $|k_x| \leq 1$ m⁻¹, see Figure 3.18b. The corresponding energy reflection coefficient has the form t_{LL}^{11} . For 1 m⁻¹ < $|k_x| \leq 2.6$ m⁻¹, mode conversion to the reflected shear wave takes place with the energy reflection coefficient t_{LS}^{11} . Vice-versa, for the same range of k_x , an incoming shear wave is reflected with mode conversion to the longitudinal wave in Figure 3.18a - the corresponding energy reflection coefficients is labelled as t_{SL}^{11} . Pure reflection of shear waves occurs for $|k_x| \leq 1$ m⁻¹ and 2.6 m⁻¹ < $|k_x| \leq 10.6$ m⁻¹ with the energy reflection coefficient t_{SS}^{11} .

Figure 3.19 presents the energy reflection coefficients of the cross-ply laminated plate for incoming shear and longitudinal waves as a function of frequency at a fixed wave number component $k_x = 5 \text{ m}^{-1}$. An incoming shear wave is reflected without mode conversion at frequencies where no energy can be reflected by longitudinal waves, that is, at frequencies 1416 Hz $\leq f < 5701$ Hz,



Figure 3.20: Energy reflection coefficients of a regular angle-ply symmetric laminate at the free edge for various incident waves as a function of wave number component k_x at a frequency 3000 Hz. The lamination scheme is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$.

see Figure 3.19a. At frequencies $f \ge 5701$ Hz, the energy reflection coefficients of incoming shear and longitudinal waves share similar shapes, that is, $t_{SL}^{11} = t_{LS}^{11}$ and $t_{SS}^{11} = t_{LL}^{11}$.

Figure 3.20 presents the energy reflection coefficients of the angle-ply laminated plate for incoming shear and longitudinal waves at a fixed frequency f = 3000 Hz. Pure reflection of longitudinal and shear waves without mode conversion occurs only in the vicinity of $k_x = 0$, see Figure 3.20a and Figure 3.20b. The phenomenon of mode conversion is observed again only between specific critical values: for an incoming shear mode, the reflected longitudinal wave is present for the k_x values between $\pm 3.7 \text{ m}^{-1}$. Furthermore, for $4.3 \text{ m}^{-1} \leq |k_x| \leq 7.6 \text{ m}^{-1}$, mode conversion to the second shear wave (labelled S_2) can occur. We encountered two pairs of incoming/outgoing shear wave numbers whilst considering shear wave vector curves in Figure 3.8. Vice-versa, the incoming second shear wave can couple to both types of shear waves, see



Figure 3.21: Energy reflection coefficients of a regular angle-ply symmetric laminate at the free edge for various incident waves as a function of frequency at a wave number component $k_x = 5 m^{-1}$. The lamination scheme is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$

Figure 3.20c.

If the wave number component k_x is fixed and the frequency is varied, one can find critical frequencies at which mode conversion phenomena start to occur. For example, in the case of an incident shear wave with $k_x = 5 \text{ m}^{-1}$, almost all energy is carried back by a longitudinal wave for $f \ge 4000 \text{ Hz}$, see Figure 3.21. Accordingly, in the same range of frequencies, an incoming longitudinal wave is almost fully reflected as a shear wave in Figure 3.21b. Mode conversion from incident shear wave to the second shear wave S_2 occurs in the frequency range $f \in [1941, 3609]$ Hz, and the energy reflection coefficients $t_{SS_2}^{11}$, t_{SS}^{11} and $t_{S_2S_2}^{11}$

The bending wave power is fully reflected at the free edge without mode conversion at all frequencies and wave number components k_x , that is, the energy



Figure 3.22: Energy scattering coefficients of an L-joint of regular cross-ply symmetric laminates with the lamination scheme $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$ for various incident waves as a function of wave number component k_x at a frequency 3000 Hz.

reflection coefficient $t_{BB}^{11} = 1$. Therefore, it is not presented in this subsection. Note that the energy reflection coefficients sum up to one according to Equation (3.41) which is used here as a check of consistency of the results (dashed blue lines in Figures 3.18, 3.19, 3.20, 3.21).

3.7.2 Two composite plates joined at a right angle

In this example, we consider two joined identical regular angle- and cross-ply laminated plates with the correspondent lamination schemes $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ and $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$, respectively.

We fix the inter-plate angle to $\psi = 90^{\circ}$, that is, the plate configuration has the form of an L-junction, similar to one in Figure 3.3. First, we consider a junction of two cross-ply laminated plates.

Figure 3.22 presents the energy scattering coefficients of the L-junction of



Figure 3.23: Energy scattering coefficients of an L-joint of regular cross-ply symmetric laminates with the lamination scheme $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$ for bending and longitudinal incident waves as a function of frequency for a wave number component $k_x = 0 \text{ m}^{-1}$.



Figure 3.24: Energy scattering coefficients of an L-joint of regular cross-ply symmetric laminates with the lamination scheme $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$ for various incident waves as a function of frequency for a wave number component $k_x = 5 \text{ m}^{-1}$.

cross-ply laminated plates for incident bending, shear and longitudinal modes as a function of the wave number component k_x at a frequency f = 3000 Hz. Since the wave number solutions of Equations (3.4) and (3.5) $\mu_{L,S,B}^- = \mu_{L,S,B}^+$, all energy scattering coefficients are symmetrical around $k_x = 0$ m⁻¹. Longitudinal and shear modes are propagating for the same range of wave number



Figure 3.25: Energy scattering coefficients of an L-joint of regular angle-ply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of wave number component k_x at a frequency 3000 Hz.

components k_x as in Figure 3.18, as expected. Consequently, outgoing longitudinal and shear waves can only occur in the narrow vicinity of normal incidence for an incoming bending wave. The phase angles of outgoing propagating longitudinal waves are limited by ~ 3° and ~ 11.5° with respect to the y axis. Furthermore, the energy scattering coefficients $t_{BL}^{11(12)}$ and $t_{BS}^{11(12)}$ are bounded by a value of 0.2 almost everywhere in the correspondent propagating ranges of k_x . On the other hand, the bending energy reflected and transmitted coefficients are equal to each other in these ranges of k_x . As can be seen in Figure 3.22a, curves corresponding to the longitudinal mode are smoothly transformed to ones corresponding to the shear mode, thus indicating the similar nature of coupling to both in-plane modes. In Figure 3.22b, an incoming shear wave energy is almost fully transmitted in the range $|k_x| \leq 2.6 \text{ m}^{-1}$, and the curve is similar to one corresponding to the shear wave reflection at



Figure 3.26: Energy scattering coefficients of an L-joint of regular angle-ply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency for a wave number component $k_x = 0 \text{ m}^{-1}$.

the free edge in Figure 3.18a. Incoming longitudinal wave energy is mainly reflected at the shared edge between plates, see Figure 3.22c.

If we consider only normal incidence of incoming waves and vary frequency, we can note that coupling can only occur between longitudinal and bending waves, see Figures 3.23a and 3.23b. Furthermore, the shear wave energy is fully transmitted at all frequencies, that is, $t_{SS}^{12} = 1$, and hence, it is not shown in Figure 3.23. At $k_x = 5 \text{ m}^{-1}$, coupling to all modes is present and occurred at the correspondent critical frequencies. For instance, at frequencies $f \ge 1416$ Hz, an incoming bending wave power P is equally distributed between reflected and transmitted bending waves with 0.4P and is converted to shear waves and longitudinal waves (at $f \ge 5701$ Hz) both with $\sim 0.2P$. Figure 3.25 shows the energy scattering coefficients of the L-junction of angleply laminated plates for incident bending, shear or longitudinal modes as a



Figure 3.27: Energy scattering coefficients of an L-joint of regular angle-ply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency for a wave number component $k_x = 5 \text{ m}^{-1}$.

function of the wave number component k_x at 3000 Hz. Note that reflection and transmission coefficients of bending, shear and longitudinal waves without mode conversion (having the form t_{XX}^{ij} with X = B, L, S and i, j = 1, 2) are symmetric around $k_x = 0$ whereas coefficients describing scattering between different modes, that is, $t_{XX'}^{ij}$ with $X \neq X'$ are non-symmetric. Again, the mode conversion can only occur between certain critical values: for an incoming bending mode, these are the k_x values between $\pm 7.8 \,\mathrm{m}^{-1}$ for reflected or transmitted shear waves, $\pm 3.7 \,\mathrm{m}^{-1}$ for longitudinal waves and $4.3 \,\mathrm{m}^{-1} \leq$ $|k_x| \leq 7.6 \,\mathrm{m}^{-1}$ for second shear waves. For example, an incoming bending wave power P with $k_x \sim -7.2 \,\mathrm{m}^{-1}$ is reflected and transmitted, both with 0.35P, converted to the shear transmitted wave with 0.2P and the second shear transmitted wave with 0.1P, see Figure 3.25a.

If the frequency is varied with a fixed wave number component k_x , one can



Figure 3.28: Schematic representation of a T-junction connecting three orthotropic plates. The angles ψ_2 and ψ_3 here are set to 180° and 90°, respectively. Red, blue and green lines represent the ply direction of the plates. The local angles of orientations ϕ_1 and ϕ_3 are both set to 45°, whereas ϕ_2 is equal to -45° .

note several things. For a normally incident waves, that is, at a fixed wave number component $k_x = 0 \text{ m}^{-1}$, all modes are propagating for all frequencies, see Figure 3.26. An incoming bending wave energy flux is equally divided between reflected and transmitted bending wave energy fluxes for frequencies 100 Hz $\leq f \leq 6000$ Hz. Mode conversion to the longitudinal wave becomes more significant as frequency increases, although all coupled energy scattering coefficients having the form t_{XY}^{ij} , with $X \neq Y$ and i, j = 1, 2 do not exceed 0.2. In Figure 3.26b, the incoming shear wave energy is mostly reflected with a constant value of 0.6 over the range of frequencies considered, whereas the energy scattering coefficient t_{LL}^{11} decreases with increasing frequency, see Figure 3.26c. If the wave number component $k_x = 5 \text{ m}^{-1}$, one can find critical frequencies at which mode conversion phenomena start to occur, see Figure 3.27. As in subsection 3.7.1, outgoing shear waves become propagating at $f \ge 1941$ Hz, whereas outgoing longitudinal waves become propagating for $f \ge 4000$ Hz, see Figure 3.27. Mode conversion from incident bending and shear waves to the second shear wave S_2 occurs in the frequency range $f \in [1941, 3609]$ Hz.



Figure 3.29: Energy scattering coefficients of an T-joint of regular angle-ply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of wave number component k_x at a frequency 3000 Hz.

3.7.3 Three composite plates joined in a T-junction

In this example, we consider three composite plates joined in the form of a T-junction with angles $\psi_1 = 0^\circ$, $\psi_2 = 180^\circ$ and $\psi_3 = 90^\circ$. The composites considered are the same as in the case of the L-junction. Figure 3.28 shows an illustrative example of such a configuration of composite plates with only one ply per thickness. In Figure 3.28, the local ply direction angles are $\phi_1 = \phi_3 = 45^\circ$ and $\phi_2 = -45^\circ$, and ϕ_2 is given a value of -45° to ensure that plates 1 and 2 form a uniform ground plate. In the similar manner, the local ply direction angles of laminated plates considered are $\phi_1^1/\ldots/\phi_1^5 = \phi_3^1/\ldots/\phi_3^5 = 45^\circ/-45^\circ/45^\circ/-45^\circ/45^\circ$, and $\phi_2^1/\ldots/\phi_2^5 = -45^\circ/45^\circ/-45^\circ/45^\circ/-45^\circ$. The case of cross-ply and laminated plates is omitted here for the sake of


Figure 3.30: Energy scattering coefficients of an T-joint of regular angle-ply symmetric laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ for various incident waves as a function of frequency at a wave number component $k_x = 5 m^{-1}$.

brevity. However, the energy scattering coefficients for junctions of cross-ply and general-ply plates are computed for comparison with the hybrid FE/WFE method based results in Chapter 4.

Figure 3.29 presents the energy scattering coefficients of the T-junction for incoming bending, shear and longitudinal waves as a function of wave number component k_x at a frequency 3000 Hz. We note that for varying wave number component k_x , most of the incoming shear and longitudinal incoming power is transmitted to the second plate with $\psi_2 = 180^\circ$, which can be explained by the fact that the first and second plates constitute a uniform ground plate, see Figure 3.29b and Figure 3.29c. However, the incoming bending wave power is mostly reflected back at the junction with minor deviations at critical wave number components $|k_x| = 3.7 \text{ m}^{-1}$, 4.3 m^{-1} , 7.6 m^{-1} , 7.8 m^{-1} , corresponding to longitudinal, second shear and shear modes, see Figure 3.29a. We note that as opposed to the case of the L-junction, symmetry of wave energy reflection and transmission coefficients without mode conversion is no longer present.

In Figure 3.30, the dependence of the energy scattering coefficients on the frequency at a fixed wave number component $k_x = 5 \text{ m}^{-1}$ is regarded. As in the fixed frequency case, most of the incoming bending wave energy is reflected at the shared edge except for frequencies 1941 Hz and 3609 Hz at which coupling to shear waves occurs. Incoming shear and longitudinal wave energies are mainly transmitted to the ground plate, see Figures 3.30b and 3.30c. Moreover, mode coupling occurs mainly between in-plane modes with energy scattering coefficients having the form t_{XY}^{ij} with $X = L, S, S_2$ and i, j = 1, 2, 3. Finally, the blue dash lines in Figures 3.29 and 3.30 represent sums of energy scattering coefficients for different incoming modes. As expected in the undamped systems, the energy scattering coefficients sum up to unity.

3.7.4 Stiffened composite plates

In the final example, we consider a composite plate with several composite plates of finite length mounted onto and/or under it at 90°, see Figures 3.11 and 3.15. All material properties of the plates are the same as in the case of the L-junction.

In the case of stiffeners put only on one side of the ground plate, the intersections (or common edge) between each stiffener and a ground plate have thus the form of a T-junction with angles $\psi_1 = 0^\circ$, $\psi_2 = 180^\circ$ and $\psi_3 = 90^\circ$ similar to the configuration shown in Figure 3.28. In the case of angle-ply laminated plates, the local ply direction angles are $\phi_1^1/\ldots/\phi_1^5 = \phi_3^1/\ldots/\phi_3^5 =$ $45^\circ/-45^\circ/45^\circ/-45^\circ/45^\circ$, and $\phi_2^1/\ldots/\phi_2^5 = -45^\circ/45^\circ/-45^\circ/45^\circ/-45^\circ$ - this is to ensure that plates 1 and 2 form a uniform ground plate.

In the case of stiffeners put on both sides of the ground plate, the intersections form a "+" shape with angles $\psi_1 = 0^\circ$, $\psi_2 = 180^\circ$, $\psi_3 = 90^\circ$ and $\psi_4 = 270^\circ$ as in Figure 3.15.

In this case, the local ply direction angles are $\phi_1^1/\ldots/\phi_1^5 = \phi_3^1/\ldots/\phi_3^5 = 45^\circ/-$

 $45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$, and $\phi_2^1/\ldots/\phi_2^5 = \phi_4^1/\ldots/\phi_4^5 = -45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}$ thus ensuring that plates 1 and 2 form a uniform ground plate and that plates 3 and 4 - a uniform stiffener plate. We will also consider the cross-ply laminated plates; in this case, all local ply direction angles are $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$, since changing signs of ply direction angles from 90° to -90° do not change the wave number solutions and correspondent wave shape modes.

We are interested, how the energy flux generated by an incident wave in the ground plate is partitioned between reflected and transmitted outgoing energy fluxes again in the ground plate, see Equation (3.43). To achieve this, one can follow the algorithm proposed in (3.73). The scattering coefficients computed in subsections 3.7.1 and 3.7.3 are used to produce effective scattering coefficients in the stiffened plate. In fact, $S^{(3)}$ and $S^{(4)}$ in Equation (3.46) denote scattering coefficients of a free edge and of a T-junction of composite plates.

First, we consider a composite plate with one stiffener plate mounted onto it. Figure 3.31 shows the energy scattering coefficients and the absolute value of the determinant in Equation (3.58) as a function of the wave number component k_x for a stiffener of length l = 30 cm with an incident bending wave at frequency 3000 Hz. The longest bending wavelength at this frequency is equal to 15.7 cm; thus, the short-wavelength approximation discussed in Section 3.6 is validated. Critical wave number components k_x for longitudinal and shear waves remain the same as discussed before, and only bending waves can propagate for $|k_x| > 7.8 \,\mathrm{m}^{-1}$. Note that the effective bending energy scattering coefficients are symmetric around $k_x = 0$, whilst in-plane mode related coefficients are slightly asymmetric in $|k_x| \leq 7.8 \text{ m}^{-1}$. Furthermore, one can observe a resonant behaviour of the reflected and transmitted energy flux coefficients t_{BB}^{11} and t_{BB}^{12} at specific k_x values, which related to the resonance condition (3.58). At most of the values of k_x , which satisfy this condition, see Figure 3.31b, an incoming wave is either totally transmitted or reflected, see Figure 3.31a. Critical wave number component values $|k_x| = 3.7$ yield local minima of $|\det \left(I - \tilde{P} \tilde{\rho}_{33} P \rho_{33}\right)|$, however, the effective bending reflection



(b) The absolute value of the determinant as in Equation (3.58) and resonances.

Figure 3.31: Energy scattering coefficients and resonance conditions of a stiffened plate for an incident bending wave as a function of wave number component k_x at a frequency 3000 Hz. The length of the stiffener plate is 30 cm. The lamination scheme of plates is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$.

and transmission coefficients do not exhibit resonant behaviour. A similar phenomenon can be observed for the case of fixed wave number component k_x and varying frequency. Figure 3.32 presents effective bending energy scattering coefficients and resonances of Equation (3.58) for the frequency range 1000 Hz $\leq f \leq 6000$ Hz at $k_x = 0$. A short-wavelength approximation still holds for this frequency range, since the maximum bending wavelength at the lowest frequency f = 1000 Hz is ~ 27 cm < 30 cm - the length of the stiffener plate. Mode conversion to shear and longitudinal waves is present only in the vicinity of the resonant frequencies, becoming more significant as frequency increases. Bending reflection and transmission coefficients behave similarly as



(b) The absolute value of the determinant as in Equation (3.58) and resonances.

Figure 3.32: Energy scattering coefficients and resonance conditions of a stiffened plate for an incident bending wave as a function of frequency at a wave number component $k_x = 0 \ m^{-1}$. The length of the stiffener plate is 30 cm. The lamination scheme of plates is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$.

in the case of varying k_x , that is, waves are mainly reflected or transmitted except for small coupling to in-plane modes. When $k_x = 5 \text{ m}^{-1}$, total transmission or reflection of the bending energy occurs at the resonant frequency f = 1537 Hz. Starting at f = 1915 Hz, an incoming bending wave energy can be reflected and transmitted as both shear modes and longitudinal mode at $f \ge 4000$, see Figure 3.33.

Now, we consider the effective bending energy scattering coefficients of a symmetrically stiffened cross-ply plate as in Figure 3.16. The lamination scheme is the same as before - $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$. As mentioned in subsection 3.6.2, this configuration prohibits the generation of the outgoing in-plane modes when



(b) The absolute value of the determinant as in Equation (3.58) and resonances.

Figure 3.33: Energy scattering coefficients and resonance conditions of a stiffened plate for an incident bending wave as a function of frequency at a wave number component $k_x = 5 m^{-1}$. The length of the stiffener plate is 30 cm. The lamination scheme of plates is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$.

a bending wave is incident upon this structure. In Figure 3.34, the effective scattering coefficients are plotted with respect to the wave number component k_x at a frequency f = 3000 Hz. It is clear that there only occurs reflection or transmission of the bending wave without mode conversion. Since the plates considered are specially orthotropic, and scattering coefficients are symmetric around $k_x = 0$ m⁻¹, only the right side of the k_x axis is considered. We can note that most of the incoming bending energy is reflected, whilst at the resonant values of k_x , perfect transmission occurs. In Figure 3.35, the effective bending energy scattering coefficients are plotted for the range of frequencies 1000 Hz $\leq f \leq 6000$ Hz at fixed wave number components



(b) The absolute value of the determinant as in Equation (3.58) and resonances.

Figure 3.34: Energy scattering coefficients and resonance conditions of a symmetrically stiffened plate for an incident bending wave as a function of wave number component k_x at a frequency 3000 Hz. The lengths of the stiffener plates are 30 cm. The lamination scheme of plates is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

 $k_x = 0 \text{ m}^{-1}$ and $k_x = 5 \text{ m}^{-1}$. As opposed to the previous cases, figures for $k_x = 0 \text{ m}^{-1}$ and $k_x = 5 \text{ m}^{-1}$ are similar. Namely, the number of resonant frequencies is the same for both cases, and they are shifted along the frequency axis with increasing k_x . Furthermore, it appears that the scattering coefficients behave in the similar manner at frequencies f > 1416 Hz, a frequency at which shear waves start to propagate in non-stiffened plate cases, see Figures 3.19 and 3.24. The energy scattering coefficients sum up to one in the absence of damping, as expected.

In the final example, we consider a symmetrically stiffened plate with four pairs of stiffener plates attached to it. The lamination scheme of all plates is



Figure 3.35: Energy scattering coefficients of a symmetrically stiffened plate for an incident bending wave as a function of frequency at a wave number components $k_x = 0 \ m^{-1}$ and $k_x = 5 \ m^{-1}$. The lengths of the stiffener plates are 30 cm. The lamination scheme of plates is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

the same as in the previous example. The spacing between pairs of stiffeners is equal to the stiffener lengths 30 cm. In Figure 3.36, the energy scattering coefficients are shown in the form of a contour plot as a function of both the wave number component k_x and frequency f. The wave number component k_x varies between 0 m⁻¹ and 57 m⁻¹, whereas frequency is varied from 1000 Hz to 6000 Hz. Resonance conditions are satisfied along the frequency and k_x axes, as before in a stiffened plate with only one or a pair of stiffeners. However, since there are several pairs of stiffeners attached to the ground plate, instead of distinct resonant frequencies and values of k_x , we get so-called *pass* and *stop bands* of frequencies and wave number components k_x , at which total transmission or reflection of an incoming bending wave energy is observed. These bands are represented by blue colours in Figures 3.36a and 3.36b. Several things can



(a) Frequencies and wave numbers k_x at which total energy reflection occurs (coloured in blue). Partial energy transmission are presented in yellow colour.



(b) Frequencies and wave numbers k_x at which total energy transmission occurs (coloured in blue).

Figure 3.36: Contour plots of energy scattering coefficients of a symmetrically stiffened plate with four pairs of stiffeners for an incident bending wave as a function of frequency and wave number component k_x . The spacings between stiffeners and lengths of the stiffeners are 30 cm. The lamination scheme of plates is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

be noted from these figures. Firstly, bending waves can propagate within a bounded surface of possible values of frequency and wave number component k_x defined by Equations (3.4) and (3.5). Secondly, there is no single pass or stop bands at particular frequencies, which is the case in one-dimensional systems like stiffened beams or periodically supported beams. This feature was highlighted for isotropic stiffened plates by Tso and Hansen [196]. However, in that work, stiffeners are assumed to be periodically attached to the ground plate, and using periodic structure theory [113, 186], propagation factors are computed to derive pass and stop bands of the stiffened plate. The method presented in this chapter is suitable for any configuration of the stiffened plate



(a) Frequencies and phase angles $\theta = \arctan(k_x/k_y)$ at which total energy reflection occurs (coloured in blue). Partial energy transmission are presented in yellow colour.



(b) Frequencies and phase angles $\theta = \arctan(k_x/k_y)$ at which total energy transmission occurs (coloured in blue).

Figure 3.37: Contour plots of energy scattering coefficients of a symmetrically stiffened plate with four pairs of stiffeners for an incident bending wave as a function of frequency and phase angle $\theta = \arctan(k_y/k_x)$. The spacings between stiffeners and lengths of the stiffeners are 30 cm. The lamination scheme of plates is $0^{\circ}/90^{\circ}/0^{\circ}/0^{\circ}$.

with arbitrary material and geometrical parameters. Thirdly, perfect transmission zones presented in blue colour in Figure 3.36b are narrower and more discrete than the yellow zones in Figure 3.36a. This is because yellow zones represent not only total transmission of bending wave energy but also partial transmission, that is, the effective scattering coefficients $0 < t_{BB,eff}^{12} < 1$. Finally, we can express the effective energy coefficients with respect to the phase angle $\theta = \arctan(k_x/k_y)$, which is varied from 0° to 90° in the cross-ply laminated stiffened plate, see Figure 3.37. This allows to represent pass and stop bands over a fixed (f, θ) plane.

3.8 Conclusion

In this chapter, we have derived the semi-analytical method for computing scattering coefficients of structural junctions made up of composite laminated plates in the line junction approximation. Composite laminates have been modelled in the context of the CLP theory introduced in Section 2.2.

By analysing dispersion relations, we have observed and discussed interesting phenomena such as negative refraction, negative group velocity and evanescent waves decaying along a direction axis inclined to the coordinate axis. Furthermore, we have derived expressions quantifying transmission and reflection coefficients as a function of the frequency and the wave number component k_x . Also, we have computed an effective scattering matrix for a plate with multiple finite stiffeners attached to it. Conditions at which resonances are achieved have been discussed.

Finally, in the numerical case studies, we have computed the scattering coefficients for examples of two and three composite plates joined together in an L and T geometry and effective scattering coefficients for a plate with one and four stiffeners attached to it. The method discussed in this chapter will be used for comparison with a wave finite element approach of extracting reflection and transmission coefficients in Chapter 4.

Chapter 4

Wave and Finite Element Analysis of wave propagation in composite plates

4.1 Introduction

In this chapter, the wave and finite element method based approach for calculation of energy scattering coefficients for arbitrary junctions of composite plates is derived. The wave and finite element method is based on modelling only a periodic segment of a plate using an FE software and applying periodic structure theory to extract wave propagation characteristics such as wave numbers, wave mode shapes and group velocities. Once these are determined for each plate considered, the displacement continuity and force equilibrium conditions at the junction between plates are used to express amplitudes of outgoing waves in terms of amplitudes of incoming waves, thus producing the scattering matrix. The plates considered are modelled using two different finite element types, two-dimensional shell and three-dimensional solid elements. We consider the implication and influence of each element type on the scattering coefficients.

In contrast to the semi-analytical approach presented in Chapter 3, we only require that individual laminas of the plates are homogeneous. Numerical results on selected examples of junctions of composite plates are compared with semi-analytical results from Chapter 3.

This chapter is organised as follows. In Section 4.2, the WFE method for mod-

elling composite plates is briefly reviewed summarising key moments discussed in Section 2.3. An eigenvalue problem whose solutions yield wave numbers and mode shapes is set up. Then, the classification of wave numbers and the wave basis setting are given in Section 4.3. Having established a wave basis representation of displacement and force vectors in individual plates, we combine these solutions with the equations of motion of the joint via application of continuity of displacements and force equilibrium at the joint boundaries, thus producing scattering coefficients in Section 4.4. Finally, Section 4.5 presents numerical case studies for two and three coupled composite plates. The energy scattering coefficients of L junctions of regular cross-ply and angle-ply composite plates are computed and compared with semi-analytical results presented in Section 3.7.

4.2 Summary of the WFE method for composite plates

As presented in Section 2.3, a unit cell of a composite plate with plane dimensions d_x and d_y is modelled using two-dimensional shell elements with a composite lay-up or three-dimensional solid elements stacked up one on top of the other, representing different composite layers. Assuming that the structure undergoes a harmonic vibration with angular frequency ω and no external forces is applied, we can write the governing equation of motion of the unit cell as

$$\mathbf{Dq} = \mathbf{f}, \quad \mathbf{D} = \mathbf{K} (1 + i\eta) - \omega^2 \mathbf{M},$$
(4.1)

where **D** is the dynamic stiffness matrix with **M** and **K** being mass and stiffness matrices, respectively. η denotes a uniform structural damping coefficient. This equation relates the displacement and force vectors **q** and **f**. Now, we can use the periodic structure theory [186, 187] to relate displacement and force sub-vectors along the x axis of the plate as

$$\mathbf{q}_T = \lambda_x \, \mathbf{q}_B \,, \quad \mathbf{f}_T = -\lambda_x \, \mathbf{f}_B \,, \quad \lambda_x = e^{-\mathrm{i}k_x d_x} \tag{4.2}$$

and reduce Equation (4.1) to the form equivalent to Equation (2.64) as

$$\begin{bmatrix} \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR} \\ \tilde{\mathbf{D}}_{RL} & \tilde{\mathbf{D}}_{RR} \end{bmatrix} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{cases} = \begin{cases} \tilde{\mathbf{f}}_L \\ \tilde{\mathbf{f}}_R \end{cases}, \qquad (4.3)$$

where the matrix $\tilde{\mathbf{D}}$ is a dynamically-condensed version of the modified dynamic stiffness matrix $\mathbf{T}^*\mathbf{DT}$, see subsection 2.3.2. The transformation matrix \mathbf{T} is given by Equation (2.60). The vectors $\tilde{\mathbf{q}}_L$ and $\tilde{\mathbf{q}}_R$ represent the nodal displacement sub-vectors \mathbf{q}_{LB} and \mathbf{q}_{RB} in Figure 2.3, whereas the vectors $\tilde{\mathbf{f}}_L$ and $\tilde{\mathbf{f}}_R$ can be computed as

$$\left\{ \begin{aligned} \mathbf{\tilde{f}}_{L} \\ \mathbf{\tilde{f}}_{R} \end{aligned} \right\} = \mathbf{T}^{H} \mathbf{f} = \left\{ \begin{aligned} \mathbf{f}_{LB} + \sum_{j=1}^{n_{x}-1} \lambda_{x}^{-j/n_{x}} \mathbf{f}_{L}^{j} + \lambda_{x}^{-1} \mathbf{f}_{LT} \\ \mathbf{f}_{RB} + \sum_{j=1}^{n_{x}-1} \lambda_{x}^{-j/n_{x}} \mathbf{f}_{R}^{j} + \lambda_{x}^{-1} \mathbf{f}_{RT} \end{aligned} \right\}$$

$$(4.4)$$

with n_x denoting the number of mesh cells in the x direction of the periodic cell.

Finally, by applying the periodic structure theory and force equilibrium in the y direction, which can be written as

$$\tilde{\mathbf{q}}_R = \lambda_y \, \tilde{\mathbf{q}}_L \,, \quad \tilde{\mathbf{f}}_R = -\lambda_y \, \tilde{\mathbf{f}}_L \,, \quad \lambda_y = e^{-\mathrm{i}k_y d_y} \,, \tag{4.5}$$

one can get from Equation (4.3) the following eigenvalue problem for the propagation factor λ_y

$$\mathbf{S} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} = \lambda_y \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} \quad \text{with} \quad \mathbf{S} = \begin{bmatrix} -\tilde{\mathbf{D}}_{LR}^{-1} \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR}^{-1} \\ -\tilde{\mathbf{D}}_{RL} + \tilde{\mathbf{D}}_{RR} \tilde{\mathbf{D}}_{LR}^{-1} \tilde{\mathbf{D}}_{LL} & -\tilde{\mathbf{D}}_{RR} \tilde{\mathbf{D}}_{LR}^{-1} \end{bmatrix}.$$

$$(4.6)$$

The dimension of the matrices $\tilde{\mathbf{D}}$ and \mathbf{S} is 2m, where m is the dimension of the displacement and force vectors $\tilde{\mathbf{q}}_L$ and $\tilde{\mathbf{f}}_L$. Therefore, the solution of the eigenvalue problem (4.6) consists of 2m propagation factors $\lambda_{y,i}$ and the correspondent eigenvectors $\left\{\phi_{q,i} \ \phi_{f,i}\right\}^T$ provided that the circular frequency ω and the wave number component k_x are fixed. The wave number components $k_{y,i}$ can be computed as

$$k_{y,i} = \ln\left(\frac{\lambda_{y,i}}{-\mathrm{i}d_y}\right), \quad i = 1,\dots,2m.$$
 (4.7)

Consequently, by varying the wave number component k_x , one can extract wave vector curves (k_x, k_y) for a fixed value of ω . It is worth noting that obtained wave number components $k_{y,i}$ can be real, imaginary or complex, making the correspondent plane waves propagating, evanescent or attenuating in the y direction - more on that in subsection 4.3.1.

4.3 Calculation of wave numbers and group velocity

The solution of the eigenvalue problem (4.6) might be prone to ill-conditioning of the matrix S, since the formulation of this matrix involves calculation of $\tilde{\mathbf{D}}_{LR}^{-1}$ [127]. The equivalent form of the eigenvalue problem can be formulated by eliminating $\tilde{\mathbf{f}}_L$ and $\tilde{\mathbf{f}}_R$ in Equation (4.3) as suggested in [199]:

$$\mathbf{N} \left\{ \begin{aligned} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{aligned} \right\} = \lambda_y \mathbf{L} \left\{ \begin{aligned} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{aligned} \right\}, \quad \mathbf{N} = \begin{bmatrix} 0 & \mathbf{I} \\ -\tilde{\mathbf{D}}_{RL} & -\tilde{\mathbf{D}}_{RR} \end{bmatrix} , \quad \mathbf{L} = \begin{bmatrix} \mathbf{I} & 0 \\ \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR} \end{bmatrix},$$

$$(4.8)$$

with **I** being the *m*-by-*m* identity matrix. The matrices **N** and **L** consist of block parts of the reduced dynamic stiffness matrix $\tilde{\mathbf{D}}$ with no matrix inversion as in Equation (4.6), thus reducing numerical ill-conditioning of the method to some extent. However, there might be a difference of several orders of magnitude between **I** and $-\tilde{\mathbf{D}}_{RL(RR)}$ and $\tilde{\mathbf{D}}_{LL(LR)}$; therefore, the condition numbers of the matrices **N** and **L** can still be large, causing numerical errors in the evaluation of eigenvalues and eigenvectors.

We use the following form of the eigenvalue problem (4.8) proposed by Fan et

al. [157, 158]

$$\tilde{\mathbf{N}} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{cases} = \lambda_y \tilde{\mathbf{L}} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{cases}, \quad \tilde{\mathbf{N}} = \begin{bmatrix} 0 & \sigma \mathbf{I} \\ -\tilde{\mathbf{D}}_{RL} & -\tilde{\mathbf{D}}_{RR} \end{bmatrix}, \quad \tilde{\mathbf{L}} = \begin{bmatrix} \sigma \mathbf{I} & 0 \\ \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR} \end{bmatrix}, \quad (4.9)$$

where $\sigma = \frac{\|\tilde{\mathbf{D}}_{RR}\|_2}{m^2}$, $\| \|_2$ representing the largest singular value of a matrix. This formulation is an improved version of the eigenvalue problem (4.8), and the factor σ is introduced here to reduce the condition number of the matrices **N** and **L**. However, the formulations are still equivalent, and the eigenvalue solutions are the same. In fact, writing the characteristic equation of the eigenvalue problem (4.9) yields

$$0 = \det(\tilde{\mathbf{N}} - \lambda_y \tilde{\mathbf{L}}) = \sigma^m \det(\mathbf{N} - \lambda_y \mathbf{L}) = \det(\mathbf{N} - \lambda_y \mathbf{L}) = 0, \qquad (4.10)$$

where we have applied a property of the determinant which tells that if a row of a matrix is multiplied by a non-zero constant, the determinant of that matrix is multiplied by the same constant. Note that eigenvectors in Equation (4.9) consist of left and right nodal displacements sub-vectors. To compute the nodal force sub-vectors $\tilde{\mathbf{f}}_L$, one can apply the following transformation

$$\begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} = \begin{bmatrix} \mathbf{I} & 0 \\ \tilde{\mathbf{D}}_{LL} & \tilde{\mathbf{D}}_{LR} \end{bmatrix} \begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{q}}_R \end{cases} .$$
 (4.11)

4.3.1 Incoming, outgoing waves and wave basis

Eigensolutions of Equation (4.8) can be separated into m pairs of roots, $\lambda_{y,j}^{\pm}$; these correspond to negative or incoming waves (with superscripts "-") and positive or outgoing waves (with superscripts "+"). Furthermore, the waves can be categorised as propagating, evanescent or attenuating. The standard wave classification [80, 160, 200] consists of checking whether $|\lambda_{y,j}| \leq 1$ to identify outgoing or incoming waves, respectively. The transfer matrix **S** in Equation (4.6) is assumed to be symplectic, hence the eigenvalues of Equa-



Figure 4.1: Bending wave vector curve in a $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ composite plate at a frequency 3000 Hz. Blue dots represent wave numbers related to outgoing waves while red squares correspond to incoming waves. Angles $\alpha_{1(2)}$ and $\theta_{1(2)}$ represent group and wave vector or phase angles, respectively. The wave number component k_x^{\max} is equal to $53 \,\mathrm{m}^{-1}$.

tions (4.6) and (4.8) must appear in pairs $(\lambda_{y,j}, 1/\lambda_{y,j})$. It follows that $k_{y,j}^+$ must equal to $-k_{y,j}^-$, and this is true for isotropic and some special types of composite plates, e.g. cross-ply laminates, which consist of layers with ply direction angles 0° or 90°. However, in general, the transfer matrix **S** is not symplectic because it depends on the wave number component k_x via the transformation matrix **T**, and $k_{y,j}^+ \neq -k_{y,j}^-$. Figure 4.1 shows an example of a bending wave vector curve, where one can see the inequality between incoming and outgoing wave number components k_y represented by red squares and blue dots, respectively. This inequality was demonstrated for a similar wave vector curve by Taupin et al. [201] using the Semi-Analytical Finite Element method. Furthermore, one can note the difference between phase and group angles in composite plates shown with $\theta_{1,2}^{\pm}$ and $\alpha_{1,2}^{\pm}$, respectively. This property was highlighted and discussed previously using the semi-analytical approach in Section 3.3.

To correctly categorise incoming/outgoing propagating/attenuating waves, we

establish the following algorithm for the jth wave

if
$$\operatorname{Re}(k_{y,j}) > c \operatorname{Im}(k_{y,j}) \Rightarrow$$
 the wave j is propagating
if $\operatorname{Re}(\mathrm{i}\omega\phi_{q,j}^*\phi_{f,j}) > 0 \Rightarrow$ the wave j is outgoing
else \Rightarrow the wave j is incoming
(4.12)
else \Rightarrow the wave j is attenuating

if
$$\operatorname{Im}(k_{y,j}) < 0 \Rightarrow$$
 the wave j is outgoing

else \Rightarrow the wave j is incoming,

where $\operatorname{Re}(\mathrm{i}\omega\phi_{q,j}^*\phi_{f,j})$ is the energy flux of the *j*th wave in the positive *y* direction, and *c* is a real parameter defined empirically and used to separate propagating from attenuating waves; in this work, it is equal to 10.

Categorising correctly whether a wave is incoming/outgoing and propagating/evanescent/attenuating is not enough to further proceed with the calculation of scattering coefficients. In fact, for a range of frequencies and wave number components k_x , we obtain a set of unclassified branches of propagating, evanescent and attenuating waves. To identify the eigensolutions of Equation (4.9) corresponding to the same wave type, we apply the so-called MAC criterion [132, 202, 203]. For an eigenvector solution $\Phi_i = \left\{ \phi_{q,i}^{\pm} \ \phi_{f,i}^{\pm} \right\}^T$ defined at frequency ω for a fixed wave number component k_x , we find an eigenvector solution $\Phi_j = \left\{ \phi_{q,i}^{\pm} \ \phi_{f,i}^{\pm} \right\}^T$ at frequency $\omega + d\omega$ with sufficiently small $d\omega$ such that

$$M^{\pm}(\omega) = \frac{\left(\Phi_i^T(\omega)\Phi_j^*(\omega+d\omega)\right)\left(\Phi_j^T(\omega)\Phi_i^*(\omega+d\omega)\right)}{\left(\Phi_i^T(\omega)\Phi_i^*(\omega+d\omega)\right)\left(\Phi_j^T(\omega)\Phi_j^*(\omega+d\omega)\right)}$$
(4.13)

is maximised. The same criterion can be utilised to classify wave types for a range of wave number components k_x and a fixed frequency with corresponding step size dk_x being sufficiently small.

Once the appropriate pairs of wave number components $k_{y,j}^{\pm}$ and wave mode shapes $\left\{\phi_{q,j}^{\pm} \ \phi_{f,j}^{\pm}\right\}^{T}$ are determined, we can express nodal displacements and

forces in the basis of wave mode shapes as follows

$$\begin{cases} \tilde{\mathbf{q}}_L \\ \tilde{\mathbf{f}}_L \end{cases} = \sum_{j=1}^m \left(a_j^+ \begin{cases} \phi_{q,j}^+ \\ \phi_{f,j}^+ \end{cases} + a_j^- \begin{cases} \phi_{q,j}^- \\ \phi_{f,j}^- \end{cases} \right) = \begin{cases} \Phi_{\mathbf{q}}^+ \mathbf{a}^+ + \Phi_{\mathbf{q}}^- \mathbf{a}^- \\ \Phi_{\mathbf{f}}^+ \mathbf{a}^+ + \Phi_{\mathbf{f}}^- \mathbf{a}^- \end{cases} , \quad (4.14)$$

where a_j^+ and a_j^- are the amplitudes of the *j*th outgoing and incoming waves. Using the expression of nodal displacements and forces in terms of wave mode shapes, we are now able to compute the energy scattering matrix at the junction of several plates which will be done in the next section.

4.3.2 Group velocity estimation

Calculation of the group velocity vector is essential for determining directions of rays carrying the wave energy. It is also used to relate the energy density to the velocity field in Chapter 2. Furthermore, it is present in the calculation of spatial density integrals for composite structures in Chapter 5.

As noted in subsection 3.3.1, the group velocity vector \mathbf{c}_g is defined as

$$\mathbf{c}_{g} = \left\{ c_{g,x} \quad c_{g,y} \right\}^{T} = \left\{ \frac{\partial \omega}{\partial k_{x}} \quad \frac{\partial \omega}{\partial k_{y}} \right\}^{T} .$$
(4.15)

Since $k_x = k \cos \theta$ and $k_y = k \sin \theta$, where θ is the wave vector polar angle, the partial derivatives $\frac{\partial \omega}{\partial k}$ and $\frac{\partial \omega}{\partial \theta}$ can be expressed in the form

$$\frac{\partial\omega}{\partial k} = \frac{\partial\omega}{\partial k_x}\frac{\partial k_x}{\partial k} + \frac{\partial\omega}{\partial k_y}\frac{\partial k_y}{\partial k} = \frac{\partial\omega}{\partial k_x}\cos\theta + \frac{\partial\omega}{\partial k_y}\sin\theta$$

$$\frac{\partial\omega}{\partial \theta} = \frac{\partial\omega}{\partial k_x}\frac{\partial k_x}{\partial \theta} + \frac{\partial\omega}{\partial k_y}\frac{\partial k_y}{\partial \theta} = -\frac{\partial\omega}{\partial k_x}k\sin\theta + \frac{\partial\omega}{\partial k_y}k\cos\theta$$
(4.16)

It then follows that x and y components of the group velocity vector \mathbf{c}_g may be written as

$$\begin{cases} c_{g,x} \\ c_{g,y} \end{cases} = \begin{bmatrix} \cos\theta & -\sin\theta \\ \sin\theta & \cos\theta \end{bmatrix} \begin{cases} \frac{\partial\omega}{\partial k} \\ \frac{\partial\omega}{k\partial\theta} \end{cases} .$$
 (4.17)

Considering the wave vector curve in the form $\omega(k,\theta) = \omega_0$ with a fixed value ω_0 , we can express $\frac{\partial \omega}{\partial \theta}$ in terms of $\frac{\partial \omega}{\partial k}$ and $\frac{\partial k}{\partial \theta}$ using a chain rule as follows

$$0 = d\omega = \frac{\partial\omega}{\partial k}\frac{\partial k}{\partial \theta} + \frac{\partial\omega}{\partial \theta} \Rightarrow \frac{\partial\omega}{\partial \theta} = -\frac{\partial\omega}{\partial k}\frac{\partial k}{\partial \theta}.$$
 (4.18)

Now, introducing Equation (4.18) into Equation (4.17) yields the final expressions for the group velocity vector components

$$\begin{cases} c_{g,x} \\ c_{g,y} \end{cases} = \begin{bmatrix} \cos\theta & -\sin\theta \\ \sin\theta & \cos\theta \end{bmatrix} \begin{cases} \frac{\partial\omega}{\partial k} \\ -\frac{\partial\omega}{\partial k} \frac{\partial k}{\partial \theta} \end{cases} .$$
(4.19)

Now, $\frac{\partial \omega}{\partial k}$ and $\frac{\partial k}{\partial \theta}$ need be computed for a polar angle $\theta = \arctan(k_y/k_x)$, hence for a wave number component k_x , and there are several methods to accomplish that.

Firstly, one can approximate these partial derivatives by finite differences as

$$\frac{\partial \omega}{\partial k}(k_x)|_{\omega=\omega_1} \approx \frac{\Delta \omega}{k|_{\omega_1+\Delta\omega} - k|_{\omega_1}}, \qquad (4.20)$$

$$\frac{\partial k}{\partial \theta}(k_x)|_{\theta=\theta_1} \approx \frac{k|_{\theta_1+\Delta\theta} - k|_{\theta_1}}{\Delta \theta},$$

where $\Delta \omega$ and $\Delta \omega$ are sufficiently small increment values of frequency and polar angle. The second equation in (4.20) can be easily utilised in the context of the presented wave finite element method, since wave vector curves are computed at the fixed frequency - an example of a bending wave vector curve is shown in Figure 4.1. However, the first equation in (4.20) requires wave vector solutions at two very close frequencies, ω_1 and $\omega_1 + \Delta \omega$, which increases the computational cost of the approach.

Lastly, one can approximate these derivatives by performing sensitivity analysis on the eigenvalue problem (4.9) in a similar way as in [140] and [137]. Nonetheless, this is beyond the scope of this thesis; therefore, the finite difference method is used to approximate partial derivatives.

Finally, one can calculate the direction angle of the energy flow as

$$\alpha = \arctan\left(\frac{c_{g,y}}{c_{g,x}}\right) \,. \tag{4.21}$$

4.4 Computation of scattering coefficients at junctions of composite plates

In this section, the energy scattering coefficients of the junction of several composite plates are derived. There are three cases considered. Firstly, a wave energy reflection occurring at the free edge of the composite plate is described. Secondly, we compute the wave energy reflection and transmission at the shared edge between plates modelled with SHELL181 elements. Finally, a case of a solid joint connecting several plates modelled with SOLID185 elements is regarded. In all cases, boundary conditions at the junction, such as continuity of displacement vectors and force equilibrium, are postulated.

4.4.1 Reflection matrix for a single plate system

This subsection presents the case of wave energy scattering at the boundary of a composite plate. The boundary can be an edge placed along the x axis for shell elements and a face lying on the (x, z) plane for solid elements. Since we consider a single plate, an incoming wave is fully reflected at the boundary with no transmission, although coupling to different modes can occur. Any boundary condition of a plate can be represented in terms of nodal displacements and forces as follows [153]

$$\mathbf{C}_{q}\tilde{\mathbf{q}}_{L} + \mathbf{C}_{f}\tilde{\mathbf{f}}_{L} = 0. \qquad (4.22)$$

For instance, if free boundary conditions are considered, i.e. all tractions at the boundary are equal to zero, then $\mathbf{C}_q = 0$ and $\mathbf{C}_f = \mathbf{I}$. Furthermore, the same formulation with appropriate matrices \mathbf{C}_q and \mathbf{C}_f can describe the case of the fixed edge shared between plates modelled with SHELL181 elements, more on this in subsection 4.4.2. Introducing Equation (4.14) into Equation (4.22) yields

$$\mathbf{a}^+ = \mathbf{s} \, \mathbf{a}^-, \quad \text{with} \quad \mathbf{s} = -(\mathbf{C}_q \mathbf{\Phi}_q^+ + \mathbf{C}_f \mathbf{\Phi}_f^+)^{-1} \left(\mathbf{C}_q \mathbf{\Phi}_q^- + \mathbf{C}_f \mathbf{\Phi}_f^-\right).$$
 (4.23)

The expression for **s** defines a $m \times m$ scattering matrix relating the amplitudes \mathbf{a}^- and \mathbf{a}^+ of incoming and outgoing waves, respectively. Specifically, we write the scattering matrix elements in the form $s_{ij}^{nm}(\omega, k_x)$, which relates an incoming wave *i* in the plate *n* and a reflected or transmitted wave *j* in the plate *m* at angular frequency ω and wave number component k_x . To compute the energy scattering coefficients, one needs to compute the outgoing wave energy flux ratio over incoming wave energy flux. For the associated energy fluxes, we obtain the energy scattering coefficients as

$$t_{ij}^{nm}(\omega, k_x) = \begin{cases} \frac{J_{j,m}^+}{J_{i,n}^-} |s_{ij}^{nm}|^2 & \text{if wave } j \text{ is propagating.} \\ 0 & \text{otherwise.} \end{cases}, \qquad (4.24)$$

where

$$\begin{cases} J_{i,n}^{-} = \left| \operatorname{Re} \left(i \omega \phi_{q,i,n}^{-*} \phi_{f,i,n}^{-} \right) \right|. \\ J_{j,m}^{+} = \left| \operatorname{Re} \left(i \omega \phi_{q,j,m}^{+*} \phi_{f,j,m}^{+} \right) \right|. \end{cases}$$

$$(4.25)$$

In the absence of damping, total energy must be conserved, hence the sum of energy scattering coefficients over the outgoing modes equals one, that is,

$$\sum_{m=1}^{N} \sum_{j} t_{ij}^{nm} = 1.$$
(4.26)

Note that since in this subsection we consider only one plate, the energy scattering coefficients have the form $t_{ij}^{11}(\omega, k_x)$, that is, only wave energy reflection can occur with or without coupling to other wave types.

4.4.2 Scattering at the shared edge between shell plates

In this subsection, we consider N composite plates modelled using SHELL181 elements connected via a shared edge, see Figure 4.2. The local y axes of all plates are directed away from the shared edge. The local x axes of all plates are all coinciding. The displacement continuity and force equilibrium conditions must be fulfilled at the shared edge. Taking the first plate as a reference plate,



Figure 4.2: A schematic representation of N plates connected via the shared edge. The plates are modelled using two-dimensional SHELL181 elements. The x axis of all plates is the same.

we write

$$\tilde{\mathbf{q}}_{L,1} = \mathbf{R}_2 \tilde{\mathbf{q}}_{L,2} = \ldots = \mathbf{R}_N \tilde{\mathbf{q}}_{L,N}, \qquad (4.27)$$

where $\tilde{\mathbf{q}}_{L,k}$ denotes the nodal displacement vector of the kth plates, and $\mathbf{R}_k, k = 2, \ldots, N$ is a matrix that transforms the local coordinate system of the kth plate to the coordinate system of the first plate. Since the local x_k axis is aligned with the x_1 axis, the transformation matrix \mathbf{R}_k can be written as

$$\mathbf{R}_{k} = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & \cos(\psi_{k}) & -\sin(\psi_{k}) & 0 & 0 & 0 \\ 0 & \sin(\psi_{k}) & \cos(\psi_{k}) & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & \cos(\psi_{k}) & -\sin(\psi_{k}) \\ 0 & 0 & 0 & 0 & \sin(\psi_{k}) & \cos(\psi_{k}) \end{bmatrix}$$
(4.28)

where ψ_k denotes the angle of rotation between y_k and y_1 . The matrices R_k are 6-by-6 since there are 6 degrees of freedom in $\mathbf{q}_{L,k}$ for plate modelled with SHELL181 elements. The sum of of forces at the shared edge must be zero,

therefore, it follows that

$$\tilde{\mathbf{f}}_{L,1} + \sum_{k=2}^{N} \mathbf{R}_k \tilde{\mathbf{f}}_{L,k} = 0. \qquad (4.29)$$

Now, expressing the nodal displacement and force vectors in the basis of wave mode shapes in each plate using Equation (4.14), we can write

$$C^{+}A^{+} + C^{-}A^{-} = 0, \qquad (4.30)$$

where

$$\mathbf{C}^{\pm} = \begin{bmatrix} \mathbf{\Phi}_{f,1}^{\pm} & \mathbf{R}_{2} \mathbf{\Phi}_{f,2}^{\pm} & \mathbf{R}_{3} \mathbf{\Phi}_{f,3}^{\pm} & \cdots & \mathbf{R}_{N} \mathbf{\Phi}_{f,N}^{\pm} \\ \mathbf{\Phi}_{q,1}^{\pm} & -\mathbf{R}_{2} \mathbf{\Phi}_{q,2}^{\pm} & 0 & \cdots & 0 \\ \mathbf{\Phi}_{q,1}^{\pm} & 0 & -\mathbf{R}_{3} \mathbf{\Phi}_{q,3}^{\pm} & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ \mathbf{\Phi}_{q,1}^{\pm} & \cdots & \cdots & -\mathbf{R}_{N} \mathbf{\Phi}_{q,N}^{\pm} \end{bmatrix}, \quad \mathbf{A}^{\pm} = \begin{cases} \mathbf{a}_{1}^{\pm} \\ \vdots \\ \mathbf{a}_{N}^{\pm} \end{cases}.$$
(4.31)

The vectors \mathbf{A}^{\pm} consist of amplitudes of incoming and outgoing waves \mathbf{a}_{k}^{\pm} of the *k*th plate, $k = 1, \ldots, N$. Finally, it follows from Equation (4.30)

$$\mathbf{A}^{+} = \mathbf{s}\mathbf{A}^{-}, \quad \mathbf{s} = -\left(\mathbf{C}^{+}\right)^{-1}\mathbf{C}^{-}, \qquad (4.32)$$

where **s** defines a $6N \times 6N$ scattering matrix. Accordingly, we can compute the energy scattering coefficients using Equation (4.24).

4.4.3 Hybrid FE/WFE method of computation of scattering coefficients in three-dimensional solid plates

This subsection considers N composite plates modelled with SOLID185 elements that are connected together via a solid joint element, see Figure 4.3. The local y axes of all plates are directed away from the joint, whereas the xaxes are aligned with the x_J axis of the joint. The joint is assumed to be periodic in the x_J axis. According to Figure 4.4, the nodal displacements vector



Figure 4.3: A schematic representation of the joint connecting N plates. The x axis of all plates are aligned with the x_J axis of the joint.

of the joint \mathbf{q}_J is organised in the following way

$$\mathbf{q}_{J} = \left\{ \mathbf{q}_{E} \quad \mathbf{q}_{O} \right\}^{\mathrm{T}}, \quad \mathbf{q}_{E} = \left\{ \mathbf{q}_{LB,1} \quad \mathbf{q}_{L,1} \quad \mathbf{q}_{LT,1} \quad \dots \quad \mathbf{q}_{LB,N} \quad \mathbf{q}_{L,N} \quad \mathbf{q}_{LT,N} \right\}, \\ \mathbf{q}_{O} = \left\{ \mathbf{q}_{B} \quad \mathbf{q}_{I} \quad \mathbf{q}_{T} \right\}$$

$$(4.33)$$

The nodal displacement vector \mathbf{q}_O consists of degrees of freedom of internal nodes. It is required that the node arrangement on the face containing $\mathbf{q}_{LB,k}$, $\mathbf{q}_{L,k}$ and $\mathbf{q}_{LT,k}$, k = 1, ..., N is coherent with one on the left face of the kth plate. We recall that m_k and $n_{x,k}$ are the number of degrees of freedom per edge of the kth plate and the number of mesh cells in the x direction of the kth plate, accordingly. Therefore, we enforce that $|\mathbf{q}_{LB,k}| = |\mathbf{q}_{L,k}| = |\mathbf{q}_{LT,k}| =$ $|\mathbf{\tilde{q}}_{L,k}| = m_k$ and $n_{J,k} = n_{x,k}$, where $n_{J,k}$ is the number of mesh cells in the joint element along the x direction of the face containing $\mathbf{q}_{LB,k}$, $\mathbf{q}_{L,k}$ and $\mathbf{q}_{LT,k}$. The nodal forces vector \mathbf{f}_J is arranged in the same way.

The governing equations of motion of the joint are of the same form as in Equation (4.1)

$$\mathbf{D}_J \mathbf{q}_J = \mathbf{f}_J, \quad \mathbf{D}_J = \mathbf{K}_J (1 + \mathrm{i}\eta) - \omega^2 \mathbf{M}_J, \qquad (4.34)$$



Figure 4.4: A finite element model of the joint connecting N plates. The degrees of freedom are grouped into internal \mathbf{q}_I , edge $\mathbf{q}_B, \mathbf{q}_T, \mathbf{q}_{L,1}, \ldots, \mathbf{q}_{L,N}$ and corner $\mathbf{q}_{LB,1}, \ldots, \mathbf{q}_{LB,N}, \mathbf{q}_{LT,1}, \ldots, \mathbf{q}_{LT,N}$ degrees of freedom.

where \mathbf{K}_J and \mathbf{M}_J are the stiffness and mass matrices of the joint. When no external forces are applied on the internal nodes of the joint, Equation (4.34) can be written as

$$\begin{bmatrix} \mathbf{D}_{EE} & \mathbf{D}_{EI} \\ \mathbf{D}_{IE} & \mathbf{D}_{II} \end{bmatrix} \begin{pmatrix} \mathbf{q}_E \\ \mathbf{q}_O \end{pmatrix} = \begin{cases} \mathbf{f}_E \\ 0 \end{cases}, \qquad (4.35)$$

where $\mathbf{f}_{O} = 0$. Consequently, one can remove internal degrees of freedom using dynamic condensation as

$$\mathbf{D}_{J,\text{cond}} \mathbf{q}_E = \mathbf{f}_E, \quad \mathbf{D}_{J,\text{cond}} = \mathbf{D}_{EE} - \mathbf{D}_{EI} \mathbf{D}_{II}^{-1} \mathbf{D}_{IE}.$$
(4.36)

Now, following the similar approach as in subsection 2.3.2, we apply periodic structure theory on the nodal displacement sub-vectors along the x_J axis as

follows

$$\mathbf{q}_{J,red} = \mathbf{T}_{J} \mathbf{q}_{E}, \quad \mathbf{q}_{J,red} = \left\{ \mathbf{q}_{LB,1} \cdots \mathbf{q}_{LB,N} \right\}^{\mathrm{T}} \\ \begin{bmatrix} \mathbf{I}_{1} & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & \vdots \\ \lambda_{x}^{j_{1}/n_{J,1}} \mathbf{I}_{1} & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & \vdots \\ \lambda_{x} \mathbf{I}_{1} & 0 & \cdots & 0 \\ 0 & \mathbf{I}_{2} & \cdots & 0 \\ \vdots & \vdots & \vdots & \vdots \\ 0 & \lambda_{x}^{j_{2}/n_{J,2}} \mathbf{I}_{2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & \lambda_{x} \mathbf{I}_{2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & \mathbf{I}_{N} \\ \vdots & \vdots & \vdots & \vdots \\ 0 & 0 & \cdots & \lambda_{x}^{j_{N}/n_{J,N}} \mathbf{I}_{N} \\ \vdots & \vdots & \vdots & \vdots \\ 0 & 0 & \cdots & \lambda_{x} \mathbf{I}_{N} \end{bmatrix}, \quad (4.37)$$

where \mathbf{I}_k is the m_k -by- m_k identity matrix and $j_k = 1, \ldots, n_{J,k} - 1, k = 1, \ldots, N$. Using Equation (4.37), we can reduce the dimension of Equation (4.36) as

$$\tilde{\mathbf{D}}_{J}\mathbf{q}_{J,red} = \mathbf{f}_{J,red}, \quad \tilde{\mathbf{D}}_{J} = \mathbf{T}_{J}^{H}\mathbf{D}_{J,\text{cond}}\mathbf{T}_{J}$$
(4.38)

with

$$\mathbf{f}_{J,red} = \left\{ \begin{aligned} \tilde{\mathbf{f}}_{LB,1} \\ \vdots \\ \tilde{\mathbf{f}}_{LB,N} \end{aligned} \right\} = \mathbf{T}_{J}^{H} \mathbf{f}_{J} = \left\{ \begin{aligned} \mathbf{f}_{LB,1} + \sum_{j_{1}=1}^{n_{J,1}-1} \lambda_{x}^{-j_{1}/n_{J,1}} \mathbf{f}_{L,1}^{j_{1}} + \lambda_{x}^{-1} \mathbf{f}_{LT,1} \\ \vdots \\ \mathbf{f}_{LB,N} + \sum_{j_{N}=1}^{n_{J,N}-1} \lambda_{x}^{-j_{N}/n_{J,N}} \mathbf{f}_{L,N}^{j_{N}} + \lambda_{x}^{-1} \mathbf{f}_{LT,N} \\ \end{aligned} \right\},$$
(4.39)

where $\mathbf{f}_{L,i}^{j_i}$ denote force sub-vectors of nodes placed at $x_{j_i} = d_x j_i / n_{L,i}$, $j_i = 1, 2, \ldots, n_{L,i} - 1$ and $i = 1, \ldots, N$. Equation (4.38) considers only interface nodes, that is, nodes shared between the plates and the joint. The force equilibrium and displacement continuity conditions must be satisfied at these nodes. Specifically, $\mathbf{q}_{LB,k}$ and $\mathbf{\tilde{f}}_{LB,k}$, $k = 1, \ldots, N$ denote nodal displacements and forces at nodes shared between the kth plate and the joint. Therefore, one can represent $\mathbf{q}_{LB,k}$ and $\mathbf{\tilde{f}}_{LB,k}$ in the wave mode shapes basis of the kth plate using Equation (4.14) as follows

$$\mathbf{q}_{LB,k} = \mathbf{R}_{k} \left(\mathbf{\Phi}_{\mathbf{q},k}^{+} \mathbf{a}_{k}^{+} + \mathbf{\Phi}_{\mathbf{q},k}^{-} \mathbf{a}_{k}^{-} \right)$$

$$\tilde{\mathbf{f}}_{LB,k} = \mathbf{R}_{k} \left(\mathbf{\Phi}_{\mathbf{f},k}^{+} \mathbf{a}_{k}^{+} + \mathbf{\Phi}_{\mathbf{f},k}^{-} \mathbf{a}_{k}^{-} \right)$$

$$(4.40)$$

where \mathbf{R}_k is the matrix that transforms the local coordinate system of the *k*th plate to the global coordinate system. Since the local x_k axis is aligned with the global x_J axis, the transformation matrix \mathbf{R}_k can be written as

$$\mathbf{R}_{k} = \begin{bmatrix} \mathbf{R}_{\text{node}} & 0 & \cdots & 0 \\ 0 & \mathbf{R}_{\text{node}} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & \mathbf{R}_{\text{node}} \end{bmatrix}, \quad \mathbf{R}_{\text{node}} = \begin{bmatrix} 1 & 0 & 0 \\ 0 & \cos(\psi_{k}) & -\sin(\psi_{k}) \\ 0 & \sin(\psi_{k}) & \cos(\psi_{k}) \end{bmatrix}$$

$$(4.41)$$

where ψ_k denotes the angle of rotation between y_k and y_J . Now, we can concatenate individual expressions (4.40) for $\mathbf{q}_{LB,k}$ and $\mathbf{f}_{LB,k}$ to express $\mathbf{q}_{J,\text{red}}$ and $\mathbf{f}_{J,\text{red}}$ in Equation (4.38)

$$\mathbf{q}_{J,\text{red}} = \mathbf{R} \left(\mathbf{\Phi}_Q^+ \mathbf{A}^+ + \mathbf{\Phi}_Q^- \mathbf{A}^- \right)$$

$$\mathbf{f}_{J,\text{red}} = \mathbf{R} \left(\mathbf{\Phi}_F^+ \mathbf{A}^+ + \mathbf{\Phi}_F^- \mathbf{A}^- \right)$$

(4.42)

with

$$\mathbf{R} = \begin{bmatrix} \mathbf{R}_{1} & \cdots & 0 \\ \vdots & \ddots & \vdots \\ 0 & \cdots & \mathbf{R}_{N} \end{bmatrix}, \ \mathbf{\Phi}_{Q}^{\pm} = \begin{bmatrix} \mathbf{\Phi}_{\mathbf{q},1}^{\pm} & \cdots & 0 \\ \vdots & \ddots & \vdots \\ 0 & \cdots & \mathbf{\Phi}_{\mathbf{q},N}^{\pm} \end{bmatrix}, \ \mathbf{\Phi}_{F}^{\pm} = \begin{bmatrix} \mathbf{\Phi}_{\mathbf{f},1}^{\pm} & \cdots & 0 \\ \vdots & \ddots & \vdots \\ 0 & \cdots & \mathbf{\Phi}_{\mathbf{f},N}^{\pm} \end{bmatrix}.$$

$$(4.43)$$

Substituting Equation (4.42) in Equation (4.38), we obtain

$$\mathbf{A}^{+} = \mathbf{s} \, \mathbf{A}^{-}, \quad \text{with} \quad \mathbf{s} = -\left(\tilde{\mathbf{D}}_{J} \mathbf{R} \boldsymbol{\Phi}_{Q}^{+} - \mathbf{R} \boldsymbol{\Phi}_{F}^{+}\right)^{-1} \left(\tilde{\mathbf{D}}_{J} \mathbf{R} \boldsymbol{\Phi}_{Q}^{-} - \mathbf{R} \boldsymbol{\Phi}_{F}^{-}\right).$$

$$(4.44)$$

As in subsection 4.4.2, the expression for **s** defines a scattering matrix relating the amplitudes \mathbf{A}^- and \mathbf{A}^+ of incoming and outgoing waves, respectively. The dimension of the scattering matrix **s** is $\sum_{k=1}^{N} m_k$, and scattering coefficients have the form $s_{ij}^{nm}(\omega, k_x)$, relating an incoming wave *i* in the plate *n* and a reflected or transmitted wave *j* in the plate *m* at angular frequency ω and wave number component k_x . Consequently, the energy scattering coefficients are defined as $t_{ij}^{nm}(\omega, k_x)$ according to Equation (4.24). It is reminded that in the absence of damping, total energy must be conserved; hence, the sum of energy scattering coefficients over the outgoing modes equals one, see Equation (4.26).

4.5 Numerical results

In this section, several numerical case examples are presented to demonstrate the applicability of the developed method. We compute energy scattering coefficients of an L-junction of regular symmetric cross- and angle-ply laminated plates for various incoming waves. These results are compared with ones obtained using the semi-analytical approach presented in Chapter 3. In all cases, the energy scattering coefficients are computed with respect to the frequency fand the wave number component k_x . We assume that the system is undamped; however, to facilitate the wave tracking process described in subsection 4.3.1, a small damping coefficient $\eta = 0.00001$ is added.



Figure 4.5: Wave vector (left) and dispersion (right) curves for a cross-ply composite plate. The wave vector curves are plotted for a fixed frequency f = 3000 Hz, whereas the dispersion curves are plotted for a fixed wave number component $k_x = 5m^{-1}$. Squares and circles denote wave numbers of incoming and outgoing waves, respectively.

4.5.1 Cross-ply laminated plates

A five-layer symmetric cross-ply laminated plate of the total thickness of h = 0.005 m is considered. The material characteristics of all layers are the same and given in Table 3.1. The lamination scheme is $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$.

A periodic cell of length $d_x = 0.001$ m and width $d_y = 0.001$ m is modelled in Ansys with 1 SHELL181 element and 3 SOLID185 elements per ply, i.e. 15 finite solid elements in total. The usage of only one element in the crosssection is justified since the laminates considered are homogeneous in their plane dimensions. However, there must be at least 6-10 FE elements per wavelength to obtain accurate results. In other words, the wave numbers $k \leq \frac{2\pi}{10 \max(d_x, d_y)}$ can be computed accurately. One can use more elements in the cross-section to alleviate round-off errors due to truncation of inertia terms in the dynamic stiffness matrix if needed [127].

As presented in Section 4.3, solving Equations (4.6) or (4.8) yields propagating wave vector pairs (k_x, k_y) for a fixed frequency ω^* or dispersion curves $k_y = k_y(\omega, k_x^*)$ for a fixed wave number component k_x^* . Figure 4.5 presents bending, shear and longitudinal wave vector curves for a fixed frequency f = 3000 Hz on the left side and the correspondent dispersion curves for a fixed wave number component $k_x = 5m^{-1}$ on the right side. These numerical dispersion relations



Figure 4.6: Energy scattering coefficients of an L-junction of two SHELL181-based cross-ply laminated plates for various incident waves plotted with respect to the wave number component k_x at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

can be used to calculate the group velocity vectors and, therefore, propagation angles of transmitted wave energies via application of (4.19) and (4.21), respectively.

Now, we consider two identical cross-ply composite plates connected at 90° if modelled by SHELL181 elements and through an L-joint modelled as in Figure 4.4 - if modelled by SOLID185 elements. The FE model of the joint consists of 55 SOLID185 elements; thus, the dimensions of the joint stiffness and mass matrices are 164-by-164. The energy scattering coefficients for SHELL181based plates are computed using Equation (4.32), whereas for SOLID185-based plates - using Equation (4.44). Furthermore, numerical results are compared with semi-analytical energy scattering coefficients obtained in Section 3.7. Figure 4.6 presents a comparison of energy scattering coefficients obtained



Figure 4.7: Energy scattering coefficients of an L-junction of two SOLID185-based cross-ply laminated plates for various incident waves plotted with respect to the wave number component k_x at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

from the SHELL181-based wave finite element approach presented in subsection 4.4.2 and from the theoretical approach developed in Chapter 3 for various incident modes as a function of wave number component k_x at the frequency f = 3000 Hz. All solid and dashed lines represent numerical results, whereas circles and squares - semi-analytical results. A slight deviation between numerical and theoretical energy scattering coefficients can be noted for an incoming bending wave results in Figure 4.6a. Overall, however, one can observe an excellent agreement between numerical and analytical results for all cases of incoming modes. This is an expected result since SHELL181 elements are modelled in ANSYS using the FSDL theory [67, 204]. The CLP theory presented in Section 2.2 and used in Chapter 3 and the FSDL theories are expected to give identical results [204].



Figure 4.8: Energy scattering coefficients of an L-junction of two SHELL181based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 0 m^{-1}$. \hat{t}_{ij}^{nm} denote semianalytical scattering coefficients obtained using the method presented in Chapter 3.



Figure 4.9: Energy scattering coefficients of an L-junction of two SOLID185-based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 0 m^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

If we consider SOLID185-based plates and a solid L-joint, then discrepancies between numerical and theoretical results appear. For instance, from Figure 4.7a it can be noted that theoretical results seem to underestimate energy reflection and hence, overestimate energy transmission of an L-junction of plates. In fact, the maximum difference observed between bending numerical and theoretical results is $\sim 20 - 22\%$. This can be referred to the fact that we used the analytical model based on the assumption that the joint can be represented as a shared line between plates. This assumption breaks down at higher frequencies since the influence of the joint size becomes more significant. Notably, the shear strain becomes more critical in the dynamic response



Figure 4.10: Energy scattering coefficients of an L-junction of two SHELL181based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 5 m^{-1}$. \hat{t}_{ij}^{nm} denote semianalytical scattering coefficients obtained using the method presented in Chapter 3.

of the joint and this effect is not considered in the analytical model [160, 205]. The energy scattering coefficients for incoming in-plane modes are in excellent agreement. Hence, modelling thin plates and junctions of thin plates using SHELL181 elements or semi-analytically is sufficient for accurate estimation of in-plane wave energy scattering coefficients.

If the wave number component k_x is fixed, then one can compare energy scattering coefficients for a range of frequencies. Figures 4.8 and 4.9 represent energy scattering coefficients obtained from the SHELL181- and SOLID185based approaches for incident longitudinal and bending modes as a function of frequency at the wave number component $k_x = 0 \text{ m}^{-1}$. An incident shear wave is fully reflected without mode conversion for the wave number component $k_x = 0$, that is, $t_{SS}^{11} = 1$. A great agreement of semi-analytical and numerical



Figure 4.11: Energy scattering coefficients of an L-junction of two SOLID185based cross-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 5 m^{-1}$. \hat{t}_{ij}^{nm} denote semianalytical scattering coefficients obtained using the method presented in Chapter 3.

results for longitudinal energy scattering coefficients can be noted. The longitudinal mode is propagating at all frequencies at the wave number component $k_x = 0$. Regarding the incident bending mode, the semi-analytical approach predicts that the bending energy reflection and transmission coefficients are nearly same in magnitude across the range of frequencies $f \in [0.1, 12]$ kHz, see Figures 4.8b and 4.9b. However, the bending energy reflection coefficient obtained from the hybrid FE/WFE approach is larger than the transmission coefficient, and the magnitude difference increases with the frequency, becoming 0.3 at f = 12 kHz. Nevertheless, the maximum difference between numerical and semi-analytical scattering coefficients is ~ 0.2.

A mode coupling phenomenon can be observed when considering the dependence of energy scattering coefficients on the frequency at the wave number component $k_x \neq 0 \text{ m}^{-1}$. For instance, in Figures 4.10 and 4.11, energy scattering coefficients are computed for a fixed wave number component $k_x = 5 \text{ m}^{-1}$ and obtained from SHELL181- and SOLID185-based hybrid FE/WFE approach, respectively. An incident bending wave generates outgoing shear and longitudinal waves at frequencies 1416 Hz $\leq f < 5701$ Hz and $f \geq 5701$ Hz, respectively, see Figures 4.10a and 4.11a. The energy scattering coefficients of coupled shear-bending waves are equal, that is, $t_{SB}^{11(12)} = t_{BS}^{11(12)}$. The same applies for shear-longitudinal wave coupling, that is, $t_{SL}^{11(12)} = t_{LS}^{11(12)}$. Overall, a great agreement between numerical and semi-analytical results can be noted with small deviations for incident bending wave energy scattering coefficients. The summation to unity of the energy reflection and transmission coefficients validates the numerical results obtained.

4.5.2 Angle-ply composite plates

A five-layer angle-ply laminated plate of the total thickness of h = 0.005 m is considered. The material characteristics of the individual layers are the same as in the previous example. The lamination scheme is $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$. A periodic cell of length $d_x = 0.001$ m and width $d_y = 0.001$ m is again modelled in Ansys with 1 SHELL181 element and 15 SOLID185 elements. Figure 4.12 presents bending, shear and longitudinal wave vector curves of an angle-ply laminated plate for a fixed frequency f = 3000 Hz on the left side and the correspondent dispersion curves for a fixed wave number component $k_x =$ $5m^{-1}$ on the right side. Note that there are two shear wave dispersion curves present on the right side of Figure 4.12. The second shear waves (denoted as S_2 and plotted in green) exhibit a negative group velocity phenomenon as in wave vector curves at a fixed frequency.

A comparison of energy scattering coefficients obtained from the SHELL181based numerical and theoretical methods is given in Figure 4.13 for various incident modes as a function of wave number component k_x at a frequency f = 3000 Hz. Again, numerical and theoretical energy scattering coefficients


Figure 4.12: Wave vector (left) and dispersion (right) curves for an angleply composite plate. The wave vector curves are plotted for a fixed frequency f = 3000 Hz, whereas the dispersion curves are plotted for a fixed wave number component $k_x = 5m^{-1}$. Squares and circles denote wave numbers of incoming and outgoing waves, respectively.

agree well for all incoming modes with slight deviation for an incoming bending mode.

A drastic change in the shape of energy reflection and transmission coefficients can be seen for an incoming bending wave in the case of SOLID185-based numerical results, see Figure 4.14. For instance, at the range of wave number component 10 m⁻¹ < $|k_x| < 20$ m⁻¹, numerical energy transmission is around 0.6 whereas semi-analytical method estimates energy transmission at only 0.4. Vice-versa, numerical reflection coefficient has value of 0.4, whereas theoretical reflection - 0.6, see Figure 4.14a. The same discrepancy can be noted between numerical results obtained from SOLID185- and SHELL181-based approaches. This phenomenon can be explained by the fact that the ESLtheories behind the semi-analytical and SHELL181-based approaches produce effective one-layer plates joined along the shared edge, thus losing complexity of the connection between individual layers of plates at the junction.

Numerical energy scattering coefficients for in-plane modes are in excellent agreement with semi-analytical results. As in the case of cross-ply laminated plates, curves for shear and bending coupled waves vary identically for a range of wave number components $|k_x| < 7.6 \text{ m}^{-1}$ - the correspondent scattering coefficients are $t_{SB}^{11(12)} = t_{BS}^{11(12)}$. Similarly, the scattering coefficients of longitu-



(a) Incoming bending wave B



Figure 4.13: Energy scattering coefficients of an L-junction of two SHELL181based angle-ply laminated plates for various incident waves plotted with respect to the wave number component k_y at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

dinal and bending coupled waves $t_{LB}^{11(12)}$ and $t_{BL}^{11(12)}$ are equal for $|k_x| < 3.7 \text{ m}^{-1}$. Furthermore, energy scattering coefficients of longitudinal and shear coupled waves are symmetric around $k_x = 0 \text{ m}^{-1}$, that is, $t_{SL}^{11(12)}(k_x) = t_{LS}^{11(12)}(-k_x)$.

Figures 4.15 and 4.16 represent energy scattering coefficients as a function of frequency at the fixed wave number components $k_x = 0 \text{ m}^{-1}$ and $k_x = 5 \text{ m}^{-1}$, respectively. It can be noted that numerical scattering coefficients based on the SHELL181 finite elements agree well with semi-analytical ones for all incident waves. This is an expected result, since the semi-analytical and SHELL181based numerical approaches use approximately equal formulations of the thin laminated plate at the frequencies considered, see Figures 4.15a, 4.15c and 4.15e for $k_x = 0 \text{ m}^{-1}$ and Figures 4.16a, 4.16c and 4.16e for $k_x = 5 \text{ m}^{-1}$.





Figure 4.14: Energy scattering coefficients of an L-junction of two SOLID185based angle-ply laminated plates for various incident waves plotted with respect to the wave number component k_y at a frequency f = 3000 Hz. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

On the other hand, in Figures 4.15b and 4.16b, numerical results based on the SOLID185 finite elements predict higher bending wave energy reflection across all frequencies considered for both values of the wave number component k_x .

4.6 Conclusion

This chapter aimed to derive the hybrid FE/WFE method to calculate of energy scattering coefficients for arbitrary junctions of composite plates. The approach is based on modelling joints with finite elements with boundary conditions given by the solutions of the WFE method for the composite plates in the infinite half-spaces connected to the joint. The displacement and force nodal vectors are expressed in the basis of positive and negative waves, and satisfying boundary conditions at the connection between the plates and the joint yields the scattering matrix. The method gives for the first time a detailed recipe for computing scattering coefficients for the generic case of an arbitrary number of composite plates connected at a junction without restrictions on the angles at which the plate meet or the orientation of the principal axis of individual plates.

The plates considered have been modelled using two different finite element types, two-dimensional SHELL181 and three-dimensional SOLID185 elements, and we have considered the implication and influence of each element type on the scattering coefficients. Also, numerical results on selected examples of junctions of composite plates have been compared with semi-analytical results obtained from Chapter 3. It has been found that modelling thin plates and their junctions using SHELL181 elements or semi-analytically using the method from Chapter 3 is sufficient for accurate estimation of in-plane wave energy scattering coefficients. However, the SOLID185-based approach provides more accurate energy scattering coefficients of the incident bending wave at higher frequencies since the influence of the shear strain becomes significant, which is neglected in the semi-analytical method and approximated by shear correction factors in the SHELL181-based approach. Nevertheless, in the selected examples, the maximum difference observed between bending numerical and theoretical results is $\sim 20 - 22\%$. The results of this chapter can be used for the computation of wave energy distribution in the SEA method in the form of coupling loss factors and as well as for angle-of-incidence dependent scattering coefficients entering the DEA method in Chapter 5.



Figure 4.15: Energy scattering coefficients of an L-junction of two SHELL181and SOLID185-based angle-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 0 \ m^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.



Figure 4.16: Energy scattering coefficients of an L-junction of two SHELL181and SOLID185-based angle-ply laminated plates for various incident waves plotted with respect to the frequency at the fixed wave number component $k_x = 5 m^{-1}$. \hat{t}_{ij}^{nm} denote semi-analytical scattering coefficients obtained using the method presented in Chapter 3.

Chapter 5

Dynamical Energy Analysis for composite structures

5.1 Introduction

Having computed and validated the scattering coefficients for the composite laminated plate junctions in Chapters 3 and 4, respectively, we will next consider the implementation of the DEA method for composite structures. The main ideas of this method were presented in Chapter 2, and the general algorithm was given in (2.42). The actual implementation of the method will be performed on two-dimensional FE triangular meshes.

This chapter is organised as follows. In Section 5.2, we first discuss how the domain Ω is meshed using two-dimensional triangular cells. In subsection 5.2.1, we consider how the ray tracing formula presented in Equation (2.21) are changed in the case of composite laminated plates. Furthermore, they define the linear transformation between the boundary (2D) and full (4D) phase space densities described in subsection 5.2.2. In Section 5.3, we introduce a finite basis of functions onto which we project the phase space densities and the boundary operator T, which is given by Equation (2.38). The discretisation of the boundary integral operator T on the triangular meshes is discussed in subsection 5.3.1. The initial boundary phase space density due to a point or an edge source is also expressed in terms of the finite basis functions in subsection 5.3.2. The stationary energy density and the intensity vector field are computed in subsection 5.3.3. Finally, Section 5.4 considers numerical case



Figure 5.1: An example of an electric vehicle gearbox being meshed using twodimensional shell elements. Colours represent elements with different thicknesses.

studies for single and two coupled composite plates. Also, it presents part of the work performed during the industrial placement at Romax Technology.

5.2 Ray tracing for composite structures

From here on, we assume that the whole domain Ω is subdivided into subdomains $\Omega_i, i = 1, \ldots, N$, with N being the total number of subsystems. For instance, this can be naturally achieved by meshing the domain using twodimensional finite shell elements in triangular form. It is also assumed that all material and geometrical parameters remain constant within the mesh cells. For instance, a shell-element based mesh cell must be homogeneous in material parameters and of uniform thickness. Figure 5.1 presents an example of a two-dimensional finite element mesh for an electric vehicle gearbox, where different colours represent different thicknesses of the shell elements.

5.2.1 Hamilton function and boundary mapping

The wave energy travels between neighbouring mesh cells through edges. Now, the trajectories of the ray-particles carrying the wave energy are governed by the ray-tracing map defined by Equation (2.21). This was outlined in subsection 2.1.3 for isotropic structures with the Hamilton function $H(\mathbf{r}, \mathbf{p}) =$ $c|\mathbf{p}| = 1$. As was seen in subsection 3.3.1, the Hamilton function for composite plates can be written in the similar manner as

$$H(\mathbf{r}, \mathbf{p}) = \beta_l \left(\arctan\left(\frac{p_{\scriptscriptstyle \parallel}}{p_{\perp}}\right) \right) |\mathbf{p}|^l = \omega, \quad l = 1, 2.$$
 (5.1)

Here, the momentum vector \mathbf{p} is equal to the wave vector \mathbf{k} with its amplitude $|\mathbf{p}|$ being the wave number k, and consequently, from here on, we will denote the wave number components k_x and k_y as p_{\parallel} and p_{\perp} , respectively. The function $\beta_l(\theta)$ is given by Equation (3.8). The parameter l = 1, 2 represents propagating in-plane and bending modes, respectively.

The Hamilton function in (5.1) represents only in-plane and out-of-plane modes in the context of the CLP theory presented in Chapter 3. The more general Hamilton function for arbitrary composite plates can be obtained numerically using the WFE method described in Chapter 4. It then can describe not only classical longitudinal, shear and bending modes but also higher-order Lamb modes. For the sake of clarity, the Hamilton function (5.1) will be used throughout this chapter; however, all calculations can be applied to the numerical Hamilton function obtained from the WFE method.

As per Equation (2.21), we can write the equations of motion of the rayparticles as

$$\begin{bmatrix} \frac{\mathrm{d}x}{\mathrm{d}\tau} \\ \frac{\mathrm{d}y}{\mathrm{d}\tau} \\ \frac{\mathrm{d}p_{\mathrm{l}}}{\mathrm{d}\tau} \\ \frac{\mathrm{d}p_{\mathrm{l}}}{\mathrm{d}\tau} \\ \frac{\mathrm{d}p_{\mathrm{l}}}{\mathrm{d}\tau} \end{bmatrix} = \begin{bmatrix} \frac{\partial H}{\partial p_{\mathrm{l}}} \\ \frac{\partial H}{\partial p_{\mathrm{l}}} \\ -\frac{\partial H}{\partial x} \\ -\frac{\partial H}{\partial y} \end{bmatrix} = \begin{bmatrix} \left(lp_{\mathrm{l}} + \frac{\partial \beta_{l}}{\beta_{l}\partial\theta}p_{\mathrm{l}} \right) \frac{\omega}{k^{2}} \\ \left(lp_{\perp} - \frac{\partial \beta_{l}}{\beta_{l}\partial\theta}p_{\mathrm{l}} \right) \frac{\omega}{k^{2}} \\ 0 \\ 0 \end{bmatrix}, \quad l = 1, 2.$$
(5.2)

Here, $(x, y, p_{\parallel}, p_{\perp})$ are the 4D coordinates in the full phase space domain. Since $H(r, p) = \omega$, then $\frac{\partial H}{\partial p_{\parallel}}$ and $\frac{\partial H}{\partial p_{\perp}}$ are the components of the group velocity vector \mathbf{c}_{g} . Therefore, ray-particles travel along the direction of the group velocity vector. As in the case of isotropic structures considered in subsection 2.1.3, the ray-particle trajectories are straight lines, since $\frac{\mathrm{d}\mathbf{p}}{\mathrm{d}\tau} = 0$.

The boundary mapping procedure described in subsection 2.1.4 is based on



Figure 5.2: A schematic representation of the boundary mapping procedure in the triangular mesh element. Green ovals depict wave vector curves. The orientation of the principal material axes is defined in the element's coordinate system (x_g, y_g, z_g) . A ray starting at the position s of edge a points in the direction of the outgoing group velocity vector \mathbf{c}_g^+ . The group velocity vector is defined by the momentum component p_{11} . The ray hits edge a' at the position s' with the new momentum component p'_{11} defined in the coordinate system (e'_x, e'_y) . However, momentum vectors \mathbf{p} and \mathbf{p}' are equal.

the integral operator T:

$$\rho_{\Gamma}'(X') = \{T\rho_{\Gamma}\}(X') = \int \lambda(X')e^{-\mu D(X,X')}\delta(X' - \Phi(X))\rho_{\Gamma}(X)dX, \quad (5.3)$$

where $\rho_{\Gamma}(X)$ is the 2D phase space density on the boundary Γ . The mapping function $\Phi : X \to X'$ is now described by the ray-tracing equations (5.2). Figure 5.2 shows the schematic representation of this mapping between the boundary phase space coordinates $X = (s, p_{\parallel})$ and $X' = (s', p'_{\parallel})$ at edges a and a' of a single mesh cell, respectively. Here, the green ovals represent the wave vector curves of a particular mode, and they are the same for both edges because the material parameters are constant in the mesh cell. The momentum component p_{\parallel} is used to compute the outgoing group velocity vector \mathbf{c}_{g}^{+} (depicted in red) and the direction angle of the trajectory α^{+} measured with respect to the local e_{y} axis. Note that $|p_{\parallel}| \leq p_{\parallel,a}$, and the maximum value $p_{\shortparallel,a}$ can be found from the dispersion relations. Furthermore, the value of $p_{\shortparallel,a}$ changes for different propagating modes. The red line connecting points s and s' represents the trajectory of the ray-particle. Now, since the momentum vector \mathbf{p} is conserved, the vector \mathbf{p}' is equal to \mathbf{p} in the global coordinate system (x_g, y_g) of the mesh cell. However, at the local edge coordinate system (e'_x, e'_y) the components p'_{\shortparallel} and p'_{\perp} will be different from p_{\shortparallel} and p_{\perp} . Also, the maximum values $p_{\shortparallel,a}$ and $p'_{\amalg,a'}$ are also different. The component p'_{\shortparallel} is further used to compute the scattering coefficients $\lambda(p'_{\shortparallel})$. These coefficients have been previously obtained in the context of the classical thin laminated plate in Equation (3.43) and using the WFE method in Equation (4.24).

5.2.2 Transformation between 2D and 4D phase space densities

Equation (5.2) defines the linear transformation Π between $(s, \tau, p_{\Pi}, \omega)$ and the full phase space coordinates $(x, y, p_{\Pi}, p_{\perp})$ [53]:

$$\Pi: \begin{bmatrix} s \\ \tau \\ p_{\Pi} \\ \omega \end{bmatrix} \mapsto \begin{bmatrix} x \\ y \\ p_{\Pi} \\ p_{\perp} \end{bmatrix} + \frac{\omega\tau}{k^{2}} \begin{bmatrix} lp_{\Pi} + \frac{\partial\beta_{L}}{\beta_{L}\partial\theta}p_{\perp} \\ lp_{\perp} - \frac{\partial\beta_{L}}{\beta_{L}\partial\theta}p_{\Pi} \\ 0 \\ 0 \end{bmatrix}.$$
(5.4)

The Jacobian matrix $D\Pi$ can be written as

$$D\Pi = \frac{\partial \left(x, y, p_{\parallel}, p_{\perp}\right)}{\partial \left(s, \tau, p_{\parallel}, \omega\right)} = \begin{bmatrix} 1 & \frac{\partial H}{\partial p_{\parallel}} & 0 & 0\\ 0 & \frac{\partial H}{\partial p_{\perp}} & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & \frac{\partial p_{\perp}}{\partial p_{\parallel}} & \frac{\partial p_{\perp}}{\partial \omega} \end{bmatrix}.$$
 (5.5)

Now, at a fixed angular frequency $\omega = \omega_0$, one can compute $\frac{\partial p_{\perp}}{\partial p_{\parallel}}$ using Equation (5.1) in the following way

$$0 = \mathrm{d}\omega = \frac{\partial H}{\partial p_{\scriptscriptstyle ||}} \mathrm{d}p_{\scriptscriptstyle ||} + \frac{\partial H}{\partial p_{\perp}} \mathrm{d}p_{\perp} \to \frac{\partial p_{\perp}}{\partial p_{\scriptscriptstyle ||}} = -\frac{c_{g,x}}{c_{g,y}}, \quad l = 1, \ 2.$$
(5.6)

This relation will also be important in the computation of the stationary energy and intensity densities in subsection 5.3.3. Finally, the determinant of the Jacobian matrix $\det(D\Pi) = \frac{\partial H}{\partial p_{\perp}} \frac{\partial p_{\perp}}{\partial \omega} = 1$, therefore, the transformation Π is phase space volume preserving, and the full phase space density can be related to the 2D phase space density as

$$\rho(x, y, p_{\shortparallel}, p_{\perp}) = \rho_{\Gamma}(x, p_{\shortparallel})\delta(\omega(p_{\shortparallel}, p_{\perp}) - \omega_0).$$
(5.7)

Recall that the 2D phase space density represents the directional energy flux through the boundary Γ at the position x and momentum component p_{\parallel} . The relation (5.7) will be used to define the initial 2D phase space density $\rho_{\Gamma,0}(s, p_{\parallel})$ and the stationary energy density $\rho^{\infty}(\mathbf{r})$ in further subsections.

5.3 Basis set expansion of the problem

To proceed further with the DEA method, we need to introduce a finite basis set of functions onto which the 2D phase space densities and the boundary integral operator T will be projected. As outlined in Chapter 2, Legendre polynomials will be used for this purpose because of their orthogonality property in the L^2 space. They can be defined by *Bonnet's recursion formula* as

$$P_0(x) = 1, P_1(x) = x, \quad x \in [-1, 1].$$

$$(m+1)P_{m+1}(x) = (2m+1)xP_m(x) - mP_{m-1}(x), \quad m \ge 2.$$
(5.8)

Legendre polynomials are orthogonal with respect to the L^2 -norm on the interval $-1 \le x \le 1$:

$$\int_{-1}^{1} P_n(x) P_m(x) dx = \frac{2}{2m+1} \delta_{nm} , \qquad (5.9)$$

where δ_{mn} denotes the Kronecker delta. Since the boundary operator T and phase space densities $\rho_{\Gamma}(s, p_{\parallel})$ operate on the 2D phase space (s, p_{\parallel}) , the basis functions must be defined accordingly both for position and momentum spaces. We define the basis functions in a similar way as in [53]:

$$F_{\mathbf{n}=(a,m,l)}(s,p_{\scriptscriptstyle ||}) = \frac{1}{\sqrt{L_a,p_{\scriptscriptstyle ||,a,l}}} \mathbb{1}_a(s) P_m\left(\frac{p_{\scriptscriptstyle ||}}{p_{\scriptscriptstyle ||,a,l}}\right) , \ \mathbb{1}_a(s) = \begin{cases} 1 & \text{for } s \text{ on edge } a. \\ 0 & \text{otherwise} \end{cases}$$
(5.10)

where $\mathbb{1}_a$ is the characteristic function of edge a, and L_a is the length of edge a. The multi-index $\mathbf{n} = (a, m, l)$ combines the index of the edge a, the degree m of the Legendre polynomial and the index l of the propagating mode, where l = L, S, B represents longitudinal, shear and bending modes propagating in thin plates, see subsection 3.3.1. As outlined earlier, the higher-order Lamb modes can be taken into account by computing the Hamilton function numerically in the WFE method context, thus enlarging the dimension of the index l.

The Legendre polynomials are written for the transformed argument $|x| \leq 1 \rightarrow |p_{\parallel}| \leq p_{\parallel,a,l}$. By defining the inner product as

$$(u,v) = \int u(X)v(X)dX, \qquad (5.11)$$

we get the orthogonality property of the basis functions:

$$(F_{\mathbf{n}}, F_{\mathbf{n}'}) = \frac{2}{2m+1} \delta_{a,a'} \,\delta_{m,m'} \,\delta_{l,l'} \,. \tag{5.12}$$

5.3.1 Boundary integral operator

After introducing the finite basis set of functions, we can now express the phase space densities and the operator T in this basis. The 2D phase space densities $\rho_{\Gamma}(s, p_{\parallel})$ and $\rho'_{\Gamma}(s', p'_{\parallel})$ can be approximated as

$$\rho_{\Gamma}(s, p_{\scriptscriptstyle ||}) \approx \sum_{\mathbf{n}} \rho_{\mathbf{n}} F_{\mathbf{n}}(s, p_{\scriptscriptstyle ||}), \quad \rho_{\Gamma}'(s', p_{\scriptscriptstyle ||}') \approx \sum_{\mathbf{n}'} \rho_{\mathbf{n}'}' F_{\mathbf{n}'}(s', p_{\scriptscriptstyle ||}'), \tag{5.13}$$

where $\rho'_{\Gamma}(s', p'_{\parallel})$ is the 2D phase space density on edge a' obtained after a single application of the boundary integral operator T, that is, $\rho'_{\Gamma} = \{T\rho_{\Gamma}\}$. The coefficients $\rho_{\mathbf{n}}$ and $\rho'_{\mathbf{n}'}$ are given as

$$\rho_{\mathbf{n}} = \frac{(\rho_{\Gamma}, F_{\mathbf{n}})}{(F_{\mathbf{n}}, F_{\mathbf{n}})}, \quad \rho_{\mathbf{n}'}' = \frac{(\rho_{\Gamma}', F_{\mathbf{n}'})}{(F_{\mathbf{n}'}, F_{\mathbf{n}'})}.$$
(5.14)

Consequently, the coefficients ρ'_n are connected with ρ_n through the following relation

$$\rho_{\mathbf{n}'}' = \sum_{\mathbf{n}} T_{\mathbf{n}'\mathbf{n}}\rho_{\mathbf{n}}, \quad T_{\mathbf{n}'\mathbf{n}} = \frac{(F_{\mathbf{n}'}, TF_{\mathbf{n}})}{(F_{\mathbf{n}'}, F_{\mathbf{n}'})}, \quad (5.15)$$

where $T_{\mathbf{n'n}}$ is the matrix representation of the boundary integral operator T. After substituting Equation (5.3) into the matrix $T_{\mathbf{n'n}}$, we obtain the following integral form:

$$T_{\mathbf{n'n}} = \frac{2m'+1}{2} \iint F_{\mathbf{n'}}(X')\lambda(X')e^{-\mu D(X,X')}\delta(X'-\Phi(X))F_{\mathbf{n}}(X)dXdX' = = \frac{2m'+1}{2} \int \lambda(\Phi(X))e^{-\mu D(X,\Phi(X))}F_{\mathbf{n'}}(\Phi(X))F_{\mathbf{n}}(X)dX.$$
(5.16)

Finally, in the chosen finite set of basis functions, the matrix elements $T_{n'n}$ can be written as

$$T_{\mathbf{n'n}} = \frac{2m' + 1}{2\sqrt{L_a L_{a'} p_{{\scriptscriptstyle \parallel},a,l} p'_{{\scriptscriptstyle \parallel},a',l'}}} \int_{0-p_{{\scriptscriptstyle \parallel},a,l}} \int_{0-p_{{\scriptscriptstyle \parallel},a,l}} \int_{0-p_{{\scriptscriptstyle \parallel},a,l}} \lambda(p'_{{\scriptscriptstyle \parallel}}) e^{-\mu D(s,s')} \mathbb{1}_{a'}(s') \mathbb{1}_{a}(s) P_{m}\left(\frac{p_{{\scriptscriptstyle \parallel}}}{p_{{\scriptscriptstyle \parallel},a,l}}\right) P_{m'}\left(\frac{p'_{{\scriptscriptstyle \parallel}}}{p'_{{\scriptscriptstyle \parallel},a',l'}}\right) dp_{{\scriptscriptstyle \parallel}} ds .$$
(5.17)

The double integral can be reduced to a single integral with respect to the momentum component p_{\parallel} as follows:

$$T_{\mathbf{n'n}} = \frac{2m'+1}{2\sqrt{L_a L_{a'} p_{{\scriptscriptstyle ||,a,l}} p'_{{\scriptscriptstyle ||,a',l'}}}} \int_{-p_{{\scriptscriptstyle ||,a,l}}}^{p_{{\scriptscriptstyle ||,a,l}}} \lambda(p'_{{\scriptscriptstyle ||}}) h(\mu, p_{{\scriptscriptstyle ||}}) P_m\left(\frac{p_{{\scriptscriptstyle ||}}}{p_{{\scriptscriptstyle ||,a,l}}}\right) P_{m'}\left(\frac{p'_{{\scriptscriptstyle ||}}}{p'_{{\scriptscriptstyle ||,a',l'}}}\right) \mathrm{d}p_{{\scriptscriptstyle ||}}$$
(5.18)

with

$$h(\mu, p_{\scriptscriptstyle \Pi}) = \int_{0}^{L_a} e^{-\mu D(s,s')} \mathbb{1}_{a'}(s') \mathbb{1}_a(s) \mathrm{d}s = \int_{s_{\min}}^{s_{\max}} e^{-\mu D(s,s')} \mathrm{d}s \,. \tag{5.19}$$

The function $h(\mu, p_{\parallel})$ can be computed analytically. In fact, D(s, s') is the distance between points s and s' in Figure 5.3, and it can be computed using



Figure 5.3: A representation of the geometrical relations used for computation of the position integral in Equation (5.19). A ray emitting from edge a can only hit edge a' if its direction angle $\alpha^+ > \xi^-$. D(s, s') is the ray length. $s_{min(max)}$ is the minimum (maximum) value of the position s in the local coordinate system (e_x, e_y) such that a ray with the direction angle α^+ reach edge a' at some position s'.

the sine rule in the triangle connecting points s, s' and s_{\max} as

$$D(s,s') = \frac{s_{\max} - s}{\sin \alpha'} \sin \phi = \frac{s_{\max} - s}{\cos(\phi - \alpha^+)} \sin \phi, \qquad (5.20)$$

where the angle α^+ is defined by the momentum component p_{\shortparallel} through the dispersion relations. After substituting Equation (5.20) into Equation (5.19), one obtains

$$h(\mu, p_{\rm H}) = \frac{1 - e^{-\tilde{\mu}(s_{\rm max} - s_{\rm min})}}{\tilde{\mu}}, \quad \tilde{\mu} = \frac{\mu \sin \phi}{\cos(\phi - \alpha^+)}, \quad \mu \neq 0.$$
(5.21)

Now, we need to compute $s_{\max} - s_{\min}$, which is equal to $h(0, p_{\parallel})$, according to Equation (5.19). For $\alpha^+ \in [-\frac{\pi}{2}, \xi_-]$, $h(0, p_{\parallel})$ is equal to zero, since no rays with such direction angles α^+ can reach the edge a' from the edge a, see Figure 5.3. On the other hand, for $\alpha^+ \in [\xi_+, \frac{\pi}{2}]$, $h(0, p_{\parallel})$ is equal to L_a , that is, the length of the edge a, because all points on the edge a can be mapped to the edge a'by the rays with such direction angles α^+ . Finally, for $\alpha^+ \in (\xi_-, \xi_+)$, using the sine rule in the triangle connecting points s_{\min} , s_{\max} and the top vertex of the mesh cell, one gets $h(0, p_{\parallel}) = L_{a'} \frac{\cos(\phi - \alpha^+)}{\cos \alpha^+}$. We can summarise the definition of $h(0, p_{\parallel})$ as follows

$$h(0, p_{\text{\tiny II}}) = \begin{cases} 0 & \text{for } p_{\text{\tiny II}} : \alpha^+(p_{\text{\tiny II}}) \in [-\frac{\pi}{2}, \xi_-] \\ L_{a'} \frac{\cos(\phi - \alpha^+)}{\cos \alpha^+} & \text{for } p_{\text{\tiny II}} : \alpha^+(p_{\text{\tiny II}}) \in (\xi_-, \xi_+) \\ L_a & \text{for } p_{\text{\tiny II}} : \alpha^+(p_{\text{\tiny II}}) \in [\xi_+, \frac{\pi}{2}] \end{cases}$$
(5.22)

Recall that if the material of the structure is isotropic, then all ray direction angles α^+ are equal to the wave vector angles θ^+ , thus simplifying calculation of the function $h(\mu, p_{\parallel})$. Furthermore, since wave number k is constant for any direction angle θ , the variable of integration in Equation (5.18) can be changed from the momentum component p_{\parallel} to the wave vector angle θ as follows

$$\mathrm{d}p_{\scriptscriptstyle ||} = k\cos\theta\mathrm{d}\theta\,,\tag{5.23}$$

thus removing the need of wave vector curves and rendering the integrand to be defined by the direction angle θ alone [53]. It is worth noting that if the number of Legendre polynomials used in the basis set is equal to one, then the formulation in Equation (5.18) is equivalent to the SEA approximation of the system [50, 51].

5.3.2 Initial phase space densities

In this subsection, we define and discretise the initial boundary phase space density $\rho_{\Gamma,0}(s, p_{\parallel})$ arising due to a point or an edge source. First, we consider the case of a point source at a position \mathbf{r}_0 , which is in the interior of the mesh cell. The full phase space density at a point \mathbf{r} is written in the similar way as in Equation (2.34):

$$\rho_0(\mathbf{r}, \mathbf{p}) = \frac{R}{|\mathbf{c}_{g,0}|} \frac{e^{-\mu |\mathbf{r} - \mathbf{r}_0|}}{2\pi |\mathbf{r} - \mathbf{r}_0|} \delta(\mathbf{p} - \mathbf{p}_0), \quad \mathbf{p}_0 : \mathbf{c}_{g,0} = |\mathbf{c}_{g,0}| \frac{\mathbf{r} - \mathbf{r}_0}{|\mathbf{r} - \mathbf{r}_0|}, \quad (5.24)$$

where the momentum vector \mathbf{p}_0 is such that the group velocity vector $\mathbf{c}_{g,0}(\mathbf{p}_0)$ points in the direction of $\mathbf{r} - \mathbf{r}_0$. Note that the attenuation factor μ depends on the direction of the energy flow, since $\mu = \eta \frac{\omega}{|\mathbf{c}_{g,0}|}$ by definition.

Now, one needs to determine the 2D phase space density $\rho_{a,0}(s, p_{\parallel})$ on edge a

of the mesh cell that surrounds the interior source point \mathbf{r}_0 . To accomplish this, we use the following equation relating delta functions similarly as in Equation (2.36):

$$\frac{\partial\omega}{\partial p_{\perp}}\delta(\omega(p_{\shortparallel},p_{\perp})-\omega_0)\delta(p_{\shortparallel}-p_{\shortparallel,0}) = \delta(\mathbf{p}-\mathbf{p}_0).$$
(5.25)

Applying this substitution rule to Equation (5.7), we obtain

$$\rho_{a,0}(s, p_{\rm H}) = \frac{R}{2\pi} \frac{e^{-\mu D(s, \mathbf{r}_0)}}{D(s, \mathbf{r}_0)} \cos \alpha^-(s, \mathbf{r}_0) \delta(p_{\rm H} - p_{\rm H,0}(s, \mathbf{r}_0)) , \qquad (5.26)$$

where $\cos \alpha^- = \frac{\partial \omega}{\partial p_\perp} \frac{1}{|\mathbf{c}_{g,0}^-|}$ is the direction cosine of the incoming group velocity vector $\mathbf{c}_{g,0}^-$. Projecting the phase space density $\rho_{a,0}(s, p_{\scriptscriptstyle \parallel})$ onto the finite basis as in (5.13) yields the coefficients $\rho_{\mathbf{n},0}$ as follows:

$$\rho_{\mathbf{n},0} = \frac{R}{4\pi} \frac{2m+1}{\sqrt{L_a p_{\text{II},a,l}}} \int_{0}^{L_a} \lambda(p_{\text{II},0}(s,\mathbf{r}_0)) e^{-\mu D(s,\mathbf{r}_0)} P_m\left(\frac{p_{\text{II},0}(s,\mathbf{r}_0)}{p_{\text{II},a,l}}\right) \frac{\cos\alpha^-}{D(s,\mathbf{r}_0)} \mathrm{d}s \,.$$
(5.27)

Here, the additional term $\lambda(p_{\parallel,0})$ is included in the formulation to amount for possible scattering at the edge a.

Computing numerically the integral in Equation (5.27) is not straightforward, since there is an additional step in defining the momentum component $p_{0,0} = p_{0,0}(s, \mathbf{r}_0)$, which is used later to compute the scattering coefficients $\lambda(p_{0,0})$. As mentioned earlier, since the wave number is constant in isotropic structures, the momentum component $p_{0,0}$ is not needed in such cases, and determining the polar angle $\alpha^- = \theta^-$ for a given position *s* is sufficient to define all terms of the integrand in Equation (5.27). While this task can be accomplished using simple geometrical relations in isotropic structures [53], it is not the case for composite structures. One needs to change the variable of integration from *s* to p_0 in Equation (5.27), which will be shown next. As can be seen in Figure 5.4, the following relation between differentials d*s* and d ϕ can be established:

$$\mathrm{d}\phi = \frac{\cos\alpha^{-}}{D(s,\mathbf{r}_{0})}\mathrm{d}s\,,\qquad(5.28)$$



Figure 5.4: Geometrical relations between differentials $d\phi$ and $d\alpha^-$. Rays emitting from the point source at \mathbf{r}_0 intersect edge a at the incoming angles α^- . Similarly as in the case considered in Figure 5.3, rays can only reach edge a if their direction angles $\alpha^- \in [\xi^-, \xi^+]$.

thus allowing us to rewrite the integral in Equation (5.27) in terms of the angle ϕ . The differential $d\phi$ can be connected with dp_{ii} as follows:

$$d\phi = d \arctan\left(\frac{c_{g,y}}{c_{g,x}}\right) = \frac{1}{1 + \left(\frac{c_{g,y}}{c_{g,x}}\right)^2} \left(\frac{\partial\left(\frac{c_{g,y}}{c_{g,x}}\right)}{\partial p_{\scriptscriptstyle \parallel}} + \frac{\partial\left(\frac{c_{g,y}}{c_{g,x}}\right)}{\partial p_{\scriptscriptstyle \perp}}\frac{\partial p_{\scriptscriptstyle \perp}}{\partial p_{\scriptscriptstyle \parallel}}\right) dp_{\scriptscriptstyle \parallel} = \frac{1}{|\mathbf{c}_g|^2} \left(c_{g,x}\frac{\partial c_{g,y}}{\partial p_{\scriptscriptstyle \parallel}} - c_{g,y}\frac{\partial c_{g,x}}{\partial p_{\scriptscriptstyle \parallel}} - \frac{c_{g,x}^2}{c_{g,y}}\frac{\partial c_{g,y}}{\partial p_{\scriptscriptstyle \perp}} + c_{g,x}\frac{\partial c_{g,x}}{\partial p_{\scriptscriptstyle \perp}}\right) dp_{\scriptscriptstyle \parallel},$$
(5.29)

where $\frac{\partial p_{\perp}}{\partial p_{\parallel}}$ is substituted according to Equation (5.6). The group velocity components $c_{g,x}$ and $c_{g,y}$ and their partial derivatives present in the formulation above can be computed from the Hamilton function in (5.1). Finally, after inserting the relations (5.28) and (5.29) into Equation (5.27), one obtains

$$\rho_{\mathbf{n},0} = \frac{R}{4\pi} \frac{2m+1}{\sqrt{L_a p_{\parallel,a,l}}} \int_{p_{\parallel,\min}}^{p_{\parallel,\max}} \lambda(p_{\parallel}) e^{-\mu D(s,\mathbf{r}_0)} P_m\left(\frac{p_{\parallel}}{p_{\parallel,a,l}}\right) f(p_{\parallel}) dp_{\parallel},$$

$$f(p_{\parallel}) = \frac{1}{|\mathbf{c}_g|^2} \left(c_{g,x} \frac{\partial c_{g,y}}{\partial p_{\parallel}} - c_{g,y} \frac{\partial c_{g,x}}{\partial p_{\parallel}} - \frac{c_{g,x}^2}{c_{g,y}} \frac{\partial c_{g,y}}{\partial p_{\perp}} + c_{g,x} \frac{\partial c_{g,x}}{\partial p_{\perp}} \right).$$
(5.30)

The limits of the integral $p_{\parallel,\min}$ and $p_{\parallel,\max}$ are such that the corresponding incoming ray angles α_{\min}^- and α_{\max}^- are equal to ξ^- and ξ^+ , respectively, see Figure 5.4. This is to ensure that rays incoming from the source point \mathbf{r}_0 intersect edge a. Furthermore, the distance $D(s_0, \mathbf{r}_0)$ is equal to $h_{\perp}/\cos \alpha^-$, where h_{\perp} is the height of the triangle connecting the point \mathbf{r}_0 and edge a. The described procedure is applied accordingly for other edges of the mesh cell. The case of a source along edge a can be modelled by the following phase space density

$$\rho_{a,0}(s, p_{\rm H}) = \tilde{R} \cos \alpha^+ \mathbb{1}_a(s) \delta(p_{\rm H} - p_{\rm H,0}), \qquad (5.31)$$

where \tilde{R} is the intensity of the energy flow in the direction angle α^+ [53]. The angle α^+ is defined by the momentum component $p_{0,0}$, as was discussed in subsection 5.2.1. The coefficients $\rho_{\mathbf{n},0}$ after discretisation are then given as

$$\rho_{\mathbf{n},0} = \frac{2m+1}{2\sqrt{L_a p_{\text{II},a,l}}} \tilde{R} \cos \alpha^+ P_m \left(\frac{p_{\text{II},0}}{p_{\text{II},a,l}}\right) \int_0^{L_a} \mathbb{1}_a(s) \mathrm{d}s =$$

$$= \frac{2m+1}{2} \sqrt{\frac{L_a}{p_{\text{II},a,l}}} \tilde{R} \cos \alpha^+ P_m \left(\frac{p_{\text{II},0}}{p_{\text{II},a,l}}\right) .$$
(5.32)

Having discretised the initial 2D phase space density $\rho_{\Gamma,0}$ and the boundary integral operator T, we can compute the stationary or long-time limit 2D phase space density $\rho_{\Gamma,\infty}$. Writing the coefficients $\rho_{\mathbf{n},\infty}$ and $\rho_{\mathbf{n},\mathbf{0}}$ as arrays ρ_{∞} and ρ_{0} allows to establish the discrete version of Equation (2.39) as

$$\varrho_{\infty} = \sum_{k=0}^{\infty} \mathbf{T}^{k} \varrho_{0} \to \varrho_{\infty} = \left(\mathbf{I} - \mathbf{T}\right)^{-1} \varrho_{0}$$
(5.33)

where **T** is the matrix form of the boundary operator T and **I** is the identity matrix. The size of matrices **T** and **I** is $n_a n_m n_l$, where n_a is the number of edges in the meshed structure, n_m is the number of Legendre polynomials used for discretisation of the momentum component p_{\parallel} , and n_l is the number of propagating modes in the structure. In thin composite plates, n_l is equal to 3 - longitudinal, shear and bending modes. For thick composite plates modelled by the WFE method, n_l increases with frequency.



Figure 5.5: Partitioning of the typical wave vector curve (depicted by green ovals) between edges of the mesh element. Red dashed arcs represent $\omega_a^{-1}(\omega_0)$ in Equation (5.36). They are defined by imposing the outgoing group velocity vector angle α^+ to be in range $[\xi^-, \xi^+]$, so that outgoing rays from edge a can reach the point **r**.

5.3.3 Stationary energy and intensity

As a final step, we need to compute the stationary energy density $\rho_{\infty}(\mathbf{r})$ defined by the following relation:

$$\rho_{\infty}(\mathbf{r}) = \int \rho_{\infty}(\mathbf{r}, \mathbf{p}) d\mathbf{p} \,. \tag{5.34}$$

The integrand can be developed further using Equations (5.7) and (2.41) as

$$\rho_{\infty}(\mathbf{r}, \mathbf{p}) = e^{-\mu D(s(\mathbf{r}, \mathbf{p}), \mathbf{r})} \rho_{\Gamma, \infty}(s(\mathbf{r}, \mathbf{p}), p_{\shortparallel}) \delta\left(\omega(p_{\shortparallel}, p_{\perp}) - \omega_0\right) .$$
(5.35)

Substituting this relation into Equation (5.34) and using the *coarea formula* for delta functions [169] yields

$$\rho_{\infty}(\mathbf{r}) = \int_{\omega^{-1}(\omega_0)} \rho_{\Gamma,\infty}(s, p_{\scriptscriptstyle ||}) e^{-\mu D(s, \mathbf{r})} \frac{\mathrm{d}\sigma}{|\mathbf{c}_g|} \,.$$
(5.36)

Here, $\omega^{-1}(\omega_0)$ is all solution pairs $(p_{\shortparallel}, p_{\perp})$ such that $\omega(p_{\shortparallel}, p_{\perp}) = \omega_0$, and $d\sigma$ is the measure on the curve $\omega^{-1}(\omega_0)$. After expanding the 2D phase space



Figure 5.6: Partitioning of the shear wave vector curve (depicted in green) between edges of the mesh element. Red dashed arcs represent $\omega_a^{-1}(\omega_0)$ in Equation (5.36). They are defined by imposing the outgoing group velocity vector angle α^+ to be in range $[\xi^-, \xi^+]$, so that outgoing rays from edge a can reach the point **r**.

density $\rho_{\Gamma,\infty}$ as in Equation (5.13), we obtain

$$\rho_{\infty}(\mathbf{r}) = \sum_{\mathbf{n}} \frac{\rho_{\mathbf{n},\infty}}{\sqrt{L_a p_{\text{II},a,l}}} \int_{\omega^{-1}(\omega_0)} \frac{e^{-\mu D(s,\mathbf{r})}}{|\mathbf{c}_g|} \mathbb{1}_a(s) P_m\left(\frac{p_{\text{II}}}{p_{\text{II},a,l}}\right) d\sigma =$$
$$= \sum_{\mathbf{n}} \frac{\rho_{\mathbf{n},\infty}}{\sqrt{L_a p_{\text{II},a,l}}} \int_{\omega_a^{-1}(\omega_0)} \frac{e^{-\mu D(s,\mathbf{r})}}{|\mathbf{c}_g|} P_m\left(\frac{p_{\text{II}}}{p_{\text{II},a,l}}\right) d\sigma$$
(5.37)

where $\mathbf{n} = (a, m, l)$ includes indices of edges forming the mesh cell containing the point \mathbf{r} . The domain of integration $\omega_a^{-1}(\omega_0)$ is the part of the wave vector curve at a fixed frequency ω_0 that produces rays connecting the point \mathbf{r} and some point s on the edge a. Figure 5.5 presents an illustration of how the wave vector curve (depicted by green ovals) is partitioned as $\omega_a^{-1}(\omega_0)$ (depicted by red dashed arcs) between edges a, a' and a''. It is clear that only rays with angles of the outgoing group velocity vectors $\alpha^+ \in [\xi_-, \xi_+]$ can reach the point \mathbf{r} starting from the edge a. The same applies for edges a' and a''. The wave vector curve considered in Figure 5.5 is divided smoothly into three parts, each corresponding for one edge. However, this is not always the case. In fact, for the propagating shear mode of a composite laminate considered in subsection 3.3.2, the partition is not smooth, see Figure 5.6. Note that in the local coordinate systems of edges a' and a'', there exist values of p_{\parallel} for which there are two outgoing shear waves. This feature was previously highlighted in Figures 3.6 and 3.8.

Once the partitioned parts of the wave vector curve are determined, we can change the variable of integration from $d\sigma$ to dp_{\parallel} via the following relation:

$$\mathrm{d}\sigma = \sqrt{\mathrm{d}p_{\scriptscriptstyle \parallel}^2 + \mathrm{d}p_{\perp}^2} = \sqrt{1 + \left(\frac{\partial p_{\perp}}{\partial p_{\scriptscriptstyle \parallel}}\right)^2} \mathrm{d}p_{\scriptscriptstyle \parallel} = \sqrt{1 + \left(\frac{c_{g,x}}{c_{g,y}}\right)^2} \mathrm{d}p_{\scriptscriptstyle \parallel} = \frac{\mathrm{d}p_{\scriptscriptstyle \parallel}}{\cos\alpha} \,, \ (5.38)$$

where $\frac{\partial p_{\perp}}{\partial p_{\parallel}}$ is defined by Equation (5.6). The clockwise orientation of the curve $\omega^{-1}(\omega_0)$ in the coordinate system of the mesh cell x_g, y_g is consistent with the variation of the momentum component p_{\parallel} in the respective local coordinate systems of edges. After changing the variable of integration in Equation (5.36), we obtain:

$$\rho_{\infty}(\mathbf{r}) = \sum_{\mathbf{n}} \frac{\rho_{\mathbf{n},\infty}}{\sqrt{L_a p_{\text{II},a,l}}} \int_{\Omega_{a,p_{\text{II}}}} \frac{e^{-\mu D(s,\mathbf{r})}}{|\mathbf{c}_g| \cos \alpha^+} P_m\left(\frac{p_{\text{II}}}{p_{\text{II},a,l}}\right) \mathrm{d}p_{\text{II}}, \qquad (5.39)$$

where $\Omega_{a,p_{\parallel}}$ covers all momentum components p_{\parallel} such that the momentum vectors **p** produce outgoing waves in the local coordinate system e_x, e_y of the edge a and $\omega_a(p_{\parallel}, p_{\perp}) = \omega_0$. In the case of an oval-shaped wave vector curve considered in Figure 5.5, $\Omega_{a,p_{\parallel}}$ is equal to $[p_{\parallel,\min}, p_{\parallel,\max}]$ with $p_{\parallel,\min(\max)}$ being the momentum component p_{\parallel} such that the outgoing angle $\alpha^+ = \xi_{-(+)}$. Finally, to compute the total energy density at a point **r** in the interior of the same polygon as the source point \mathbf{r}_0 , we need to include the contribution of the initial energy density at that point. Technically, the total energy density $\epsilon(\mathbf{r})$ can be written as

$$\epsilon(\mathbf{r}) = \rho_{\infty}(\mathbf{r}) + \mathbb{1}_{\mathbf{r}_0}(\mathbf{r}) \frac{R}{|\mathbf{c}_{g,0}|} \frac{e^{-\mu|\mathbf{r}-\mathbf{r}_0|}}{2\pi|\mathbf{r}-\mathbf{r}_0|}, \qquad (5.40)$$

where $\mathbb{1}_{\mathbf{r}_0}(\mathbf{r}) = 1$ if the points \mathbf{r} and \mathbf{r}_0 are in the interior of the same mesh cell, and 0 otherwise.

One can also compute the intensity vector field $I(\mathbf{r})$ which is defined as

$$I(\mathbf{r}) = \int \rho_{\infty}(\mathbf{r}, \mathbf{p}) \mathbf{c}_{\mathbf{g}} d\mathbf{p}.$$
 (5.41)

After proceeding similarly to the case of the stationary energy density, one obtains:

$$\mathbf{I}(\mathbf{r}) = \sum_{\mathbf{n}} \frac{\rho_{\mathbf{n},\infty}}{\sqrt{L_a p_{||,a,l}}} \int_{\Omega_{a,p_{||}}} \left[\tan \alpha^+ \right] e^{-\mu D(s,\mathbf{r})} P_m\left(\frac{p_{||}}{p_{||,a,l}}\right) \mathrm{d}p_{||}.$$
(5.42)

The intensity vector field $\mathbf{I}(\mathbf{r})$ allows estimating the direction of mean energy flow at the point \mathbf{r} , which can be helpful to determine energy transfer paths.

5.4 Numerical results

In this section, we consider several numerical case examples to demonstrate the application of the DEA method, both the classical method for isotropic structures and the modified version presented in this chapter for composite structures. In all examples we compute the mean acceleration, which can be estimated as

$$\bar{a}(\mathbf{r}) = \sqrt{\epsilon(\mathbf{r}) \frac{\omega^2}{\rho_{\text{vol}}(\mathbf{r})h(\mathbf{r})}},$$
(5.43)

where $\epsilon(\mathbf{r})$ is the total mean energy density given by Equation (5.40), $\rho_{\text{vol}}(\mathbf{r})$ and $h(\mathbf{r})$ are the volumetric mass density and the thickness of the polygon containing the point \mathbf{r} , respectively.

This section is organised as follows. In subsection 5.4.1, we compute the mean response of the polygonally shaped plate due to a point and an edge sources. Isotropic and composite materials are considered to demonstrate how the wave energy is transported and scattered in each of these cases. Also, the influence of the number of Legendre polynomials on the results is investigated. Then, in subsection 5.4.2, we compute and compare the DEA and FEM frequencyaveraged responses of an L-shaped angle-ply laminated plate subjected to a harmonic point excitation force.



Figure 5.7: A polygonally shaped semi-infinite plate meshed with two-dimensional triangular shell elements. An edge and a point sources are assumed to be applied on the plate on the vertical and horizontal edges, respectively. The diagonal edge is free, and rays hitting this edge reflect back into the plate. Reflection of rays at other edges is prohibited thus imitating "infinity" of the plate through these edges.

Finally, in subsection 5.4.3, we turn to the case of an electric vehicle gearbox, where we compare the results from the classical DEA and the full FEM simulation. We present a limitation of the classical DEA approach on this example and demonstrate how the composite DEA can provide more accurate results. The work presented in this subsection was performed during the industrial placement at "Romax Technology" company.

5.4.1 Test case for validation

We consider a semi-infinite plate with one free edge as shown in Figure 5.7. The mesh consists of 1216 points and 2300 triangular shell elements. The plate thickness is equal to 5 mm. The horizontal edges of the plate are 20 and 80 cm in length, the same applies for the vertical edges. The diagonal edge of the plate is inclined at 45° to the vertical axis and equal to 84.85 cm in length. The "infinity" condition at the straight edges is achieved by prohibiting reflection



Figure 5.8: A geometrical illustration of the ray paths (shown in red) starting from the edge source at the vertical edge. Wave vector curves (shown in green) are a circle for the isotropic material case (a) and an oval for the composite material case (b).

	Aluminium	Stainless Steel
$\overline{\text{Young's Modulus (N/m^2)}}$	71×10^9	210×10^9
Poisson's ratio	0.33	0.3
Density (kg/m^3)	2740	7800

Table 5.1: Engineering constants of aluminium and stainless steel materials.

of rays hitting these edges. The correspondent elements of the matrix **T** in (5.33) are equal to zero. This setup allows us to work with a small geometrical mesh model and consider local energy scattering events without diffusive type mixing of infinite number of reflections that would occur in a finite model. First, we assume an edge source of strength R = 1 at the vertical edge connecting points with coordinates (0, 48), (0, 50). Other parameters of the initial energy density given by Equation (5.31) are l = B, $\eta = 0$, $\omega = 3000$ Hz and $p_{\parallel,0} = 0$, that is, the bending mode is excited at the edge of the undamped plate, and the momentum vector **p** points in the direction normal to the edge. It is important to emphasise that the stationary energy density solution given by the infinite sum $\sum_{k=0}^{\infty} T^k \rho_{\Gamma,0}$ is still converging in the absence of damping because of artificial "infinity" condition at the vertical and horizontal edges. We consider two plate materials: the first one is isotropic stainless steel, and



Figure 5.9: The mean acceleration response levels of the polygonally shaped semiinfinite isotropic plate due to an edge source as a function of the number of Legendre polynomials N_m . The acceleration is given in m/s^2 .

the second one is a composite regular cross-ply laminate with the lamination scheme $0^{\circ}/90^{\circ}/0^{\circ}/90^{\circ}/0^{\circ}$, considered in-depth in Sections 3.7 and 4.5. The material parameters of stainless steel and the composite laminae are given in Tables 5.1 and 3.1, respectively.

The schematic illustrations presented in Figure 5.8 can give an idea of how initial energy might distribute over the structure in isotropic and composite material cases.

In the case of isotropic material, the Hamilton function fixes the amplitude of the momentum vector $|\mathbf{p}|$; therefore, the curve of possible momentum values has the form of a circle, see Figure 5.8a. Furthermore, the wave energy travels along the momentum vector since the group velocity and momentum vectors are collinear and co-directed. Rays emitting from the vertical edge reflect from the free diagonal edge at the right angle and travel until hitting and leaving the horizontal edge due to the "infinity" condition imposed. Reflection at the right angle happens because the momentum components p_{\parallel} and p'_{\parallel} must be equal due to continuity condition, see subsection 3.3.1. In isotropic structures, this condition yields *Snell's law* stating that the angles of incidence and reflection must be equal.

In the case of composite material, the Hamilton function does not fix the amplitude of the momentum vector, and the wave energy travels along the direction of the group velocity vector. The bending wave vector curve has the form of an oval, see Figure 5.8b. Upon reaching the diagonal edge, rays reflect in the direction of the outgoing group velocity vector \mathbf{c}_g^+ and which is derived from the continuity condition $p_{\parallel} = p'_{\parallel}$. The angles of incidence and reflection are not equal in this case.

Figures 5.9 and 5.10 present the mean acceleration response amplitudes of the isotropic and composite plate due to an edge source as a function of the number of Legendre polynomials used in discretisation of the momentum component $p_{\rm n}$. As mentioned at the end of subsection 5.3.1, when only the first Legendre polynomial $P_0(x) = 1$ is used, the DEA method works just like the SEA method. The wave energy field is diffusive and consists of a superposition of rays travelling uniformly in all directions, see Figures 5.9a and 5.10a for the isotropic and composite material cases, respectively. As the number of Legendre polynomials used in the basis set increases, the vibrational response of the structure changes so that the initial energy density at the source edge propagates in a less diffusive way through the structure. At $N_m = 25$, one can notice a specular type of reflection of the energy density at the free edge, following the geometrical ray paths described in Figure 5.8. Therefore, as the number of Legendre polynomials increases, the DEA method works like a full ray-tracing method [50, 51].

One can compute the intensity vector field $I(\mathbf{r})$ using Equation (5.42). Fig-



Figure 5.10: The mean acceleration response levels of the polygonally shaped semiinfinite composite plate due to an edge source as a function of the number of Legendre polynomials N_m .

ure 5.11 presents the intensity vector field in the isotropic and composite material cases and how this field varies as a function of the number of Legendre polynomials. In the diffuse field case, that is, when $N_m = 1$, one can clearly see that the wave energy is transported in all directions, see Figures 5.11a and 5.11b for the isotropic and composite material cases. Nevertheless, in the lower part of the plate, the intensity vectors are nearly normal to the horizontal edge in the isotropic material case and directed towards the left corner in the composite material case, visually representing that the group velocity vector field governs the wave energy paths. In the full ray-tracing approximation, that is, when $N_m = 25$, the intensity vector field is consistent with the mean acceleration distribution over the structure, see Figures 5.11c and 5.11d.



Figure 5.11: The intensity vector field of the isotropic (a,c) and composite (b,d) plate cases arising due to an edge source at the vertical edge as a function of the number of Legendre polynomials used in the basis set.

Now, we consider a point source of strength R = 1 at the point (0.4, 0) on the horizontal edge, and the bending mode is excited at this point. In experimental setups, this is usually achieved by using the shaker excitation method. Even though one might argue that the shaker cannot be physically applied on the edge of the structure, it is reminded that the structure is assumed to be infinite through the vertical and horizontal edges by imposing no reflection of the wave energy on them.

Figures 5.12 and 5.13 represent the mean acceleration response levels arising due to the point source at the horizontal edge in the isotropic and composite



Figure 5.12: The mean acceleration response levels of the polygonally shaped semiinfinite isotropic plate due to an edge source as a function of the number of Legendre polynomials N_m . The acceleration is given in m/s^2 .

material cases, respectively. As the number of Legendre polynomials used to represent the momentum space increases, the vibrational response of the plate becomes more diffusive; however, one can still see individual ray stripes reflecting from the free edge in Figures 5.12c and 5.13c. Nevertheless, it appears that using four Legendre polynomials in the basis set is sufficient to adequately predict the vibrational response of the structure to a point source. Finally, as discussed in subsection 3.7.1, the bending wave energy is fully reflected at the free edge without mode coupling at the frequency considered; therefore, the wave energy is not stored in in-plane modes.



Figure 5.13: The mean acceleration response levels of the polygonally shaped semiinfinite isotropic plate due to an edge source as a function of the number of Legendre polynomials N_m . The acceleration is given in m/s^2 .

5.4.2 Two composite plates joined at a right angle

In this subsection, we compute the mean acceleration response levels of an L-shaped composite plate due to a harmonic point excitation force at the middle of the ground plate. The geometrical configuration of the model is shown in Figure 5.14. The constitutive plates are regular angle-ply laminates with the lamination scheme $45^{\circ}/-45^{\circ}/45^{\circ}/-45^{\circ}/45^{\circ}$ and the total thickness of 5 mm. Hysteretic damping of 5% is assumed.

Figure 5.15 shows a comparison of mean acceleration levels obtained from DEA and frequency-averaged FEM simulations at the frequency f = 3150 Hz. Four Legendre polynomials were used in the basis set to discretise the momentum space. DEA acceleration amplitudes are averaged between three propagating modes in the structure: bending, shear, and longitudinal modes. In Fig-



Figure 5.14: An L-shaped composite plate meshed with two-dimensional triangular shell elements. A point source is assumed to be applied on the middle of the ground plate.

ures 5.15c and 5.15d, lines representing high acceleration levels (coloured in red and yellow) are inclined at 45° and -45° with respect to the global x axis. This is because the wave vector curves are 45° rotated with respect to the global x axis, see Figures 3.7 and 3.8. The wave energy scatters from corners of the ground plate and propagates into the L-plate. The overall DEA and FEM acceleration distributions agree well, despite the presence of phase shifts at the wavelength scale in the FEM result, see Figures 5.15a and 5.15b.

The direction of energy flow can be found out by computing the intensity vector field. In contrast to the polygonally shaped plate case, mode coupling occurs at the junction between plates; therefore, we can visualise the energy flow of all modes. Figure 5.16a presents the intensity vector field of the bending mode. Note that energy is transported in all directions as expected. However, the highest intensity values in the ground plate occur along directions of 45° and -45° with respect to the global x axis. The intensity field of the



(c) DEA, top view

(d) FEM, top view

Figure 5.15: The mean DEA and FEM acceleration response levels of an L-shaped composite plate due to a harmonic point excitation force of 1 N at the frequency f = 3150 Hz. FEM results are frequency-averaged over one-third octave band with the centre frequency f = 3150 Hz. The acceleration is given in m/s^2 .

bending mode is in accordance with the mean acceleration distribution in the L-plate since the bending mode mainly governs the energy transport, whereas the contribution from in-plane modes is small. Nevertheless, one can note how the energy stored in the longitudinal mode propagates through the structure, see Figure 5.16b. The most significant intensity values arise at the corner area of the junction between plates. This is consistent with the intensity distribution of the bending mode, having the highest values at the correspondent corner area at the ground plate.



Figure 5.16: The intensity vector field of an L-shaped composite plate arising due to a point source at the middle of the ground plate. Vector colouring represents intensity amplitudes.



Figure 5.17: The FE meshes of the full gearbox housing (a) and the gearbox housing without stiffeners along the longitudinal axis of the cylindrical part (b).

5.4.3 Electric Vehicle Gearbox

This subsection presents part of the work performed during an industrial placement at Romax Technology. A gearbox housing of the electric vehicle powertrain is considered - its geometrical mesh model is shown in Figure 5.17a. The structure consists of a lower stiffened cylindrical part and an upper nonstiffened part. The FE model contains 96672 nodes and 48713 triangular shell elements of size $\sim 3 \text{ mm}$. The structure is made of isotropic aluminium with material parameters given in Table 5.1. The volume of the structure is 6176.3 cm³, and its mass is 17 kg. The goal of the work was to perform a modal frequency response analysis of the structure subjected to a harmonic



Figure 5.18: The mean DEA and FEM acceleration response levels of the gearbox housing of an electric vehicle powertrain due to a harmonic point excitation force of 1 N at the frequency f = 8000 Hz. The acceleration is given in mm/s^2 .

point excitation force and compare results obtained from the FEM and DEA methods for frequencies between 5 kHz and 12.5 kHz. However, for the sake of brevity, results will be shown only for the frequency f = 8000 Hz. The FEM results are frequency-averaged over one third octave bands, whereas DEA results are computed at the centre frequencies of these bands. The structure is meshed in the FEM software "Hypermesh", and the FEM simulations were performed using the software "Optistruct". A harmonic point excitation force of 1 N is applied at the front side of the structure in the positive Z direction, and hysteretic damping of 0.5% is assumed.

Figure 5.18 presents DEA and FEM acceleration response amplitudes of the gearbox housing at the centre frequency f = 8000 Hz. It is noted that the DEA results agree reasonably well with the FEM results across the upper non-



Figure 5.19: The intensity vector field of the gearbox housing with stiffeners (a) and without stiffeners (b) arising due to a point source at its front side. The structure colouring represents the mean acceleration levels of the bending mode.

cylindrical part of the structure. However, it appears that the DEA acceleration levels decrease faster than the correspondent FEM levels in the stiffened part of the structure. During the project, it was found that this discrepancy is connected to the limited applicability of the DEA method at frequencies, where the correspondent wavelengths are of the same order as characteristic lengths of the structure. In fact, the average length between stiffeners along cylinder's longitudinal axis is only 10 mm, and the bending wavelength is $\sim 104 \text{ mm}$ at the frequency f = 8000 Hz. Consequently, the individual stiffeners are not resolved on the wavelength scale. In contrast, in DEA, rays are assumed to interact with each stiffener via scattering events, thus enormously increasing their overall path lengths D(X, X'). Therefore, the overall damping term $\exp(-\mu D(X, X'))$ in Equation (5.3) is overestimated, thus decreasing the acceleration levels at higher rate compared to the FEM results. Figure 5.19a presents the intensity vector field obtained from the DEA method at the frequency f = 8000 Hz, and one can see that the wave energy scatters at stiffeners of the cylindrical part.

The following steps were performed to resolve this problem:

Firstly, stiffeners along the longitudinal axis of the cylindrical part were re-


(a) effective density



(b) effective thickness

Figure 5.20: Two modification methods of the FE model of the non-stiffened structure to obtain the same structural mass as the full structure. The first one (a) is to compute and assign an effective value of material density to the cylindrical part 2, which is different from 1, where the structure is made of aluminium. The second one (b) is to compute and assign effective thickness of values of shell elements in parts 1 and 2 by calculating the total volume of removed stiffeners and keeping the same material density.

moved from the model, and new FEM and DEA simulations were performed on the non-stiffened gearbox housing. Figure 5.17b presents its FE model, and Figure 5.19b shows the intensity vector field produced by the point source. Secondly, since DEA results of the non-stiffened structure cannot be compared with the FEM results of the initial structure due to the mass difference of 2.9 kg, further modifications of the model were performed. Namely, to obtain the same structural mass, an effective value of material density was computed and assigned to the non-stiffened cylindrical part, see Figure 5.20a. In this



Figure 5.21: The nodes of the FE mesh, where the acceleration response amplitudes were computed. Input node and nodes 1, 2 and 3 are located on the non-stiffened part, whereas nodes 4 and 5 are on the stiffened part of the structure.

figure, finite elements on the part labelled 1 are made of aluminium as before. On the other hand, elements on the part labelled 2 have a new value of material density 4032.55 kg/m^3 .

The same mass was also achieved by a different method. Namely, effective thickness values were assigned to elements on the non-stiffened cylindrical part by calculating the total volume of removed stiffeners and keeping the same material density, see Figure 5.20b. In this figure, shell elements on the part, labelled 1, have thickness value 8.86 mm - before removing stiffeners, and 15.51 mm - after. Elements on the part, labelled 2, have thickness value 10.04 mm - before removing stiffeners, and 15.76 mm - after.

Finally, another method to resolve the issue of artificially damped acceleration levels is to compute and assign the effective orthotropic material parameters to the cylindrical part based on its geometry. These parameters can be computed using the formula presented, for example, in [206] and in Figure 3.28 of [142]. Following this approach, one can obtain results using a composite DEA method presented in this chapter. However, this task was not accomplished due to time restrictions and is regarded as further work to be done outside of the scope of the current thesis.



Figure 5.22: Comparison of mean acceleration amplitudes (given in dB), calculated with different methods on the node positions, described in Figure 5.21, at the frequency f = 8000 Hz.

Several nodes on the structure are chosen for the point-by-point data comparison, and they are described in Figure 5.21. Note that the input and first three nodes are located on the non-cylindrical part of the structure, whereas nodes 4 and 5 are on the cylindrical part. A point-by-point comparison of different DEA results with the FEM ones at the frequency f = 8000 Hz can be found in Figure 5.22. The accelerations are given in the dB scale. We can note that the acceleration levels at the input and first three nodes agree well for all different cases. This is because these nodes are located on the unchanged non-cylindrical part of the structure. The acceleration levels obtained from DEA on the complete structure (denoted in dark blue) are much lower than in other cases at node 5. This discrepancy was discussed earlier, and it was also seen in Figure 5.18c. The mass smearing approaches do not appear to change much in acceleration levels at nodes 4 and 5.

5.5 Conclusion

This chapter has considered the extension and implementation of the DEA method for composite structures that can be meshed by two-dimensional shell elements. This has been accomplished for the first time in this work, and we have used the findings of Chapters 3 and 4 to derive the stationary wave energy density arising in the structure due to a point or an edge source. To achieve this, first, we have discussed how the ray tracing formula that governs the trajectories of energy rays is modified in the case of composite laminated plates. The boundary integral operator T has been discretised on the triangular FE meshes, using Legendre polynomials. We have shown how the expressions for initial 2D phase space densities need to be changed in composite structures. Solving the linear system consisting of the initial 2D phase space density and the boundary integral operator yields the stationary 2D phase space density, which is used to compute the stationary wave energy density at any point of the structure.

Finally, we have considered several numerical case studies such as a polygonally shaped plate, an L-shaped composite plate and an electric vehicle gearbox. In the first case, we have computed the mean response of the polygonally shaped plate due to a point and an edge sources. Isotropic and composite materials have been considered to demonstrate how the wave energy is transported and scattered in each of these cases. Also, the influence of the number of Legendre polynomials on the results has been investigated.

In the second case, we have computed and compared the DEA and FEM frequency-averaged responses of an L-shaped angle-ply laminated plate subjected to a harmonic point excitation force. Also, we have computed the intensity vector field of the bending and longitudinal modes.

In the last case, we have computed and compared the results from the classical DEA and the full FEM simulation for an electric gearbox housing. We have presented a limitation of the classical DEA approach on this example and demonstrated how the composite DEA could provide more accurate results. This work has been performed during the industrial placement at "Romax Technology" company.

Chapter 6 Conclusion

In this chapter, a brief overview of the main findings of the thesis is provided. Furthermore, we discuss further research that could serve as a basis for the extension of the work.

6.1 Main Contributions

In this section, we briefly review the main goals and contributions of the thesis. We have been interested in the problem of high-frequency wave energy scattering and propagation in composite structures. In particular, structures made of the two-dimensional plate- or shell-like elements with composite laminate material were considered. We wanted to provide an extension of the DEA method to such structures.

Chapter 2 aimed to present the main theoretical basics of the DEA method for two-dimensional isotropic structures. Furthermore, the solution of Helmholtz and biharmonic wave equations have been presented. Finally, we have identified and discussed the main modifications needed to allow for DEA application on composite structures.

Chapter 3 has described in all generality how to compute energy scattering coefficients of structural junctions made up of thin composite laminated plates in the line junction approximation. Expressions quantifying transmission and reflection coefficients as a function of the frequency and the wave number component k_x have been derived. Interesting phenomena such as negative refraction and negative group velocity have been observed and analysed. As a final step, an effective scattering matrix for a plate with multiple finite stiffeners attached to it has been derived and computed for the cases of one and four stiffeners. The scattering coefficients have been computed explicitly for examples of two and three composite plates joined together in an L and T geometry.

A hybrid FE/WFE model that predicts the scattering properties for different junctions of two-dimensional anisotropic composite plates has been developed in Chapter 4. The influence of the angle of incidence and the frequency on the distribution of the power flow of incident bending, shear and longitudinal type waves has been investigated. A detailed comparison with semi-analytical evaluations of scattering coefficients derived in Chapter 3 has been presented. The method gives for the first time a detailed recipe for computing scattering coefficients for the generic case of an arbitrary number of composite plates connected at a junction without restrictions on the angles at which the plate meet or the orientation of the principal axis of individual plates.

Finally, in Chapter 5, we have presented the modified theoretical base of DEA for composite structures. The findings of Chapters 3 and 4 have been used to derive the stationary wave energy density arising in the structure due to a point or an edge sources. Numerical results for the cases of a polygonally shaped plate, an electric vehicle gearbox and an L-shaped composite plate has been presented.

6.2 Further work

The current thesis has provided a solid ground for further research regarding wave energy scattering and propagation in composite structures. Several possible continuation works have been identified. These include

• The thin plate assumption used in theory presented in Chapter 3 can be relaxed, and the correspondent governing equations can be derived from the First- or Third-Order laminated theories [67]. This would allow us to consider relatively thick composite plates in the derivation of energy scattering coefficients and to validate the WFE method results for such structures. Also, one can extend the curved shell theory presented, for example, in [98, 138] and compute energy scattering coefficients of composite shell-plate junctions.

- The work presented in Section 3.6 has considered the case of the stiffened plate. Following the approach presented in [194, 195, 207], one can extend this work to the case of periodic grillages consisting of composite beams and plates. This would allow us to compute effective wave vector curves of such metamaterials for the whole frequency domain, not only for high frequencies. In structures consisting of such metamaterials, one can compute effective scattering coefficients at their junction parts.
- The WFE method presented in Chapter 4 can be extended for curved composite structures and their junctions with composite plates. This can be easily done by following the approach presented in [208, 209].
- The extension of the DEA method for composite structures presented in Chapter 5 is based on the same assumptions as of the classical DEA method. In particular, the wavelengths of propagating modes must be shorter than the characteristic dimensions of the structure. We have seen the implication of breaking this assumption in the case of an electric vehicle gearbox considered in subsection 5.4.3. In the low-frequency regime, stiffened plates can be regarded as orthotropic plates, allowing application of the presented composite DEA method.
- The other limitation of the DEA method is that the curved parts of the structure are considered as the set of plate-like elements connected in the mesh. Energy rays follow not straight, but rather geodesic lines on curved structures [175, 210]. This limitation can be circumvented by allowing

the momentum vector \mathbf{p} to be dependent on the curved geometry, thus complexifying the Hamilton equations (5.2).

• Finally, it is assumed that energy rays scatter in a specular manner at discontinuities of a structure. This restriction can be relaxed by representing the kernel of the phase-space density propagating operator T as a sum of specular and diffusive components [211]. This would allow accounting for irregular or random surfaces.

Bibliography

- R. M. Jones, *Mechanics of composite materials*, 2nd ed. Washington, D.C.: Taylor & Francis, 1999.
- [2] A. P. Mouritz, E. Gellert, P. Burchill, and K. Challis, "Review of advanced composite structures for naval ships and submarines," pp. 21–42, 2001.
- [3] Z. Hashin, "Analysis of composite materials: A survey," J. Appl. Mech. Trans. ASME, vol. 50, no. 3, pp. 481–505, 1983.
- [4] T. J. Hertz, M. H. Shirk, R. H. Ricketts, and T. A. Weisshaar, "On the track of practical forward-swept wings." *Astronaut. Aeronaut. New York, N.Y.*, vol. 20, no. 1, pp. 40–52, 1982.
- [5] O. C. Zienkiewicz, R. L. Taylor, and D. Fox, The Finite Element Method for Solid and Structural Mechanics. Elsevier, 2014.
- [6] A. Iserles, A First Course in the Numerical Analysis of Differential Equations. Cambridge: Cambridge University Press, 2008.
- [7] K. K. Gupta and J. L. Meek, "A brief history of the beginning of the finite element method," Int. J. Numer. Methods Eng., vol. 39, no. 22, pp. 3761–3774, 1996.
- [8] N. Perrone and R. Kao, "A general finite difference method for arbitrary meshes," *Comput. Struct.*, vol. 5, no. 1, pp. 45–57, 1975.
- [9] T. Liszka and J. Orkisz, "The finite difference method at arbitrary irregular grids and its application in applied mechanics," *Comput. Struct.*, vol. 11, no. 1-2, pp. 83–95, 1980.
- [10] A. H. Cheng and D. T. Cheng, "Heritage and early history of the boundary element method," *Eng. Anal. Bound. Elem.*, vol. 29, no. 3, pp. 268– 302, 2005.
- [11] M. A. Jaswon, G. T. Symm, and T. A. Cruse, "Integral equation methods in potential theory and elastostatics," *J. Appl. Mech.*, vol. 45, no. 4, pp. 966–966, dec 1978.
- [12] C. A. Brebbia, J. C. F. Telles, L. C. Wrobel, and S. Mukherjee, "Boundary element techniques. theory and applications in engineering," J. Appl. Mech., vol. 52, no. 1, pp. 241–241, 1985.

- [13] R. H. Lyon, R. G. DeJong, and M. Heckl, Theory and Application of Statistical Energy Analysis. Elsevier, 1995.
- [14] R. J. Craik, "The prediction of sound transmission through buildings using statistical energy analysis," J. Sound Vib., vol. 82, no. 4, pp. 505– 516, 1982.
- [15] A. J. Keane and W. G. Price, "Statistical energy analysis of strongly coupled systems," J. Sound Vib., vol. 117, no. 2, pp. 363–386, 1987.
- [16] D. Wilson and C. Hopkins, "Analysis of bending wave transmission using beam tracing with advanced statistical energy analysis for periodic boxlike structures affected by spatial filtering," J. Sound Vib., vol. 341, pp. 138–161, apr 2015.
- [17] A. Le Bot and V. Cotoni, "Validity diagrams of statistical energy analysis," J. Sound Vib., vol. 329, no. 2, pp. 221–235, 2010.
- [18] N. Totaro and J. L. Guyader, "Sea substructuring using cluster analysis: The mir index," J. Sound Vib., vol. 290, no. 1-2, pp. 264–289, 2006.
- [19] I. Bosmans, G. Vermeir, and P. Mees, "Coupling loss factors for coupled anisotropic plates," J. Sound Vib., vol. 250, no. 2, pp. 351–355, 2002.
- [20] E. Glushkov, N. Glushkova, A. Eremin, R. Lammering, and M. Neumann, "Frequency dependent directivity of guided waves excited by circular transducers in anisotropic composite plates," J. Acoust. Soc. Am., vol. 132, no. 2, pp. EL119–EL124, 2012.
- [21] E. Glushkov, N. Glushkova, A. Eremin, and R. Lammering, "Group velocity of cylindrical guided waves in anisotropic laminate composites," *J. Acoust. Soc. Am.*, vol. 135, no. 1, pp. 148–154, 2014.
- [22] E. A. Magliula, J. G. McDaniel, and A. D. Pierce, "Far-field approximation for a point-excited anisotropic plate," J. Acoust. Soc. Am., vol. 131, no. 6, pp. 4535–4542, nov 2012.
- [23] F. Fahy, "Statistical energy analysis: a critical overview," Philos. Trans. R. Soc. London. Ser. A Phys. Eng. Sci., vol. 346, no. 1681, pp. 431–447, mar 1994.
- [24] R. S. Langley, "A wave intensity technique for the analysis of high frequency vibrations," J. Sound Vib., vol. 159, no. 3, pp. 483–502, 1992.
- [25] R. S. Langley and A. N. Bercin, "Wave intensity analysis of high frequency vibrations," *Philos. Trans. R. Soc. London. Ser. A Phys. Eng. Sci.*, vol. 346, no. 1681, pp. 489–499, 1994.
- [26] V. D. Belov, S. A. Rybak, and B. D. Tartakovskii, "Propagation of vibrational energy in absorbing structures." Sov Phys Acoust, vol. 23, no. 2, pp. 115–119, 1977.

- [27] M. N. Ichchou, A. Le Bot, and L. Jezequel, "Energy models of onedimensional, multi-propagative systems," J. Sound Vib., vol. 201, no. 5, pp. 535–554, 1997.
- [28] Y. H. Park, "Energy flow finite element analysis of general mindlin plate structures coupled at arbitrary angles," Int. J. Nav. Archit. Ocean Eng., vol. 11, no. 1, pp. 435–447, 2019.
- [29] Y. H. Park and S. Y. Hong, "Vibrational energy flow analysis of corrected flexural waves in timoshenko beam - part i: Theory of an energetic model," *Shock Vib.*, vol. 13, no. 3, pp. 137–165, 2006.
- [30] N. Vlahopoulos, L. O. Garza-Rios, and C. Mollo, "Numerical implementation, validation, and marine applications of an energy finite element formulation," J. Sh. Res., vol. 43, no. 3, pp. 143–156, 1999.
- [31] H. Kuttruff, Room acoustics. Crc Press, 2016.
- [32] A. Le Bot, "A vibroacoustic model for high frequency analysis," J. Sound Vib., vol. 211, no. 4, pp. 537–554, 1998.
- [33] A. Le Bot and E. Sadoulet-Reboul, "High frequency vibroacoustics: A radiative transfer equation and radiosity based approach," *Wave Motion*, vol. 51, no. 4, pp. 598–605, 2014.
- [34] G. I. Koutsouris, J. Brunskog, C.-h. Jeong, and F. Jacobsen, "Combination of acoustical radiosity and the image source method," J. Acoust. Soc. Am., vol. 133, no. 6, pp. 3963–3974, jun 2013.
- [35] R. S. Langley, "On the vibrational conductivity approach to high frequency dynamics for two-dimensional structural components," pp. 637– 657, 1995.
- [36] M. J. Smith, "A hybrid energy method for predicting high frequency vibrational response of point-loaded plates," J. Sound Vib., vol. 202, no. 3, pp. 375–394, 1997.
- [37] X. Kong, H. Chen, D. Zhu, and W. Zhang, "Study on the validity region of energy finite element analysis," J. Sound Vib., vol. 333, no. 9, pp. 2601–2616, 2014.
- [38] A. Le Bot, "Comparison of vibrational conductivity and radiative energy transfer methods," J. Sound Vib., vol. 283, no. 1-2, pp. 135–151, 2005.
- [39] —, "Energy transfer for high frequencies in built-up structures," J. Sound Vib., vol. 250, no. 2, pp. 247–275, 2002.
- [40] A. Carcaterra and A. Sestieri, "Energy density equations and power flow in structures," J. Sound Vib., vol. 188, no. 2, pp. 269–282, 1995.
- [41] N. Vlahopoulos, N. Schiller, and S. Lee, "Energy finite element analysis developments for vibration analysis of composite aircraft structures," *SAE Int. J. Aerosp.*, vol. 4, no. 2, pp. 593–601, nov 2011.

- [42] V. Cerveny and M. G. Brown, "Seismic ray theory," J. Acoust. Soc. Am., vol. 113, no. 1, pp. 14–14, jan 2003.
- [43] K. S. Chae and J. G. Ih, "Prediction of vibrational energy distribution in the thin plate at high-frequency bands by using the ray tracing method," *J. Sound Vib.*, vol. 240, no. 2, pp. 263–292, 2001.
- [44] D. C. Anderson and J. Cychosz, "An introduction to ray tracing," Image Vis. Comput., vol. 8, no. 2, p. 169, 1990.
- [45] O. Runborg, "Mathematical models and numerical methods for high frequency waves," *Commun. Comput. Phys.*, vol. 2, no. 5, pp. 827–880, 2007.
- [46] V. Cerveny and I. Psencik, "Gaussian beams in inhomogeneous anisotropic layered structures," *Geophys. J. Int.*, vol. 180, no. 2, pp. 798–812, feb 2010.
- [47] L. Savioja and U. P. Svensson, "Overview of geometrical room acoustic modeling techniques," J. Acoust. Soc. Am., vol. 138, no. 2, pp. 708–730, aug 2015.
- [48] E. Reboul, A. Le Bot, and J. Perret-Liaudet, "Radiative transfer equation for multiple diffraction," J. Acoust. Soc. Am., vol. 118, no. 3, pp. 1326–1334, sep 2005.
- [49] A. Le Bot, "A functional equation for the specular reflection of rays," J. Acoust. Soc. Am., vol. 112, no. 4, pp. 1276–1287, 2002.
- [50] G. Tanner, "Dynamical energy analysis—determining wave energy distributions in vibro-acoustical structures in the high-frequency regime," J. Sound Vib., vol. 320, no. 4-5, pp. 1023–1038, mar 2009.
- [51] D. J. Chappell and G. Tanner, "Solving the stationary liouville equation via a boundary element method," J. Comput. Phys., vol. 234, no. 1, pp. 487–498, feb 2013.
- [52] G. Tanner and S. Giani, "Wave transport in complexe vibro-acoustic structures in the high-frequency limit," in *IUTAM Symp. Vib. Anal. Struct. with Uncertainties*, 2011.
- [53] T. Hartmann, S. Morita, G. Tanner, and D. J. Chappell, "High-frequency structure- and air-borne sound transmission for a tractor model using dynamical energy analysis," *Wave Motion*, vol. 87, pp. 132–150, 2019.
- [54] G. Tanner, "the future of high-frequency vibro-acoustic modelling in engineering applications," 2015, pp. 1–11.
- [55] S. Kulkarni, F. G. Leppington, and E. G. Broadbent, "Vibrations in several interconnected regions: A comparison of sea, ray theory and numerical results," *Wave Motion*, 2001.

- [56] K. A. Beklemysheva, A. O. Kazakov, and I. B. Petrov, "Numerical modeling of ultrasound phased array for non-destructive testing of composites," *Lobachevskii J. Math.*, vol. 40, no. 4, pp. 415–424, 2019.
- [57] A. Le Bot, "Energy exchange in uncorrelated ray fields of vibroacoustics," J. Acoust. Soc. Am., vol. 120, no. 3, pp. 1194–1208, sep 2006.
- [58] D. J. Chappell, G. Tanner, D. Löchel, and N. Søndergaard, "Discrete flow mapping: transport of phase space densities on triangulated surfaces," *Proc. R. Soc. A Math. Phys. Eng. Sci.*, vol. 469, no. 2155, p. 20130153, jul 2013.
- [59] D. Chappell, D. Löchel, N. Søndergaard, and G. Tanner, "Dynamical energy analysis on mesh grids: A new tool for describing the vibro-acoustic response of complex mechanical structures," *Wave Motion*, vol. 51, no. 4, pp. 589–597, jun 2014.
- [60] T. Hartmann, G. Tanner, G. Xie, D. Chappell, and J. Bajars, "Modelling of high-frequency structure-borne sound transmission on fem grids using the discrete flow mapping technique," *J. Phys. Conf. Ser.*, vol. 744, no. 1, 2016.
- [61] T. Hartmann, G. Xie, J. Bajars, D. Chappell, and G. Tanner, "Vibroacoustic energy flow through spot-welds in dynamical energy analysis," *Proc. INTER-NOISE 2016 - 45th Int. Congr. Expo. Noise Control Eng. Towar. a Quieter Futur.*, pp. 1665–1673, 2016.
- [62] S. Morita, G. Tanner, and M. Richter, "High frequency vibration transmission analysis on agricultural tractor by using combined dynamical energy analysis and transfer path analysis approach," in *SAE Tech. Pap.*, vol. Part F1637, 2020.
- [63] J. Bajars, D. J. Chappell, N. Søndergaard, and G. Tanner, "Transport of phase space densities through tetrahedral meshes using discrete flow mapping," J. Comput. Phys., vol. 328, pp. 95–108, 2017.
- [64] S. B. DONG, K. S. PISTER, and R. L. TAYLOR, "On the theory of laminated anisotropic shells and plates," J. Aerosp. Sci., vol. 29, no. 8, pp. 969–975, 1962.
- [65] J. M. Whitney and A. W. Leissa, "Analysis of heterogeneous anisotropic plates," J. Appl. Mech., vol. 36, no. 2, pp. 261–266, jun 1969.
- [66] M. Qatu, Vibration of Laminated Shells and Plates. Elsevier, 2004.
- [67] J. N. J. N. Reddy, Mechanics of laminated composite plates and shells : theory and analysis, 2nd ed. Boca Raton, Fla. ; London: CRC Press, 2004.
- [68] S. Ghinet and N. Atalla, "Vibro-acoustic behaviors of flat sandwich composite panels," *Trans. Can. Soc. Mech. Eng.*, vol. 30, no. 4, pp. 473–493, dec 2006.

- [69] K. N. Cho, C. W. Bert, and A. G. Striz, "Free vibrations of laminated rectangular plates analyzed by higher order individual-layer theory," J. Sound Vib., vol. 145, no. 3, pp. 429–442, 1991.
- [70] A. Nosier, R. K. Kapania, and J. N. Reddy, "Free vibration analysis of laminated plates using a layerwise theory," AIAA J., vol. 31, no. 12, pp. 2335–2346, 1993.
- [71] S. Ghinet, N. Atalla, and H. Osman, "The transmission loss of curved laminates and sandwich composite panels," J. Acoust. Soc. Am., vol. 118, no. 2, pp. 774–790, aug 2005.
- [72] T. S. Plagianakos and D. A. Saravanos, "Higher-order layerwise laminate theory for the prediction of interlaminar shear stresses in thick composite and sandwich composite plates," *Compos. Struct.*, vol. 87, no. 1, pp. 23– 35, 2009.
- [73] S. Ghinet and N. Atalla, "Modeling thick composite laminate and sandwich structures with linear viscoelastic damping," *Comput. Struct.*, vol. 89, no. 15-16, pp. 1547–1561, 2011.
- [74] J. F. Doyle, Wave Propagation in Structures-Spectral Analysis Using Fast Discrete Fourier Transforms, 1997.
- [75] S. K. Datta, R. L. Bratton, T. Chakraborty, and A. H. Shah, "Wave propagation in laminated composite plates," *J. Acoust. Soc. Am.*, vol. 83, no. 6, pp. 2020–2026, 1988.
- [76] L. Gavrić, "Finite element computation of dispersion properties of thinwalled waveguides," pp. 113–124, 1994.
- [77] O. M. Mukdadi, S. K. Datta, and M. L. Dunn, "Elastic guided waves in a layered plate with a rectangular cross section," Am. Soc. Mech. Eng. NDE, vol. 20, no. 2002, pp. 167–173, 2001.
- [78] A. Chakraborty and S. Gopalakrishnan, "A spectrally formulated finite element for wave propagation analysis in layered composite media," *Int.* J. Solids Struct., vol. 41, no. 18-19, pp. 5155–5183, 2004.
- [79] D. J. Mead and S. Parthan, "Free wave propagation in two-dimensional periodic plates," J. Sound Vib., vol. 64, no. 3, pp. 325–348, 1979.
- [80] B. R. Mace, D. Duhamel, M. J. Brennan, and L. Hinke, "Finite element prediction of wave motion in structural waveguides," J. Acoust. Soc. Am., vol. 117, no. 5, pp. 2835–2843, may 2005.
- [81] J. M. Mencik and M. N. Ichchou, "Multi-mode propagation and diffusion in structures through finite elements," *Eur. J. Mech. A/Solids*, vol. 24, no. 5, pp. 877–898, 2005.
- [82] B. R. Mace and E. Manconi, "Modelling wave propagation in twodimensional structures using finite element analysis," J. Sound Vib., vol. 318, no. 4-5, pp. 884–902, 2008.

- [83] E. Manconi, "Modelling wave propagation in two-dimensional structures using a wave/finite element technique," Ph.D. dissertation, 2008.
- [84] Y. Waki and B. R. Mace, "On the application of finite element analysis to wave motion in one-dimensional waveguides," Ph.D. dissertation, 2007.
- [85] N. Aimakov, G. Tanner, and D. Chronopoulos, "Wave scattering in coupled composite plates," in *INTER-NOISE NOISE-CON Congr. Conf. Proc.* Ibiza: Institute of Noise Control Engineering, 2018, pp. 421–431.
- [86] E. Carrera, "Assessment of theories for free vibration analysis of homogeneous and multilayered plates," *Shock Vib.*, vol. 11, no. 3-4, pp. 261–270, 2004.
- [87] A. S. Sayyad and Y. M. Ghugal, "On the free vibration analysis of laminated composite and sandwich plates: A review of recent literature with some numerical results," *Compos. Struct.*, vol. 129, pp. 177–201, oct 2015.
- [88] A. Maji and P. K. Mahato, "Development and applications of shear deformation theories for laminated composite plates: An overview," J. *Thermoplast. Compos. Mater.*, pp. 1–44, 2020.
- [89] N. J. Pagano, "Exact solutions for composite laminates in cylindrical bending," J. Compos. Mater., vol. 3, no. 3, pp. 398–411, 1969.
- [90] —, "Exact solutions for rectangular bidirectional composites and sandwich plates," J. Compos. Mater., vol. 4, no. 1, pp. 20–34, 1970.
- [91] N. Pagano, "Exact solutions for rectangular bidirectional composites and sandwich plates," *Composites*, vol. 1, no. 4, p. 257, 1970.
- [92] Y. Y. Wang, K. Y. Lam, G. R. Liu, J. N. Reddy, and J. Tani, "A strip element method for bending analysis of orthotropic plates," *JSME Int. Journal, Ser. A Mech. Mater. Eng.*, vol. 40, no. 4, pp. 398–406, 1997.
- [93] I. Bosmans, P. Mees, and G. Vermeir, "Structure-borne sound transmission between thin orthotropic plates: Analytical solutions," J. Sound Vib., vol. 191, no. 1, pp. 75–90, mar 1996.
- [94] N. Aimakov, G. Tanner, and D. Chronopoulos, "Transmission and reflection of waves at structural junctions connecting thin orthotropic plates," *Wave Motion*, vol. 102, p. 102716, apr 2021.
- [95] E. Reissner, "The effect of transverse shear deformation on the bending of elastic plates," J. Appl. Mech., vol. 12, no. 2, pp. A69–A77, 1945.
- [96] R. D. Mindlin, "Influence of rotatory inertia and shear on flexural motions of isotropic, elastic plates," J. Appl. Mech., vol. 18, no. 1, pp. 31–38, 1951.
- [97] S. N. Chatterjee and S. V. Kulkarni, "Shear correction factors for laminated plates," AIAA J., vol. 17, no. 5, pp. 498–499, 1979.

- [98] S. Ghinet, N. Atalla, and H. Osman, "Diffuse field transmission into infinite sandwich composite and laminate composite cylinders," J. Sound Vib., vol. 289, no. 4-5, pp. 745–778, feb 2006.
- [99] A. Mejdi and N. Atalla, "Vibroacoustic analysis of laminated composite panels stiffened by complex laminated composite stiffeners," Int. J. Mech. Sci., vol. 58, no. 1, pp. 13–26, may 2012.
- [100] J. N. Reddy, "A simple higher-order theory for laminated composite plates," J. Appl. Mech. Trans. ASME, vol. 51, no. 4, pp. 745–752, 1984.
- [101] N. D. Phan and J. N. Reddy, "Analysis of laminated composite plates using a higher-order shear deformation theory," Int. J. Numer. Methods Eng., vol. 21, no. 12, pp. 2201–2219, 1985.
- [102] B. N. Pandya and T. Kant, "Flexural analysis of laminated composites using refined higher-order c° plate bending elements," *Comput. Methods Appl. Mech. Eng.*, vol. 66, no. 2, pp. 173–198, 1988.
- [103] S. Ghinet and N. Atalla, "Modeling thick composite laminate and sandwich structures with linear viscoelastic damping," *Comput. Struct.*, vol. 89, no. 15-16, pp. 1547–1561, 2011.
- [104] U. Lee, "Spectral element method in structural dynamics," Spectr. Elem. Method Struct. Dyn., pp. 1–454, 2009.
- [105] D. R. Mahapatra and S. Gopalakrishnan, "A spectral finite element for analysis of wave propagation in uniform composite tubes," J. Sound Vib., 2003.
- [106] D. Roy Mahapatra and S. Gopalakrishnan, "A spectral finite element model for analysis of axial-flexural-shear coupled wave propagation in laminated composite beams," *Compos. Struct.*, vol. 59, no. 1, pp. 67–88, 2003.
- [107] A. Chakraborty and S. Gopalakrishnan, "A spectral finite element model for wave propagation analysis in laminated composite plate," J. Vib. Acoust., vol. 128, no. 4, pp. 477–488, aug 2006.
- [108] O. M. Mukdadi and S. K. Datta, "Transient ultrasonic guided waves in layered plates with rectangular cross section," J. Appl. Phys., vol. 93, no. 11, pp. 9360–9370, 2003.
- [109] D. Samaratunga, R. Jha, and S. Gopalakrishnan, "Wave propagation analysis in laminated composite plates with transverse cracks using the wavelet spectral finite element method," *Finite Elem. Anal. Des.*, vol. 89, pp. 19–32, oct 2014.
- [110] —, "Wavelet spectral finite element for wave propagation in shear deformable laminated composite plates," *Compos. Struct.*, vol. 108, no. 1, pp. 341–353, 2014.

- [111] —, "Wave propagation analysis in adhesively bonded composite joints using the wavelet spectral finite element method," *Compos. Struct.*, vol. 122, pp. 271–283, 2015.
- [112] N. Nanda, "Wave propagation analysis of laminated composite shell panels using a frequency domain spectral finite element model," *Appl. Math. Model.*, vol. 89, pp. 1025–1040, 2021.
- [113] G. Floquet, "Sur les équations différentielles linéaires à coefficients périodiques," Ann. Sci. l'École Norm. supérieure, vol. 12, pp. 47–88, 1883.
- [114] D. Mead, "A general theory of harmonic wave propagation in linear periodic systems with multiple coupling," J. Sound Vib., vol. 27, no. 2, pp. 235–260, mar 1973.
- [115] A. Y. A. Abdel-Rahman, "Matrix analysis of wave propagation in periodic systems," Ph.D. dissertation, University of Southampton, 1979.
- [116] D. Duhamel, B. R. Mace, and M. J. Brennan, "Finite element analysis of the vibrations of waveguides and periodic structures," J. Sound Vib., vol. 294, no. 1-2, pp. 205–220, 2006.
- [117] M. L. Accorsi and M. S. Bennett, "A finite element based method for the analysis of free wave propagation in stiffened cylinders," J. Sound Vib., vol. 148, no. 2, pp. 279–292, 1991.
- [118] C.-m. Nilsson, "Waveguide finite elements applied on a car tyre," Ph.D. dissertation, KTH Royal Institute of Technology, 2004.
- [119] L. Houillon, M. Ichchou, and L. Jezequel, "Wave motion in thin-walled structures," J. Sound Vib., vol. 281, no. 3-5, pp. 483–507, mar 2005.
- [120] C. W. Zhou, J. P. Lainé, M. N. Ichchou, and A. M. Zine, "Wave finite element method based on reduced model for one-dimensional periodic structures," *Int. J. Appl. Mech.*, vol. 7, no. 2, 2015.
- [121] W. J. Zhou, M. N. Ichchou, and J. M. Mencik, "Analysis of wave propagation in cylindrical pipes with local inhomogeneities," J. Sound Vib., vol. 319, no. 1-2, pp. 335–354, 2009.
- [122] M. Maess, N. Wagner, and L. Gaul, "Dispersion curves of fluid filled elastic pipes by standard fe models and eigenpath analysis," J. Sound Vib., vol. 296, no. 1-2, pp. 264–276, sep 2006.
- [123] J. M. Mencik and M. N. Ichchou, "Wave finite elements in guided elastodynamics with internal fluid," *Int. J. Solids Struct.*, vol. 44, no. 7-8, pp. 2148–2167, 2007.
- [124] C. Droz, J.-P. P. Lainé, M. N. Ichchou, and G. Inquiété, "A reduced formulation for the free-wave propagation analysis in composite structures," *Compos. Struct.*, vol. 113, no. 1, pp. 134–144, jul 2014.

- [125] C. Droz, Z. Zergoune, R. Boukadia, O. Bareille, and M. N. Ichchou, "Vibro-acoustic optimisation of sandwich panels using the wave/finite element method," *Compos. Struct.*, vol. 156, pp. 108–114, 2016.
- [126] F. Errico, S. De Rosa, M. Ichchou, F. Franco, and O. Bareille, "Dispersion curves of infinite laminate panels through a modal analysis of finite cylinders," *Wave Motion*, vol. 83, pp. 80–93, 2018.
- [127] Y. Waki, B. R. Mace, and M. J. Brennan, "Numerical issues concerning the wave and finite element method for free and forced vibrations of waveguides," J. Sound Vib., vol. 327, no. 1-2, pp. 92–108, 2009.
- [128] J.-M. Mencik and M. Ichchou, "A substructuring technique for finite element wave propagation in multi-layered systems," *Comput. Methods Appl. Mech. Eng.*, vol. 197, no. 6-8, pp. 505–523, jan 2008.
- [129] D. Chronopoulos, "Wave steering effects in anisotropic composite structures: Direct calculation of the energy skew angle through a finite element scheme," *Ultrasonics*, vol. 73, pp. 43–48, jan 2017.
- [130] E. Manconi and B. R. Mace, "Wave propagation in viscoelastic laminated composite plates using awave/finite element method," 7th Eur. Conf. Struct. Dyn. EURODYN 2008, 2008.
- [131] L. Alimonti, N. Atalla, and B. Gardner, "Assessment of a contour integral method for the nonlinear eigenvalue problem arising from the wave finite element description of two-dimensional periodic waveguides involving dissipative and frequency dependent properties media," in *INTER-NOISE NOISE-CON Congr. Conf. Proc.* Ibiza: Institute of Noise Control Engineering, 2018.
- [132] D. Chronopoulos, M. Ichchou, B. Troclet, and O. Bareille, "Computing the broadband vibroacoustic response of arbitrarily thick layered panels by a wave finite element approach," *Appl. Acoust.*, vol. 77, pp. 89–98, 2014.
- [133] D. Chronopoulos, C. Droz, R. Apalowo, M. Ichchou, and W. J. Yan, "Accurate structural identification for layered composite structures, through a wave and finite element scheme," *Compos. Struct.*, vol. 182, no. September, pp. 566–578, 2017.
- [134] V. Thierry, L. Brown, and D. Chronopoulos, "Multi-scale wave propagation modelling for two-dimensional periodic textile composites," *Compos. Part B Eng.*, vol. 150, no. June, pp. 144–156, 2018.
- [135] M. N. Ichchou, S. Akrout, and J. M. Mencik, "Guided waves group and energy velocities via finite elements," J. Sound Vib., vol. 305, no. 4-5, pp. 931–944, 2007.
- [136] S. Finnveden, "Evaluation of modal density and group velocity by a finite element method," J. Sound Vib., vol. 273, no. 1-2, pp. 51–75, 2004.

- [137] A. Cicirello, B. R. Mace, M. J. Kingan, and Y. Yang, "Sensitivity analysis of generalised eigenproblems and application to wave and finite element models," J. Sound Vib., vol. 478, p. 115345, 2020.
- [138] R. Langley, "Elastic wave transmission coefficients and coupling loss factors for structural junctions between curved panels," J. Sound Vib., vol. 169, no. 3, pp. 297–317, jan 1994.
- [139] L. WANG and F. YUAN, "Group velocity and characteristic wave curves of lamb waves in composites: Modeling and experiments," *Compos. Sci. Technol.*, vol. 67, no. 7-8, pp. 1370–1384, jun 2007.
- [140] D. Chronopoulos, M. Collet, and M. Ichchou, "Wave sensitivity analysis for periodic and arbitrarily complex composite structures," *Eng. Comput.*, vol. 34, no. 5, pp. 1572–1597, jul 2017.
- [141] J. Zhao, J. Qiu, H. Ji, and N. Hu, "Four vectors of lamb waves in composites: Semianalysis and numerical simulation," J. Intell. Mater. Syst. Struct., vol. 24, no. 16, pp. 1985–1994, nov 2013.
- [142] L. Cremer and M. Heckl, *Structure-Borne Sound*. Springer-Verlag, 1973.
- [143] P. G. Craven and B. M. Gibbs, "Sound transmission and mode coupling at junctions of thin plates, part i: Representation of the problem," J. Sound Vib., vol. 77, no. 3, pp. 417–427, 1981.
- [144] B. M. Gibbs and P. G. Craven, "Sound transmission and mode coupling at junctions of thin plates, part ii: Parametric survey," J. Sound Vib., vol. 77, no. 3, pp. 429–435, 1981.
- [145] W. Wöhle, T. Beckmann, and H. Schreckenbach, "Coupling loss factors for statistical energy analysis of sound transmission at rectangular structural slab joints, part i," J. Sound Vib., vol. 77, no. 3, pp. 323–334, aug 1981.
- [146] —, "Coupling loss factors for statistical energy analysis of sound transmission at rectangular structural slab joints, part ii," J. Sound Vib., vol. 77, no. 3, pp. 335–344, aug 1981.
- [147] R. S. Langley and K. H. Heron, "Elastic wave transmission through plate/beam junctions," J. Sound Vib., vol. 143, no. 2, pp. 241–253, 1990.
- [148] B. Mace, "Reciprocity, conservation of energy and some properties of reflection and transmission coefficients," J. Sound Vib., vol. 155, no. 2, pp. 375–381, jun 1992.
- [149] P. Mees and G. Vermeir, "Structure-borne sound transmission at elastically connected plates," J. Sound Vib., vol. 166, no. 1, pp. 55–76, 1993.
- [150] M. D. McCollum and J. M. Cuschieri, "Thick plate bending wave transmission using a mobility power flow approach," J. Acoust. Soc. Am., vol. 88, no. 3, pp. 1472–1479, 1990.

- [151] —, "Bending and in-plane wave transmission in thick connected plates using statistical energy analysis," J. Acoust. Soc. Am., vol. 88, no. 3, pp. 1480–1485, 1990.
- [152] M. B. Skeen and N. J. Kessissoglou, "An investigation of transmission coefficients for finite and semi-infinite coupled plate structures," J. Acoust. Soc. Am., vol. 122, no. 2, pp. 814–822, 2007.
- [153] J. M. Renno and B. R. Mace, "Calculation of reflection and transmission coefficients of joints using a hybrid finite element/wave and finite element approach," J. Sound Vib., vol. 332, no. 9, pp. 2149–2164, 2013.
- [154] B. Chouvion, A. Popov, S. McWilliam, and C. Fox, "Vibration modelling of complex waveguide structures," *Comput. Struct.*, vol. 89, no. 11-12, pp. 1253–1263, jun 2011.
- [155] M. N. Ichchou, F. Bouchoucha, M. A. Ben Souf, O. Dessombz, and M. Haddar, "Stochastic wave finite element for random periodic media through first-order perturbation," *Comput. Methods Appl. Mech. Eng.*, vol. 200, no. 41-44, pp. 2805–2813, 2011.
- [156] M. A. Ben Souf, O. Bareille, M. N. Ichchou, B. Troclet, and M. Haddar, "Variability of coupling loss factors through a wave finite element technique," J. Sound Vib., vol. 332, no. 9, pp. 2179–2190, 2013.
- [157] Y. Fan, M. Collet, M. Ichchou, L. Li, O. Bareille, and Z. Dimitrijevic, "Energy flow prediction in built-up structures through a hybrid finite element/wave and finite element approach," *Mech. Syst. Signal Process.*, vol. 66-67, pp. 137–158, 2016.
- [158] Y. Fan, C. W. Zhou, J. P. Laine, M. Ichchou, and L. Li, "Model reduction schemes for the wave and finite element method using the free modes of a unit cell," *Comput. Struct.*, vol. 197, pp. 42–57, 2018.
- [159] J. M. Mencik, "A wave finite element-based formulation for computing the forced response of structures involving rectangular flat shells," Int. J. Numer. Methods Eng., vol. 95, no. 2, pp. 91–120, 2013.
- [160] G. Mitrou, N. Ferguson, and J. Renno, "Wave transmission through twodimensional structures by the hybrid fe/wfe approach," J. Sound Vib., vol. 389, pp. 484–501, 2017.
- [161] W. Karunasena, A. H. Shah, and S. K. Datta, "Guided waves in a jointed composite plate," J. Acoust. Soc. Am., vol. 95, no. 3, pp. 1206–1212, mar 1994.
- [162] S. Lee and N. Vlahopoulos, "Vibratory power transmission through coupled composite plates with joint compliance and damping," J. Vib. Control, vol. 21, no. 10, pp. 1902–1913, jul 2015.
- [163] D. Chronopoulos, "Calculation of guided wave interaction with nonlinearities and generation of harmonics in composite structures through a wave finite element method," *Compos. Struct.*, vol. 186, no. October 2017, pp. 375–384, 2018.

- [164] R. K. Apalowo and D. Chronopoulos, "A wave-based numerical scheme for damage detection and identification in two-dimensional composite structures," *Compos. Struct.*, vol. 214, no. January, pp. 164–182, 2019.
- [165] G. Mitrou and J. Renno, "Scattering properties of joints in composite plates," in *DYNCOMP 2015*, no. June, 2015.
- [166] K. F. Graff, Wave motion in elastic solids. Oxford: Clarendon Press, 1975.
- [167] G. Tanner and N. Søndergaard, "Wave chaos in acoustics and elasticity," J. Phys. A Math. Theor., vol. 40, no. 50, pp. R443–R509, dec 2007.
- [168] E. A. Skelton, "Free-space green's functions of the reduced wave equation," Admiralty Marine Technology Establishment Teddington (England), Tech. Rep., 1982.
- [169] D. G. Crighton, A. P. Dowling, J. E. F. Williams, M. Heckl, and F. G. Leppington, "Generalized functions," in *Mod. Methods Anal. Acoust.* London: Springer London, 1992, pp. 46–79.
- [170] C. Linton and P. McIver, Handbook of mathematical techniques for wave/structure interactions, 2001.
- [171] M. Smith, "Wave propagation through periodic structures in thin plates," Ph.D. dissertation, University of Auckland, 2013.
- [172] P. Cvitanovic, R. Artuso, R. Mainieri, G. Tanner, and G. Vattay, *Chaos: Classical and Quantum*, 2020.
- [173] M. C. Gutzwiller, Chaos in Classical and Quantum Mechanics, ser. Interdisciplinary Applied Mathematics. New York, NY: Springer New York, 1990, vol. 1, no. 9.
- [174] B. Engquist and O. Runborg, "Computational high frequency wave propagation," Acta Numer., vol. 12, no. July 2003, pp. 181–266, 2003.
- [175] L. Ryzhik, G. Papanicolaou, and J. B. Keller, "Transport equations for elastic and other waves in random media," *Wave Motion*, vol. 24, no. 4, pp. 327–370, dec 1996.
- [176] A. Le Bot, Foundation of Statistical Energy Analysis in Vibroacoustics. Oxford University Press, may 2015.
- [177] G. Gradoni, S. C. Creagh, and G. Tanner, "Radiation of complex sources in reflecting environments: A wigner function approach," *IEEE Int. Symp. Electromagn. Compat.*, no. October, pp. 889–894, 2014.
- [178] —, "A wigner function approach for describing the radiation of complex sources," *IEEE Int. Symp. Electromagn. Compat.*, vol. 2014-Septe, no. September, pp. 882–887, 2014.

- [179] G. Gradoni, S. C. Creagh, G. Tanner, C. Smartt, and D. W. Thomas, "A phase-space approach for propagating field-field correlation functions," *New J. Phys.*, vol. 17, no. 9, 2015.
- [180] S. C. Creagh, G. Gradoni, T. Hartmann, and G. Tanner, "Propagating wave correlations in complex systems," J. Phys. A Math. Theor., vol. 50, no. 4, 2017.
- [181] G. Gradoni, D. M. Ramapriya, S. C. Creagh, G. Tanner, M. H. Baharuddin, H. Nasser, C. Smartt, and D. W. Thomas, "Near-field scanning and propagation of correlated low-frequency radiated emissions," *IEEE Trans. Electromagn. Compat.*, vol. 60, no. 6, pp. 2045–2048, 2018.
- [182] S. C. Creagh, M. Sieber, G. Gradoni, and G. Tanner, "Diffraction of wigner functions," J. Phys. A Math. Theor., vol. 54, no. 1, p. 015701, jan 2021.
- [183] D. J. Chappell, S. Giani, and G. Tanner, "Dynamical energy analysis for built-up acoustic systems at high frequencies," J. Acoust. Soc. Am., vol. 130, no. 3, pp. 1420–1429, sep 2011.
- [184] D. J. Chappell, G. Tanner, and S. Giani, "Boundary element dynamical energy analysis: A versatile method for solving two or three dimensional wave problems in the high frequency limit," J. Comput. Phys., vol. 231, no. 18, pp. 6181–6191, 2012.
- [185] J. E. Ashton and J. M. Whitney, *Theory of laminated plates*. Technomic, 1970.
- [186] L. Briillouin, Wave Propagation in Periodic Structures. New York: McGraw-Hill Publishing Company, 1946.
- [187] R. S. Langley, "A note on the force boundary conditions for twodimensional periodic structures with corner freedoms," pp. 377–381, 1993.
- [188] D. W. Miller and A. von Flotow, "A travelling wave approach to power flow in structural networks," pp. 145–162, 1989.
- [189] Y. I. Bobrovnitskii, "Letters to the editor on the energy flow in evanescent waves," J. Sound Vib., vol. 152, no. 1, pp. 175–176, 1992.
- [190] B. Auld, Acoustic fields and waves in solids. John Wiley & Sons, 1973.
- [191] X. Pan and C. H. Hansen, "Active control of vibratory power transmission along a semi-infinite plate," J. Sound Vib., vol. 184, no. 4, pp. 585–610, 1995.
- [192] I. Thompson, I. D. Abrahams, and A. N. Norris, "On the existence of flexural edge waves on thin orthotropic plates," J. Acoust. Soc. Am., vol. 112, no. 5, pp. 1756–1765, nov 2002.
- [193] D. D. Zakharov, "Konenkov's waves in anisotropic layered plates," Acoust. Phys., vol. 48, no. 2, pp. 171–175, mar 2002.

- [194] A. D. J. Allwright, M. Blakemore, and J. Woodhouse, "Vibration transmission through symmetric resonant couplings," *Philos. Trans. R. Soc. London. Ser. A Phys. Eng. Sci.*, vol. 346, no. 1681, pp. 511–524, 1994.
- [195] R. S. Langley, "Wave transmission through one-dimensional near periodic structures: optimum and to random disorder," J. Sound Vib., vol. 188, no. 5, pp. 717–743, 1995.
- [196] Y. K. Tso and C. H. Hansen, "The transmission of vibration through a coupled periodic structure," J. Sound Vib., vol. 215, no. 1, pp. 63–79, 1998.
- [197] T. Kottos and U. Smilansky, "Quantum chaos on graphs," Phys. Rev. Lett., vol. 79, no. 24, pp. 4794–4797, dec 1997.
- [198] C. Brewer, S. C. Creagh, and G. Tanner, "Elastodynamics on graphs—wave propagation on networks of plates," J. Phys. A Math. Theor., vol. 51, no. 44, p. 445101, nov 2018.
- [199] W. Zhong and F. Williams, "On the direct solution of wave propagation for repetitive structures," J. Sound Vib., vol. 181, no. 3, pp. 485–501, mar 1995.
- [200] J. M. Renno and B. R. Mace, "Calculating the forced response of two-dimensional homogeneous media using the wave and finite element method," J. Sound Vib., vol. 330, no. 24, pp. 5913–5927, nov 2011.
- [201] L. Taupin, A. Lhémery, V. Baronian, and A. S. Bonnet-Bendhia, "Scattering of obliquely incident guided waves by a stiffener bonded to a plate," J. Phys. Conf. Ser., vol. 353, no. 1, 2012.
- [202] D. J. Ewins, Modal testing : theory and practice. Letchworth, Hertfordshire, England; New York: Research Studies Press; Wiley, 1984.
- [203] R. J. Allemang, "The modal assurance criterion twenty years of use and abuse," Sound Vib., vol. 37, no. 8, pp. 14–21, 2003.
- [204] B. Banerjee, J. Chen, R. Das, and A. Kathirgamanathan, "Comparison of anys elements," pp. 1–53, 2011.
- [205] V. Zalizniak, Y. Tso, and L. Wood, "Waves transmission through plate and beam junctions," Int. J. Mech. Sci., vol. 41, no. 7, pp. 831–843, jul 1999.
- [206] M. N. Ichchou, J. Berthaut, and M. Collet, "Multi-mode wave propagation in ribbed plates: Part i, wavenumber-space characteristics," Int. J. Solids Struct., vol. 45, no. 5, pp. 1179–1195, 2008.
- [207] M. J. Leamy, "Exact wave-based bloch analysis procedure for investigating wave propagation in two-dimensional periodic lattices," J. Sound Vib., vol. 331, no. 7, pp. 1580–1596, mar 2012.

- [208] E. Manconi and B. R. Mace, "Wave characterization of cylindrical and curved panels using a finite element method," J. Acoust. Soc. Am., vol. 125, no. 1, pp. 154–163, 2009.
- [209] J. M. Renno and B. R. Mace, "Calculating the forced response of cylinders and cylindrical shells using the wave and finite element method," J. Sound Vib., vol. 333, no. 21, pp. 5340–5355, oct 2014.
- [210] N. Søndergaard and D. J. Chappell, "Ray and wave scattering in smoothly curved thin shell cylindrical ridges," J. Sound Vib., vol. 377, pp. 155–168, 2016.
- [211] V. Blakaj, G. Gradoni, S. C. Creagh, G. Tanner, and M. D. Despande, "Reflected wave fields from a fluctuating earth surface - a phase-space approach," 14th Eur. Conf. Antennas Propagation, EuCAP 2020, no. 3, pp. 3–7, 2020.
- [212] S. W. Tsai, "Structural behavior of composite materials," Tech. Rep., 1964.

Appendix A

Special laminated plate stiffness coefficients derivation

In this appendix, we describe how extensional, bending-extension coupling and bending stiffness coefficients of regular cross- and angle-ply symmetric laminates with alternating material angle of orientation between layers are related to plane-stress reduced stiffnesses of an orthotropic lamina (see [1, 212]).

A.1 Cross-ply symmetric laminates

We consider a cross-ply symmetric laminate consisting of n specially orthotropic layers of the same material with principal material directions alternating between 0° and 90°. The number of layers n must be odd for the laminate to be symmetric. The odd-numbered layers are reinforced in the x axis of the laminate. The principal direction of the even-numbered layers coincides with the y axis of the laminate. Furthermore, the thicknesses of the odd-numbered layers are all equal. The same applies for the even-numbered layers, however, this thickness can be different from that of the odd-numbered layers. We denote as m the ratio of total thickness of the odd-numbered layers to that of the even-numbered layers. Further, F is defined as the ratio of principal stiffnesses of a layer, i.e. $F = Q_{22}/Q_{11} = E_2/E_1$. The extensional and bending stiffnesses of the laminate can be expressed as

$$A_{11} = h \frac{m + F}{1 + m} Q_{11}$$

$$A_{22} = h \frac{1 + mF}{1 + m} Q_{11}$$

$$A_{12} = h Q_{12}$$

$$A_{66} = h Q_{66}$$

$$D_{11} = \frac{h^3}{12} [(F - 1)P + 1] Q_{11}$$

$$D_{22} = \frac{h^3}{12} [(1 - F)P + F] Q_{11}$$

$$D_{12} = \frac{h^3}{12} Q_{12}$$

$$D_{66} = \frac{h^3}{12} Q_{66}$$
with
$$P = \frac{1}{(1 + m)^3} + \frac{m(n - 3) [m(n - 1) + 2(n + 1)]}{(n^2 - 1)(1 + m)^3}$$
(A.1)

A.2 Angle-ply symmetric laminates

We consider an angle-ply symmetric laminate consisting of n generally orthotropic layers of the same material with principal material directions alternating between $-\alpha$ and α , $\alpha \in (0^{\circ}, 90^{\circ})$. The number of layers n must be odd for the laminate to be symmetric. The odd-numbered layers are oriented at $-\alpha$, the even-numbered layers are oriented at α with respect to the x axis of the laminate. All layers are assumed to have the same thickness. The extensional and bending stiffnesses of the laminate can be expressed as

$$D_{11} = \frac{h^3}{12} Q_{11}$$

$$A_{11} = h Q_{11}$$

$$A_{22} = h Q_{22}$$

$$A_{12} = h Q_{12}$$

$$A_{66} = h Q_{66}$$

$$D_{12} = \frac{h^3}{12} Q_{12}$$

$$A_{16} = \frac{h}{n} Q_{16}$$

$$D_{66} = \frac{h^3}{12} Q_{66}$$

$$D_{16} = \frac{h^3}{12} \left[\frac{3n^2 - 2}{n^3}\right] Q_{16}$$

$$D_{26} = \frac{h^3}{12} \left[\frac{3n^2 - 2}{n^3}\right] Q_{26}$$
(A.2)

- 9

Appendix B Plane stress-reduced stiffnesses

The plane stress-reduced stiffnesses Q_{ij} of a generally orthotropic lamina in (2.46) can be related to the stiffness coefficients \bar{Q}_{ij} of the same lamina described in a local coordinate system aligned with the principal material coordinate axes as follows:

$$\begin{aligned} Q_{11} &= \bar{Q}_{11} \cos^4 \phi + 2 \left(\bar{Q}_{12} + 2 \bar{Q}_{66} \right) \sin^2 \phi \cos^2 \phi + \bar{Q}_{22} \sin^4 \phi, \\ Q_{12} &= \left(\bar{Q}_{11} + \bar{Q}_{22} - 4 \bar{Q}_{66} \right) \sin^2 \phi \cos^2 \phi + \bar{Q}_{12} \left(\sin^4 \phi + \cos^4 \phi \right), \\ Q_{22} &= \bar{Q}_{11} \sin^4 \phi + 2 \left(\bar{Q}_{12} + 2 \bar{Q}_{66} \right) \sin^2 \phi \cos^2 \phi + \bar{Q}_{22} \sin^4 \phi, \\ Q_{16} &= \left(\bar{Q}_{11} - \bar{Q}_{12} - 2 \bar{Q}_{66} \right) \sin \phi \cos^3 \phi + \left(\bar{Q}_{12} - \bar{Q}_{22} + 2 \bar{Q}_{66} \right) \sin^3 \phi \cos \phi, \\ Q_{26} &= \left(\bar{Q}_{11} - \bar{Q}_{12} - 2 \bar{Q}_{66} \right) \sin^3 \phi \cos \phi + \left(\bar{Q}_{12} - \bar{Q}_{22} + 2 \bar{Q}_{66} \right) \sin \phi \cos^3 \phi, \\ Q_{66} &= \left(\bar{Q}_{11} + \bar{Q}_{22} - 2 \bar{Q}_{12} - 2 \bar{Q}_{66} \right) \sin^2 \phi \cos^2 \phi + \bar{Q}_{66} \left(\sin^4 \phi + \cos^4 \phi \right). \end{aligned}$$
(B.1)

Here, ϕ is the angle of rotation of the principal material coordinate system with respect to the local coordinate system of the lamina. Furthermore, \bar{Q}_{ij} can be expressed in terms of the material constants as follows:

$$\bar{Q}_{11} = \frac{E_x}{1 - \nu_{xy}\nu_{yx}}, \quad \bar{Q}_{22} = \frac{E_y}{1 - \nu_{xy}\nu_{yx}},$$

$$\bar{Q}_{12} = \nu_{xy}\bar{Q}_{22}, \quad \bar{Q}_{66} = G_{xy}, \quad \nu_{yx} = \nu_{xy}\frac{E_y}{E_x}.$$
(B.2)

Note, that for $\phi = \pi n/2, n \in \mathbb{Z}$, the plane stress-reduced stiffnesses Q_{16} and Q_{26} are equal to zero, the corresponding laminas are called *specially orthotropic*.

Appendix C

List of Abbreviations

$\rm FE$	Finite Element
FD	Finite Difference
BE	Boundary Element
SEA	Statistical Energy Analysis
WIA	Wave Intensity Analysis
EFEA	Energy Finite Element Analysis
DEA	Dynamical Energy Analysis
ESL	Equivalent Single Layer
SE	Spectral Element
WFE	Wave Finite Element
CLP	Classical Laminated Plate
FSDL	First-order Shear Deformation Laminated
HSDL	Higher-order Shear Deformation Laminated

TSDL Third-order Shear Deformation Laminated